27766 NATIONAL LIBRARY OTTAWA



BIBLIOTHÈQUE NATIONALE OTTAWA

NAME OF AUTHOR. KILDY W-H. YUEN TITLE OF THESIS. PREVEN FILL SOCIAL SERVICES AS COMMUNITY DEVELOPMENT.

UNIVERSITY .... OF ... ALBERTA.....

Permission is hereby granted to THE NATIONAL LIBRARY OF CANADA to microfilm this thesis and to lend or sell copies of the film.

The author reserves other publication rights, and neither the thesis nor extensive extracts from it may be printed or otherwise reproduced without the author's written permission.

(Signed).

PERMANENT ADDRESS:

Box 145 University of Alberta Edmonton T64:258

DATED. April 2.6..... 1970

NL-91 (10-68)

#### INFORMATION TO USERS

### THIS DISSERTATION HAS BEEN MICROFILMED EXACTLY AS RECEIVED

This copy was produced from a microfiche copy of the original document. The quality of the copy is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

PLEASE NOTE: Some pages may have indistinct print. Filmed as received. AVIS AUX USAGERS

LA THESE A ETE MICROFILMEE TELLE QUE NOUS L'AVONS RECUE

Cette copie a été faite à partir d'une microfiche du document original. La qualité de la copie dépend grandement de la qualité de la thèse soumise pour le microfimage. Nous avons tout fait pour assurer une qualité supérieure de reproduction.

NOTA BENE; La qualité d'impressionde certaines pages peut laisser à désirer. Microfilmée telle que nous l'avons reçue.

Canadian Theses Division Cataloguing Branch National Library of Canada Ottawa, Canada KIA ON4 Division des thèses canadiennes Direction du catalogage Bibliothèque nationale du Canada Ottawa, Canada KIA ° ON4 THE UNIVERSITY OF ALBERTA .

PREVENTIVE SOCIAL SERVICES AS COMMUNITY DEVELOPMENT

by i,

KILDY WING-HAN YUEN

A THESIS

SUBMITTED TO THE PACULTY OF GRADUATE STUDIES AND RESEARCH IN PARTIAL FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE.

DIVISION OF COMMUNITY DEVELOPMENT

EDMONTON, ALBERTA SPRING, 1976 THE UNIVERSITY OF ALBERTA FACULTY OF GRADUATE STUDIES AND RESEARCH

Super

Date Dellege 2, 1975

The broad purpose of this study is to provide an understanding of the nature of the Preventive Social Services Program. More specifically, the Preventive Social Service projects in Edmonton are analyzed and examined in terms of principles and methods of community development.

The population for the study includes the twenty Preventive Social Services projects funded in the City of Edmonton in the Spring of 1975. Data for the study were collected through personal interviews with the key personnel of respective projects, and through a review of accessible documents and reports. General information with regard to the Preventive Social Services Program were obtained through interviews with the staff members of the Provincial Department of Health and Social Development and the Social Service Department of the City.

The major concepts of the Preventive Social Service Program are described. The historical background, sponsorship, objectives, clientere, methods of work, and organizational structure of these twenty projects

are then studied in this thesis.

An analytical framework, developed from a review of community development literature, has been used to study the various aspects of the projects. The initiation of the projects is examined in terms of the extent of the local participation on the part of the community. Furthermore, the needs inherent in the objectives, the preventative orientation

ABSTRACT

versus rehabilitative prientation of the objectives and the development of the objectives has been analyzed. The types of clientel's served by respective projects is studied in terms of geographic and functional communities. With regard to the methods of work; the process-orientation versus task-orientation of the strategy, the scope of the project, the recruitment of volunteers as a means of mobilization of human resources form the focus of analysis. Finally, the form of citizen participation

in the organizational structure has been examined. The findings from the analysis of the various aspects of the

projects revealed that: eighteen percent of the projects have been identified as community development activities; thirty-five percent of the projects have been employing some of the principles and methods of community development to a large extent; fifteen percent of the projects have shown the general characteristics of a community development activity; and the other thirty-two percent have not been identified as community development.

The findings also reveal that the objectives of most of the projects are preventative in nature and are basically social objectives. With regard to citizen participation at the policy decision-making level, most of the projects have chosen the form of Directive Boards or Advisory Committees.

#### ACKNOWLEDGEMENTS

Acknowledgements are due to many people--those who have taught me to have questioned me, and those who have helped me. of all, I would like to express my gratitude to my those supervisor Dr. Dhaza Gill, not only for his interest and enthusiasm in making this thesis a reality, but also his patience, encouragement, and valuable suggestions, well beyond the normal prescribed duties of a thesis supervisor. Similarly, I would like to extend my sincere appreciation to the other members of my thesis committee, Dr. Charles Hynam and Dr. George Kupfer, for their perceptive critique, sustained interest and cooperation.

Special thanks also go to the Andividual respondents who gave time to the interviews. This thesis simply would not have been possible without their cooperation.

Thanks are extended to all my classmates, Miss Cora Arends, and Dr. Glen Eyford for being so helpful throughout my study in the program.

A special word of appreciation is extended to Mrs. Schmidt who typed my thesis.

The last, but not least, I wish to thank my parents who always support me and encourage my endeavours. This thesis is dedicated to them. Also the help from my brother, Stephen, is gratefully acknowledged.

	•	
		. 🍽
. TABLE OF CONTENTS		
		<b>. 1</b>
CHAPTER	PAGE "	•
I INTRODUCTION	1	· · · · · · · · · · · · · · · · · · ·
The Nature and Significance of the Problem	L	ſ
Purpose of the Study	4	•
Rationale for the Study	5°	•
Methods of Date Dection and Analyse The Study Population	5	
	6	• • •
Data Collection and Analysis Limitations of the Study	. 8	•
Outline of the Thesis	9	•
Footnotes	11	
II GOMMUNITY DEVELOPMENT: ITS RURAL ORIGINS: AND		٠
APPLICATIONS IN AN INDUSTRIAL SOCIETY	12	•
Historical Perspective on Community Development	• 13	-
Early Experience in India	14	
Mass Education in China	18	
The British Experience in Africa	18	
Community Development as a United Nations- supported Program	21	
Towards An Understanding of Community Development in Developed Countries	22	
The Context of Community Development - A Generalization	23	۰ ۲.
Community Development in the Context of Developed Countries	25	
vii.	•	

		• y
and a second	PA	GE
CHAPTER		27
. •	Summary	28
•	Footnotes	21
III C	COMMUNITY DEVELOPMENT: THE CONCEPTUAL FRAMEWORK	31 31
	Defining Community Development	:
and and a second se	The Concept of Community	35
	The Principles of Community Development	36
• <u>·</u>	The Strategy of Community Development	40
•	Citizen Participation	43
	The Analytical Framework	47
• • • •	Initiation of the Projects	47
•	Operationalization of the Projects	47
	Summary	50
•	Footnotes	52
. •1V	PREVENTIVE SOCIAL SERVICES: CONCEPT AND ORGANIZATION	52
•	The Principles	57
•	The Objectives	54
•	The Organization and Administration	•
•	The Role of the Province	57
<i>у</i>	The Role of the Municipality	58
· · · · · · · · · · · · · · · · · · ·	The Role of the Local Director	59
•	Cost Sharing	60
•	Program Development and Selection	61
₹.	What is Prevention?	63
•	Summary	65
	Footnotes	66
•	* viij	5

<b>-</b> 3	<b>.</b>		۵. <b>۴</b> ۲.			
CHAPTER.	•	•		5 m 4	PAGE	•
V	PREVENTIVE SOCI EDMONTON: A DE	AL SERVICES PR	OJECTS IN	••••••	67	
· · · · · · · · · · · · · · · · · · ·	The Descripti	ion of Six Prov	entive Social S	ervices	, 70	
•	Family, Life	Education Cou	incil of Edmonto	n	, 70	•
•			•••••		74	
•	•Day Care •		• • • • • • • • • • • • • • • • •		. 74	
· · · · · · · · · · · · · · · · · · ·	• •		and Semi-Retired	; •••••	. 83	• •
•			centre		. 87	• •
	-	· · · · · · · · · · · · · · · · · · ·		6	. 91	
· · .					. 95	. •
\ <b>:</b>	Summary	•	• • • • • • • • • • • • • • • • • • • •		. 96	•
•	Footnotes				• .	•
VI ·	PREVENTIVE SOC COMMUNITY DEVE	IAL SERVICES V LOPMENT CONCEP	TS		• - 97-•	
· · · · · · · · · · · · · · · · · · ·	• • · · · · · · · · · · · · · · · · · ·	al Framework .			. 9.7	• •
•		of the Projec			. 98	• •
•	Operationa	alization of th	e Projects	• • • • • • • • • • • • • • • • • • •	. 100	•
	Preventive S	Social Services	in Edmonton:	An Analysis	108	
<b>~</b>	<ul> <li>• • • • • • • • • • • • • • • • • • •</li></ul>	•	ts		. 108	
•		•	ne Projects'		. 112	•
•	Summary		· · · · · · · · · · · · · · · · · · ·		. 133	•
• Constant of the second se		· · · · · · · · · · · · · · · · · · ·			. 13'5	
		LUSIONS AND IM			136	1
VII	• ۵				. 136	•
<b>4</b> • <b>4</b>	Review		• • • • • • • • • • • • • • • • •		. 137	•
	The Finding	S		• • • • • • • • • • • • • • • • • • •		•
		a	•	•,		•
•		•	.ix -			•
•		• •	•			

PAGE CHAPTER 139. Non-governmental Projects ... 142 Governmental Projects ... 143 Conclusions . 145 Implications ..... . 147 Concluding Remarks ..... 148 BIBLIOGRAPHY ... APPENDIGES 155 LIST OF QUESTIONS ..... A. A DESCRIPTION OF FOURTEEN PREVENTIVE SOCIAL SERVICES . 157



• * •		
	LIST OF FIGURES	,
FIGURE	PAGE	
1	The Eight-Rung Ladder of Citiken Participation	
2	The Administrative Structure of Preventave 6 6 53	•
3	The Administrative Structure of Preventive Social Services - Municipal	-
. 4	The Generalized Administrative Structure of Proventive Social Services - Rural	
5	The Prevention Continuum	•
6	The Organizational Structure of the Family Life of 73.	
<b>7</b>	The Organizational Structure of the Edmonton Big Sisters Society	: ა
- 8	The Organizational Structure of the Society for.	
9	The Organizational Structure of the Norwood Community Service Centre	ں د پر ف
<b>_</b> 10	The Organizational Structure of the Home Care Program	,
11	The Analytical Framework	•
12	Level of Participation in the Inftiation of 99 the Projects	
13	Preventative Orientation Versus Rehabilative Orientation of the Objectives	
14	Development of the Objectives (péople determined versus predetermined by the government	。 。
15 V <sup>i</sup>	Types of Clientele	-
16	Process Orientation Versus Task Orientation	ື 0
•		2
:		

•

. a

ø

PAGE Mobilization of Human Resources . . Level of Participation in the Initiation of the P.S.S. Projects ..... Preventative Or Tentation Versus Rehabilative Orientation of the Objectives of the P.S.S. Projects ...

107

109

114

y lin

· . . .

		a a a a a a a a a a a a a a a a a a a	
	20	The Development of the Objectives of the P.S.S. Projects	116
	21	Types of Clientele Served by the P.S.S. Projects	·119
•	22	Process-Orientation Versus Task-Orientation of the Strategy of the P.S.S. Projects	121.
X	23	Mobilization of Human Resources	126
) •	24	Arnstein's Eight-Rung Typology of Citizen Participation	129
: t	25	The Organizational Structure of Edmonton Family Planning Service	159
Ĩ.	26	The Organizational Structure of Homemaker's Service	162
	27 ,	The Organizational Structure of Urban Life Skills and Child Care Program . 2	167
	28	The Organizational Structure of the St. Francis Youth Club	170
•	29	The Organizational Structure of Teen Time	173
	30	The Organizational Structure of the Bissell - Child Care Centre	176
	31	The Organizational Structure of Operation Friendship	180
_	32	The Organizational Structures of Strathcona Place Society	184
0	<u>3</u> 3 .	The Organizational Structure of Dickensfield We Care Centre	187

FIGURE

17

18

19

3

PAGE FIGURE 190 The Organizational Structure of Open Door ..... 34 The Organizational Structure of A.I.D. .... / 192 35 The Organizational Structure of Meals-on Wheels . 199 36 0 xiv

### CHAPTER I

#### INTRODUCTION

# The Nature and Significance of the Problem

In most parts of the world today powerful forces--social, economic, and technological--are at work within and upon societies. Population growth, industrialization, urbanization, and technological developments all give rise to needs and problems in and around individual

lives and human community.

Bregha, in his broad analysis of contemporary social change in Canada, outlines the following trends:

Firstly, an irreversible change process proceeding at an accelerating rate increases social economic and political inequalities within and between countries.

Secondly, the transition from industrialism to postindustrialism will be accompanied by increasing social turbulence, created by the complexity and the size of the total environment as well as by the impact of communications. While the interdependence of the parts will be more readily recognized, the unpredictability of connections between them will also increase, widening thus the area of relevant uncertainty for individuals and organizations.

Thirdly, whereas evidence for the above described change is already available, what has not yet occurred is a corresponding transformation in our cultural values, organizational philosophies and ecological strategies. Growing alienation and social disintegration are the inevitable results.

Finally, under these conditions welfare and development are becoming increasingly interrelated, one impossible without the other.<sup>1</sup> Another trend identified by Bregha is the growing number of dependents and 'problem-groups'. No longer do these terms apply only to groups such as the aged, the Native people, the mentally and the physically-handicapped. Large numbers of the middle class are also forced out of the so-called mainstream of Canadian life.<sup>2</sup>

It is evident that the stresses of the enormous adaptation of human life to the conditions of a modern industrial society are among the major sources of contemporary social disorders. The increase in mental illness, family breakdown, alienation, and so on are not in themselves the diseases, but are the symptoms arising from rapid and only partially assimilated changes by humans in new conditions. As the personal problems and stresses of living affect everyone, social services\* are increasingly needed for the well-being of all people-tich or poor alike. In modern times, people have to look to their society to provide services that will enable them to function adequately and satisfactorily at all ages.

Under these conditions, it is not surprising to witness the emergence of a new look in social services delivery. Recently, J.S. Morgan detected seven major trends in the changing demands for social services in North America and throughout the world. They are as follows:

\*"Social Service is a broad term used to indicate the provision of services, other than income support. Thus adoption, day care, protection, probation, are all social services." Source: Andrew Armitage, <u>Social Welfare in Canada Ideals and RealPities</u>, Toronto, McClelland and Stewart Ltd., 1975, p. 210. 1) From curative to preventive services.

 From individualized to universalized services; and, in this connection, from small-scale to large-scale operations.
 From private to public responsibility for the social services.
 From services providing income only to services that provide

income with additional service benefits. 5) From the criteria of poverty to the criterion of need as a test of eligibility for social services.

- 6) From a treatment orientation of services to the social administration of a large-scale services.
- 7) From a limited view of welfare as a palliative based or current assumptions about the distribution of wealth or about personal responsibility for well-being to a broader view of welfare as encompassing the human condition.<sup>3</sup>

In response to changes in the nature of social problems and diversified social needs, the functions of social services are undergoing substantial alteration. Traditionally, Statutory Social Services (e.g. Child Welfare, Correction, etc.) performed a residual function, attending to emergencies and crisis situations, withdrawing when regular social structures (e.g. family) began to meet human needs. As band-aid approaches usually have no long-term effect--socially and economically--the wealth of experience acquired in remedial action dealing with the adverse effects of rapid change may be put to good use in planning for various social programs with primarily preventive purposes. Hence, social services are facing the need to employ new concepts and new approaches which encourage flexibility, and which also embody a more sophisticated knowledge of the workings of modern society and of the. social problems attending urbanization and industrialization.

The recent introduction of the Preventive Social Services Program in Alberta is an intent to employ a new concept and a new approach to providing social services for the community at large. The 'prevention' concept represents a departure from the traditional role of meeting needs only after some forms of crisis situations .

As will be clear from the material presented in the following chapter, the Preventive Social Services Program is designed to develop community awareness and resources, to strengthen and preserve human initiative and to prevent individual and family breakdown.<sup>4</sup>

# Purpose of the Study

As a student in the Program of Community Development, the author is interested in analyzing the Preventive Social Services in the light of her own field of study.

The purpose of this thesis is to study the various aspects of the Preventive Social Services projects in Edmonton as to what extent these projects have been employing the principles and methods of community development.

. In sum, the specific objectives of this study are:

To provide an understanding of the nature of the Preventive Social Services projects in Ethnonton through a study of the historical background, objectives, clientele, methods of work, and organizational structure.

2. To examine the level of participation on the part of the community in the initiation of the projects.

3. To analyze and examine the Preventive Social Services projects in terms of principles and methods of community development.

### Rationale for the Study

A preliminary examination of the available documents on the Preventive Social Services Program revealed that the basic tenets of the Program--such as encouraging local planning and decision-making and training of local leadership--are related to the principles of community development. But no attempt has been made to examine how well Preventive Social Services utilize community development principles and methods. Therefore, this thesis is designed to study the Preventive Social Services in the light of community development concepts, which would provide an understanding of the nature of the Program.

On the other hand, Alberta is the only province to commit the concept of prevention into legislation within the broad application of social development. As Alberta's Preventive Social Services Program is unique to Canada, a study of the Program would be of interest to many students, community workers etc. Therefore, this thesis is also an effort to provide information about the Preventive Social Services Program in Alberta and, more specifically, the Preventive Social Services projects in Edmonton at the time of the study.

Methods of Data Collection and Analysis

#### The Study Population

Since the basic objective of this thesis is to study the Preventive Social Services in the light of community development concepts, information regarding the various aspects of the Preventive Social Services, projects in Edmonton is required. A list of all the Preventive Social Services projects funded in the City of Edmonton was obtained in the fall of 1974. This list of projects was updated at the time of the study (i.e. in the spring of 1975).

Part of the rationale for choosing all the Preventive Social Services projects in Edmonton is that those projects Anclude different types of preventive social services. Also, because the author lives in Edmonton, it was possible for her to explore more extensively the projects in that area.

Data Collection<sup>®</sup>and Analysis

Several methods of data collection were employed to obtain information for the study.

The material for Chapter IV, which is a description of the Preventive Social Services Program, was obtained through interviews with various staff members of the Preventive Social Services Branch, Provincial Department of Health and Social Development; through interviews with the staff of Social Service Department of the City of Edmonton; and through a review of pamphlets and reports that were available from these two departments.

The major portion of information for the study, which includes the descriptions of twenty Preventive Social Services projects, was collected through personal interviews with the executive directors or other key persons in the respective projects. A list of questions pertinent to the various aspects of the projects (Appendix A) was used as the guidelines for obtaining information from respondents in order to develop a descriptive case study of each project. Interviews were requested and arranged by telephone. The interviews, ranging in length from approximately one to three hours, were tape-recorded. Available literature--annual reports, legislation and committee reports, evaluation reports--were also collected and utilized.

Since the identification of a project as community development is based on the objectives, the issues involved, the types of clientele, the methods employed and so on, an adialytical framework by which to examine the above aspects of the project was formulated from the review of literature on community development. The initiation of the projects was examined in terms of the extent of local participation on the part of the community. For the objectives of the projects, the needs inherent in the objectives, the preventative orientation versus rehabilitative orientation of the objectives, and the development of the objectives were analyzed. The types of clientele served by respective projects were. studied in terms of geographic and functional communities. With regard to the methods of work, the process-orientation versus task-orientation of the strategy, the scope of the project, the ecruitment of volunteers as a means of mobilization of human resources were analyzed. Finally, the form of citizen participation in the lorganization structure was studied.

Some of the elements of analysis, which are considered as polartypes or gradient-types, were constructed into continua for the purpose of analysis. One advantage of using a continuum for the analysis was that the relative positions of the projects can be illustrated in one diagram.

# Limitations of the Study

First, while the purpose of this thesis is to study Preventive Social Services as community development, the study does not seek to provide a comprehensive description of all Preventive Social Services activities throughout the Province of Alberta. Such a compendium would be undoubtedly useful, but to describe and analyze all existing Preventive Social Services projects would be beyond the scope of a Master's thesis. (At present, there are 200-plus funded projects and 400-plus non-funded projects). Instead, Preventive Social Services projects in Edmonton, which represent different approches in providing preventive social services, have been chosen for the study. Therefore, the study cannot represent the total Preventive Social Services Program.

Second, with regard to the subjective nature of the data as information obtained through personal interviews with the directors of the projects, the information does not necessarily represent the consensus of all the relevant employees and board members.

Third, the author has supplemented the findings from a limited number of other sources, e.g. brochures, pamphlets, reports, etc.; a complete review of all the literature regarding each project was not undertaken.

Finally, no attempt has been made to evaluate the projects studied

for such an evaluation was not appropriate in the absence of first-hand field observations.

#### Outline of the Thesis

This thesis consists of seven chapters. Chapter I includes the nature of the problem, the purpose of the study, rationale for the study, methods of data collection and analysis, limitations of the study, and the outline of the thesis.

Chapter II provides a descriptive account of the rural origins of community development and further, examines the application of community development to an industrial society. This Chapter attempts to lay the foundation for the discussion of the study's conceptual framework in the following chapter.

Chapter III presents the review of literature on 1) the O definitions of community development, 2) the concepts of community, 3) the principles of community development, 4) the strategy of community development, and 5) citizen involvement. At the end of this chapter, the analytical framework developed from the review of literature is described.

Chapter IV presents the major concepts and organization of the Preventive Social Services Program.

Chapter V describes the historical background, sponsorship, objectives, clientele, methods of work and organizational structures of six Preventive Social Services projects in Edmonton. (The description of the other fourteen projects is presented in Appendix B.) No attempt is made to evaluate their worth or success.

Chapter VI describes the analytical framework and presents the analysis of the twenty Preventive Social Services projects. Analyses of the projects include: the level of participation on the community (people and groups) in the initiation of the projects, the needs inherent in the objectives, the preventative orientation versus rehabilitative orientation of the objectives, the development of the objectives, types of clientele, process-orientation versus task-orientation of the methods of work, the scope of the project, mobilization of human resources, and finally, citizen participation in the organization structure.

10

Chapter VII contains the summary, conclusions, and implications of the study.

#### Footnotes

<sup>1</sup>Francis J. Bregha, "Community Development in Canada: Problems and Strategies", <u>Community Development Journal</u>, Vol. V, No. 1, Jan., 1970, p. 31.

<sup>2</sup>Ibid., p. 32.

<sup>3</sup>John S. Morgan, "The Changing Demands for Changing Clients", <u>Changing Services for Changing Clients</u>, NASW, New York, Columbia University Press, 1969, p. 3.

<sup>4</sup><u>Preventive Social Services, Administrative and Policy Outline</u>, Department of Health and Social Development, Alberta, Edmonton, 1970, p. 1.

#### CHAPTER II

# COMMUNITY DEVELOPMENT: ITS RURAL ORIGINS, "AND APPLICATIONS IN AN INDUSTRIAL SOCIETY

Community Development is not a new concept. It has many precursors who have contributed to the evolution of the practice and also a wide range of definitions. In fact, it was variously labeled as 'mass education', 'rural reconstruction', 'agricultural extension', and so on. In attempting to clarify the subject, most people begin their discussion on community development by stating a definition. However, in this study, the author will briefly look at the emergence of community development as a method of development in the rural-developing areas " before proceeding to the discussion of its definitions, principles and strategy. The reason is that community development has its roots in earlier attempts to improve the rural life and bring about social change in the less-developed countries characterized by mass illiteracy, low levels of health and standards of fiving, etc. Most<sup>0</sup> of the concepts of community development are formulated from those earlier experiences in . the rural areas of the less-developed countries. An examination of the origins of the concept of community development will help to explain the basic difference between community development in the various settings--o rural and urban, less industrial or highly industrial. Also, it will provide a partial explanation as to why only some of the elements of community development are considered as relevant for the analysis of

12

#### the styley.

In the first part of this chapter, the rigins of community development will be examined. The second section will include a general description of the context of community development. This will be followed by a discussion on the relevance of community development in highly developed countries. The basic purpose of this chapter is to lay the foundation for the discussion of the conceptual framework in the following chapter.

# A Historical Perspective on Community Development

The emergence of community development practice is no accident in history. . It has been a continuous process in various forms for many decades. Those early attempts, of course, were not called community development, but in essence, their aim was more or less the same. As

most of the literature dealing with community development traced its history from a report released by the government of the United Kingdom in 1944, entitled <u>Mass Education in African Society</u>, and took into account only the governmental program the author decided to mention briefly also some other identifiable non-governmental beginnings elsewhere. The following examples, however, are not exhaustive and cannot be regarded as the only origins of community development. The selected cases, in fact, are earlier experiments which were notable and have made outstanding contributions to the practice of community development.

The origins of the idea of community development can be traced to the initiative of Christian Missions in India who endeavoured to bring about the village uplift; to Rabindranath Tagore who tried to build a spirit of self-help in his experiments in Bengal; to Mahatma Gandhi who tried to prepare a blueprint for national reconstruction in India; to James Yen, a graduate of Yale University, who returned to China in 1920's to organize a mass education campaign; and to the British officials who suggested the Mass Education policy in African

#### Early Experiments in India

society.

ŝ

One of the earliest attempts at improving the rural economic conditions was made in 1830, by a Native Philanthropic Society under the leadership of Rev. C.T.E. Rhenius of the Church Missionary Society, London. The aim of the Society was ". . to settle Christian converts on purchased land, to erect schools and chapels and to help them in their temporal concerns."<sup>1</sup> This Society consisted mainly of native Christians and was managed in a democratic manner by a committee of which all the members except the Treasurer were Indians.

In 1850's, the Society had built many churches, schools, houses and had contributed funds for native ministers. This movement, in fact, can be considered as a prototype of community development and a powerful means of uplifting the underprivileged although, in some instances, the Christian peasant settlements failed to establish a healthy selfreliant community.<sup>2</sup> As

. . the Christian peasant settlements were primarily meant for giving refuge to upper caste, and amel@orating the condition of lower caste converts, and it was only gradually that the various services providing for educational, social and physical development were taken up. Thus they moved slowly and unconsciously towards a community development type programme.<sup>3</sup> "Like the community development programme, the Christian peasant settlements aimed at helping the people to help themselves."<sup>4</sup>

Another identifiable forerunner of community development, who had some influence on the acceptance of community development on a national scale in Indian later, was Radindranath Tagore. Tagore, the originator of village level worker idea, believed that people must learn to stand on their own feet. He said, "serve them by all means, but take care that as a result of your service, they learn to stand on their own legs."<sup>5</sup> To achieve this he believed in educating the people instead of giving charity.<sup>6</sup> As in early 1907, when feelings were running high against the Government, Tagore was advising young people to take up village reconstruction.

Tagore started his first experiment in 1908 at Silaidaha (East Bengal). He sent a group of young workers to live in the villages and establish contact with the people. For economic development, efforts were made to start a few cottage industries, introduce new commercial crops, and give demonstrations in the use of manures. A hospital was also started for the general public. "But this work came to an end abruptly in 1919 when the Tagore Estate was partitioned and the Silaidaha property was alloted to other members of the family."<sup>7</sup>

Another experiment of his was at Patisar Estate where a "Welfare Fund" was set up by contributions from the tenants and the landlord of the estate, and for public works. "Voluntary Contributions" were in the form of free labour from the poor and money from the rich. "It has been claimed that this provided the first example of carrying out public works by voluntary labour in Bengal."<sup>8</sup> There was also a provision of credit at reasonable rates through the Agricultural Bank of the Estate.<sup>9</sup> Unfortunately, these works did not last long.

Tagore was not satisfied with his earlier attempts and recognized the need for a more intensive and comprehensive experiment. Therefore, in 1922, he started his most ambitious project--an Institute of Rural

Reconstruction at Sriniketana whose purpose was

o

To bring back life in all its completeness, making the villagers self-reliant and self respectful, acquainted with the cultural tradition of their own country and competent to make an efficient use of modern resources for the fullest development of their physical, social, economic and intellectual conditions.<sup>10</sup>

This Institute aimed at an all-round development of the villages by the efforts of villagers themselves had four main departments:

Village Welfare, Agriculture, Industries and Education. The work of

rural reconstruction was achieved through the promotion of agricultural and dairy farm. A new scheme for primary education which was an

experiment combining work with pTay and play with knowledge was introduced; a village Boy Scout Association was formed; local industries were revived as encouraged by industrial workshops and the establishment of cottage industries; village cooperatives were formed as part of a health union that provided health care facilities. There was also a department of socio-economic research besides its Village Welfare

department. This experiment, undoubtedly, had some achievements, but

It also seems that Sriniketana, with its limited resources in men and money, extended its activities in too many directions and made too many experiments in rapid succession. Due to this the whole programme became unmanageable and many of the schemes never crossed the experiment stage. In short, the whole work was extensive without being intensive.<sup>11</sup> In the author's opinion, though some of these earlier experiments were not considered as completely successful, still, they have shown a basis on which the present structure of community development was being built. Tagore thus was very influential in sowing the seeds of what was later to become community development.

Another influential experiment attempted to stimulate a form of community development that was then termed 'rural reconstruction' was M.K. Gandhi's 'Constructive Programme'.<sup>12</sup> The objective of the Program was to reconstruct the villages of India and make them self-sufficient

For economic development the program stressed "1) the production and popularization of Khadi (cloth); 2) the development of village industries; 3) the improvement of cattle, and 4) the socio-economic equality."<sup>13</sup> For social development it aimed at "5) communal unity;
6) the removal of untouchability; 7) prohibition; 8) equal status for women and 9) the welfare of Adivasis (tribal population)."<sup>14</sup> On the educational side it stressed "10) the adoption of Basic Education;
11) the working for Adult Education; 12) the development of provincial languages and 13) the adoption of Hindustani as the national language."<sup>15</sup> To improve the general health of the villages, it realized the importance of "14) improving village sanitation and 15) imparting health education and 16," taking care of the lepers."<sup>16</sup> On the political front it stressed the necessity of organizing "17) peasants and labourers for securing their just rights and 18) students for doing work."<sup>17</sup>

Gandhi's concept of development was a balanced one, and included

Ū

moral, material and social development of the people. One of his many achievements was the creation of the 'constructive workers'--village workers--who were notable for their identification with villagers; . their selflessness; and the most important, their stresses on the moral quality of the work and the educational and spiritual awakening of the people.

# 'Mass Education' in China

Another precursor of community development that is praiseworthy.was James Yen's Mass Education Campaign in China in 1920's. The main objective of the campaign was to fight the Four Fundamental Weaknesses of Chinese life (ignorance, poverty, disease, and social disintegration).<sup>18</sup>

The purpose of the Chinese Mass Education Movement organized as a National Association in 1923 was to explore the potentialities of the masses and find a way of educating them not merely for life but to remake life . . . Experience in the field resulted in a shift--in emphasis, beginning in 1929 from extensive promotion of literacy to intensive study of life in rural districts.

His literacy campaign was reported to have reached millions of peasants in the sprawling region of China through 'people schools' and 'mass education association'.<sup>20</sup> Unfortunately, his work seemed to suffer from inadequate support at higher political levels and "with the communist revolution his reforms evaporated."<sup>21</sup>

# The British Experience in Africa

Another identifiable origin of community development which is considered as a landmark of the evolution of the practice was the education policy in former British African countries where The concept of community development originated in the search for a program to compensate for the limitations of the conventional school system, and to enable education to provide for the progressive evolution of the people.22

In earlier colonial times, education in the United Kingdom dependent terrftories had largely been left in the hands of the Christian missionaries; however, both missionaries and colonial governments felt that the system was inadequate.<sup>23</sup> In 1925, a White Paper entitled <u>Education Policy in British Tropical Africa</u> was released by the British government. The White Paper suggested three guiding principles

for educational policy:

з.

 that education should be related to all other efforts, whether of Government or of citizens, for the welfare of the community.

o 2. that material prosperity without corresponding growth in the moral capacity to turn it to good use constituted a danger, and

that the real difficulty lay in imparting any kind of education which would not have a disintegrating and unsettling effect upon the people of the country.<sup>24</sup>

It also stated the need for, in addition to an improved system of schools, the improvement of agriculture, local industries, health and

the training of the indigenous people.

"In 1935, the Memorandum on the Education of African Communities was issued with a deeper appreciation of a centralized planning approach

and methods of coordinating departmental spheres of action. Unfortunately, there was no immediate effect of the memoranda until the Second World War experience had revealed the disadvantages of illiteracy.

There was a sudden increase in the demand for raw materials from these areas; large numbers of tribesmen were recruited for the armed forces; and local communities were stimulated through 'grow more food' campaigns, increased production targets, organization of civil defence services and other war time measures, to work for the defeat of a distant enemy of whom many of them had previously never heard. It was necessary to explain the essence and purpose of the struggle in vivid and arresting terms; and to do this such instructional media as films, newspaper articles, public address equipment, pamphlets, photographs, and wireless broadcasting were introduced which, although commonplace in the United Kingdom, had been little used in the dependent territories before the outbreak of the war.<sup>25</sup>

The British officials therefore appointed a committee to consider the best approach to the problem of mass education. Their report <u>Mass Education in African Society</u>, which was released in 1944, stated that "the realism of citizenship begins in a small unit where common loyalty and common interests are expressed in daily activities,"<sup>2</sup>

and,

in focussing attention in the community as a unit to be educated, we are aiming at getting people everywhere to be aware of, to understand and take part in, and ultimately to control the economic and social changes which are taking place among them.<sup>27</sup>

Thus 'mass education' did introduce a rather all encompassing and new governmental approach to development. "The publication of this report in 1944 can be taken as a starting point in the evolution of community development as an area of Government policy."<sup>28</sup>

In 1948, the Cambridge Summer Conference on African Administration recommended that a new term, 'Community development', should be adopted

and defined this process as,

A movement designed to promote better living for the whole community with the active participation and communible, on the initiative of the community, but if the statistic is not forthcoming spontaneously, by the statistic techniques for arousing and stimulating it in order secure its active and enthusiastic response. (Community development) embraces all forms of betterment. It includes the whole range

of development activities in the district, whether these are undertaken by securing the adoption of better methods of soil conservation, better methods of farming and better care of livestock; in the field of health by promoting better hygiene, infant and maternity welfare; in the field of education by spreading literacy and adult education as well as by the extension and improvement of schools for children. Community development must make use of the cooperative movement and must be put into 29 effect in the closest association with local government bodies.

In concluding this brief history of the evolution of community development in British experience, it is worthy to note that the British experience, unlike the experience of other countries, was not developed by practice at home, but acquired and tested predominantly in Africa.

#### Community Development as a United Nations-supported Program

From the above brief histories, community development can be considered as an outgrowth of earlier experiments and concepts in 'mass
education' and 'rural reconstruction'. Then after the Second World War, the concept of community development was introduced within the framework of economic aid in low-income countries when the conditions of chronic poverty, illiteracy and disease in the Third World began to press forward for international attention. This technical and economic aid was given in rural areas in particular. After approximately twenty years as a United Nations-supported program, community development has gained recognition as a method for inducing social and economic change in developing countries.<sup>31</sup> In 1953, the United Nations published the well-known study 'Social Progress Through Community Development' which was the first of a series of conferences, seminars and publications devoted to community development.

> Towards an Understanding of Community Development in Developed Countries

The use of community development method, however, has not been restricted to rural situations in developing countries. It spread rapidly throughout the Third World and finally to the more developed parts of the world. The essential elements of community development, such as popular participation, self-help and the building of local leadership, which were generally associated with the efforts of modernization in developing nations, are being adapted in rural and urban areas in the United States, Canada and European countries.

# <u>The Context of Community Development</u> <u>a Generalization</u>

For community development, the aspirations for a better life and efforts to change unfavorable conditions are present in all communities but in different degrees. Although communities may be regarded as similar everywhere from the standpoint that they represent an aggregation of people, there are differences, for instance, in the standard of living, the characteristics of the population, the socio-economic structure, the machinery of government at various levels etc.

23.

In considering the context of community development, this section attempts to keep the analysis, insofar as possible, at a level that is more generally applicable.

The less-developed context. In examining rural areas in developing countries, which are mostly part of traditional or transitional societies, the economic structure is simple and is based on a very simple division of labour and simple technology. These result in low productivity. On the other hand, the literacy rate is low,

health conditions are poor, life expectancy is low; infant mortality is high, and social and welfare services are nominal. People living in these communities are also characterized by strong kinship which form the basic framework of their social organization. Community development efforts are directed largely at raising the standard of living, the level of literacy and so on. Community development is usually

appled as a means to bring about social and economic change in the

communities.

<u>The more-developed context</u>. In highly developed countries, the situations are different. The economic structure of the developed countries, or say, the technological societies, is more complex than the developing countries. The rapid technological progress has resulted in greater economic prosperity, an increasingly complex division of labour, and a considerable movement of the labour force from the primary sector of production to the secondary and tertiary sectors.<sup>32</sup> Certain basic needs become more articulated and new kinds of needs appear-intellectual, aesthetic etc. In addition, the technical progress entails two more important consequences. First, there is an increase in leisure time, and second, the expansion of the provision of formal education and the importance accorded to it. Changes in the social and economic structure have also taken place in the course of time.

Technological developments, on the other hand, have brought about an improvement of communications. This has contributed to a widening of the spheres of life and the promotion of the process of urbanization. As a result, the pace of urban change is so great that "today societies are becoming urbanized faster than urban institutions have been able to adjust."<sup>33</sup>

There is a need to provide employment opportunities for all, and along with that there are problems such as the slum of the city, the traffic congestion, inadequate housing etc. On the personal level, there are problems of unemployment, disabilities, mental illness,

family conflict, alienation, juvenile delinquency and more.

#### B. <u>Community Development in the Context</u> of Developed Countries

The felt needs of a community are the product of its culture, social and economic structure and environmental conditions. This explains why the work of the earlier community development projects in the rural areas of developing societies was directed largely toward raising the standard of living, the level of literacy, and the promotion of economic and social Change.

25

In the developed or highly developed countries, characterized by technological progress, economic prosperity, urbanization, specialization, community development is confronted with a different kind of

context.

In the developed context,

The great change has disrupted the older patterns of social relations, and that disruption in turn has brought about adaptive behavior that is experienced as the structural and personal stresses of urban industrial life.<sup>34</sup>

Community development efforts are directed largely at taking effective action to mitigate the stresses of urban industrial life. In response to the needs of the urban areas, the method has been applied extensively to the creation of conditions leading to a better solution of social

problems, and the prevention of new social problems.

Community development, although evolved as a method of development in a different type of society, has been used by Western

nations for the industrial and agricultural development of their rural areas, and their urban renewal and rehabilitation. The method has also been used for the coordination of the functioning of the large . numbers of organizations in the highly developed countries, and the improvement of the living conditions for some disadvantaged groups--

such as Indians and Eskimos in Canada, and so on.

The urban environment of the highly developed countries,

on the other hand, affects the application of community development of methods. One of the problems of urban community development, as identified by Brokensha and Hodge, is the complex nature of the urban community.

26

It is difficult to initiate community development where there is little sense of community. The sheer size of the rapidly growing cities and heterogenous mass of the urban population are but two of the factors of this complexity.<sup>35</sup>

In other words, in the urban areas, it will be less simple to achieve an identification of the population concerned, or of the total problemgroup partly due to the less manifest sense of belonging to a certain community.

On the other hand, as a result of the quickening pace of industrialization and urbanization in the past decades, there is a widening of scope for the application of community development in urban areas. But du Sautoy suggests that urban community development is likely to be effective "in the educational and social field, rather than in that of construction", for in the cities Uinsecurity, loneliness, and the disadvantages of ignorance are Tikely to be most keenly feit."<sup>36</sup> In fact, community development in urban areas has not placed the same emphasis on improving local economic conditions as did its forerunners in developing countries.

#### Summary

This chapter briefly describes some of the origins of the concept of community development. It is based on the assumption that by examining the objectives of the historical forerunners of community development and the settings in which the activities were undertaken,

the current formulation of community development will better be understood. It further examines the context of community development

understood. It further examines the context of community development is as pertinent to developed and explains why community development is as pertinent to developed.

countries as it is to developing countries. In the next chapter, the conceptual framework of community

development will be reviewed.

#### Footnotes

ly.p. Pande, <u>Yillage Community Projects in India</u>, London, Asia Publishing Co., 1961, p. 15.

<sup>2</sup><u>Ibid</u>., p. 56.

<sup>3</sup>Ibid., p. 50-51.

<sup>4</sup>Ibid., p. 51.

<sup>5</sup>Sudhir Sen, <u>Rabindranath Tagore on Rural Reconstruction</u>, Visva-Bharati, 1943, p. 68.

<sup>6</sup>V.P. Pande, <u>op</u>. <u>cit</u>., p. 89.

OM :

<sup>7</sup><u>Ibid</u>., p. 90.

٩.

<sup>8</sup>Ibid, p. 91.

9<sub>Ibid., p. 91.</sub>

<sup>10</sup>S. Dasgupta, <u>A Poet and a Plan</u>, Calcutta, Thacker Spink, 1963, p. 28.

11<sub>V.P.</sub> Pande, <u>op</u>: <u>cit</u>., p. 109.

<sup>12</sup>M.K. Gandhi, <u>Constructive Programme - Its Meaning and Place</u>; Ahmedabad, Navjivan Publishing House, Reprint, 1948, March, pp. 8-30.

<sup>13</sup>V.P. Fande, <u>op</u>. <u>cit</u>., p. 119.

14<sub>Ibid</sub>.

15 Ibid-

16Ibid.

17 Ibid.

18<sub>H</sub>. Tinker, <u>Ballot Box and Bayonets</u>, London, Oxford University Press, 1964, pp. 105-106. <sup>19</sup>J.V. Abueva, <u>Focus on the Barrio</u>, Manila, Institute of Public Administration, University of the Phillipines, 1959, pp. 128-129. ු

<sup>20</sup>Ibid. •

<sup>21</sup>H. Tinker, <u>op</u>. <u>cit</u>., p. 106.

<sup>22</sup>Social Conditions in Non-Self-Govertting Territories, New York, 1956, p. 13.

<sup>23</sup>Great Britain, Reference Division, Central Office of Information, <u>Community Development in the United Kingdom and Dependencies</u>, London, Overseas Information Services, 1958, p. 3.

<sup>24</sup>Ibid.

25<sub>Ibid</sub>

<sup>26</sup>Great Britain, Colonial Office, <u>Community Development: A Handbook</u>, London, Her Majesty's Stationery Office, 1958, p. 1 (cited from <u>Mass</u> <u>Education in African Society</u>, 1944).

27<u>Ibid</u>.

<sup>28</sup>Ibid.

<sup>29</sup>Ibid.

<sup>30</sup>David Brokensha and Peter Hodge, <u>Community Development</u>, <u>An Interpretation</u>, San Francisco, Chandler Publishing Co., 1969, pp. 34.

<sup>31</sup>Julia J. Henderson, "The Evolution of Community Development" <u>Urban and Rural Community Development</u>, Proceedings of the XIth International Conference of Social Work, 1962, pp. 65-73.

<sup>32</sup>Colin Clark, <u>The Conditions of Economic Progress</u>, London, MacMillan, 1957.

<sup>33</sup>Marshall, B. Clinard, <u>Slums and Community Development: experiments</u> in self-help, New York, The Free Press, 1966, p. 116.

<sup>34</sup>Roland J. Warren, "The Context of Community Development," in Lee J. Cary (Ed.), <u>Community Development as a Process</u>, Columbia, University of Missouri Press, 1970, p. 43. <sup>35</sup>David Brokensha and Peter Hodge, <u>op</u>. <u>cit</u>., pp. 122-123.

.30

0

<sup>36</sup>Peter du Sautoy, <u>The Organization of a Community Development</u> <u>Programme</u>, London, Oxford University Press, 1962, p. 36.

÷

#### CHAPTER III

## COMMUNITY DEVELOPMENT: THE CONCEPTUAL FRAMEWORK

The purpose of this chapter is to review related literature on the definitions of community development, the concept of community, the principles and the strategy of community development, and the concept of citizen participation. On the basis of the discussion, an analytical framework by which to examine the various aspects of the Preventive Social Services projects will be developed.

## Defining Community Development

The concept of community development, as mentioned earlier, has sprung from many sources. It earned its recognition through much trial and error by a large number of workers in the field. Since community development has been used to solve various kinds of problems, there are countless attempts to define community development. In different parts of the world today community development activities, ranging from integrated national programs to individual local projects, varying in form and approach, define community development in many ways. There is no complete agreement on a basic definition or what is precisely meant by the term 'community development'. Some confusion has arisen from the fact that community development is sometimes referred to as a 'process', sometimes as a 'method', sometimes as a 'program', and at others as a 'movement'. The word 'process' is usually used to designate a progression of changes, e.g. from state of minimum to one of maximum cooperation, from state where few participate to one where many participate, etc. Emphasis is on what happens to people socially and psychologically. When community development is referred to as a 'method' it is a means of problem-solving. Emphasis is on achieving certain objectives. When the word 'program' is used, it is referred to the objectives, procedures, activities and administrative structure. As a 'movement', community development becomes something more than a mere program. It is an emotional dynamics, as a philosophic and not a scientific concept.<sup>2</sup>

On the other hand, some writers limit the definition of community development to the educational aspect of the process, which others stress the importance of social action. C.A.S. Hynam, in his paper "Community Development, An Example of Conceptual Confusion," has illustrated the recent state of affairs in community development with the following definitions:

1) By community development I mean the movement of a population toward higher material levels of living and toward the institutional patterns associated with urban-industrial society, whether of the capitalist or the communist variation.

2) Community Development is an educational-motivational process designed to create conditions favorable to economic and social change, if possible on the initiative of the community, but if this initiative is not forthcoming spontaneously, then techniques for arousing and stimulating it in order to secure fullest participation of the community must be utilized.

He:says, "Definition 1 is referring to community development in the sense of total resources while definition 2 is referring to it as a human relationship process. They are two widely different concepts."<sup>5</sup> It must be admitted that some confusion is due to the fact that community development is practised by different professions, such as Social work, Adult education, Agricultural extensions and so on. It is not surprising therefore that the many definitions of community development are varied.

One way to clarify such a conceptual confusion and to understand the term 'community development' is to look at some of the common characteristics. From an analysis of literature in the field and observations of programs by A.F. Wileden there are about five characteristics commonly accepted as describing the community development process. These include the following:

- 1) A philosophy, in accordance with our democratic philosophy, that local people, working in their local group situations, have not only the right but also the responsibility to choose their own objectives and to make their own decisions . . .
- 2) A concern for changing the local situation with a view to improving conditions in the so-called community . . .
- 3) A belief that lasting progress can be achieved only through the development of local understanding, local initiative, and local self-help, with as broad local participation as possible, ....
- 4) An emphasis on the balanced development of all the resources, physical and human, in the community or area under consideration . . .
- 5) An assumption that outside resources, including counsel and technical assistance, must be available to local community situations . . . .

On the basis of the above common characteristics and the

earlier discussion on community development in developed countries, the

following definitions are selected as the frame of reference for this

#### study.

The first definition is from Biddle and Biddle who feel community development must deal with the effects of change:

Community development is a social process by which human beings can become more competent to live with and gain some control over local aspects of a frustrating and changing world.<sup>7</sup>

The second definition is from Arthur Dunham who regards community

development

. . as organized efforts of people to improve the conditions of community life and the capacity of the people for participation, self-direction, and integrated effort in community affairs. Community development is accomplished by people. It seeks to work through self-help, voluntary participation, and cooperation of people in the community, but usually with technical assistance from government or voluntary organizations.<sup>8</sup>

Thirdly, the Canadian Welfare Council defined community development as

A process aimed at promoting citizen participation in social affairs, developing people's awareness of problems, enabling them to define their needs in relation to the total environment, making possible that enlightened choice among various options and channelling their results into effective action for social change.<sup>9</sup>

There are some common elements in these definitions: community development must deal with social change; the focus is on the common needs of the community; must work through self-help, voluntary participation, and cooperation of people in the community; usually with assistance of some sort from the government or voluntary organizations. These definitions also embody both the task-goal and process-goal of community development. The task-goal is concerned with attempts to meet specific needs or solve particular problems. The process-goal refers to the learning process of the community in working towards self-help, cooperation, and participation.

It must be stressed that both the task and process goals are important because community development is not just an end in itself, but also a means to an end. As a strategy of social change, the goal of community development is to determine "the least cost approach to change."<sup>10</sup> Given the context of the developing societies, it is unbelievable, that the governments of these nations will stress the process goal to the exclusion of the task goal.

## The Concept of Community

The term community development gives us the basic unit of the action, namely, the community which also means different things to different people. Community may refer to "a) geographical area; b) to social, economic, cultural conditions of the people; and c) to the people themselves--the group of persons more or less related one to another in the achievement of a common endervour or sharing their common hopes, desires, etc."<sup>11</sup> However, based on Hillery's analysis of ninety-four definitions of community there are three elements enter into most sociological definitions of community including (1) geographic area, (2) social interactions, and (3) common tie or ties.<sup>12</sup>

With this as a background, community, in this study, is to be understood in terms of Ross's two categories--one defined by reference to common location and the other by common interest.

1) Community may refer to all the people in a specific geographic area i.e. in a village, a town, a city, a neighbourhood or a district in the city . . . the geographic community. Community is used here also to include groups of people who share some common interest and junction such as welfare, agriculture, education, religion--the functional community.<sup>13</sup>

In early community development practices, the unit of action was basically the local community--the rural village--where face-to-face interaction was most possible. In modern urban areas, characterized by high mobility and corresponding decline in community cohesion and autonomy, the geographic definition of community is not always useful. The focus of action in community development is then based upon the local community--the neighbourhood, the city, the town etc.--as well as the community of common interest.

Using Ross's definitions, it appears that community development can take place in at least, three kinds of community situations in the city: a functional community drawing from the whole city or several areas in the city; a small geographic community that would allow face-toface interaction which characterizes the rural village; or a larger geographic community within which community development would be carried on with the participation of members from functional communities. These functional communities may or may not unite to deal-with problems they perceive to have in common.

The Principles of Community Development.

Community development; as an approach or process, is conceived in terms of certain principles and concepts for working with the community. In the literature, several different lists of principles have been compiled as guidelines for effective community development practice. From the several lists available, there are three widely accepted lists: Arthur Dunham's eighteen suggested principles which are based on 142 statements, DiFranco's collection of principles and guides, and the principles outlined in the United Nations Report. All the principles from these three sources are listed in Table 1. As most of the principles associated with community development were formulated largely by practitioners from their experiences in rural areas of developing societies, a few of them are not particularly related to the urban context of developed countries. In other words, the principles underlying the community development process may differ in accordance with the kind of society in which the process is being carried on. The above comments, however, do not suggest that the principles being applied in the rural areas cannot be useful as the guidelines for effective community development practices in urban communities of developed countries. In fact, most of the principles are found applicable to the context of industrialized society. Only a few principles must be modified. The author therefore will use some core principles which are relevant to the urban setting and on the other hand, are considered as necessary for community development in action.

These principles include:

1. Community development should be based upon the economic and social needs of the community.

2. Community development should encourage citizen participation at all t levels of operation.

3. Community development should encourage cooperation between individuals

# Principles of Community Development

· . . .

,

- :\*

?

• .

Table 1

A. Dunham <sup>14</sup>	Di Franco <sup>15</sup>	United Nations <sup>16</sup>
1. based upon economic and social needs of community	<ol> <li>should be based on conditions that exist (local, re- gional, national)</li> </ol>	<pre>1. improve the level of     living</pre>
2. Concerned with total population	2. involve people in act <sup>2</sup> ions which promote their welfare	2. self-help
3. Should start with felt needs	3. develop programs gradually	3. local communities basic units, also groups with special interests
4. should be demo- cratic	4. aim basically at people's interests and needs	4. within the framework of a national plan
5. local community the basic unit	5. use democratic methods	5. an educational and organizational process
6. voluntary coopera- tion	6. use local leaders	6. reorientation of existing institutions or the creation of new types of institu- tions
7. self-help	7. use existing agencies	7. changing attitudes
8. development of attitudes	8. utilized trained specialists	8. respond to the felt- needs
9. an educational process	9. work with all family members	9. use integrated approach
10. understanding of culture	10. evaluate continually	10. require technical assistance

•

38

•

11. use community 11. program should be well organized, integrated, practical, realistic flexible

12. part of a national 12. keep in with program of economic and social development

13. leadership and the 13. help people role of the employed worker is to provide professional assistance and indirect rather than official leadership

14. training of personnel required

15. importance of local voluntary leadership 🖉

15. work through understanding culture

approach

federal and provincial

recognize their

policies

needs

16. democratic government and participation on all levels

17. utilization of government and voluntary agencies -----18. the community should have a Large degree of freedom and

autonomy within the framework of the nation

11. efforts of people united with government

12. utilize local organizations

> encourage ' participation

14. should encourage program flexibility and overall policy should be fixed, but implementation should be flexible

16. working with all classes of the society 👘

and groups, and the government.

interests.

4. Community development should aim at the development of local leader-

5. Local communities are basic units of action, also groups with special

As principles of community development were formulated from many experiences in the field, the core principles are presumed to be indicators of successful community development.

The Strategy of Community Development

It is very difficult to define a community development strategy use an effective strategy depends in large part on a number of Quartors related particularly to the point of development of a community. Of course, there are many strategy-models of community development such as power-conflict model, information model, social action models problemsolving model etc. but most of the times, especially in urban communities, the mode of intervention involves a mixed strategy.

Regardless of the model or a combination of models used in any situation, a community-development strategy is people oriented. The community-development strategy, on the other hand; is very different from the service-oriented strategy. As Francis J. Bregha comments,

This is what distinguishes community-development strategies from more service-oriented strategies. The provision for and the delivery of services is only and perhaps the least important aspect of community development. Its main thrust and principle raison d'etre as a method of intervention is to transform the causes and conditions shaping the quality of life in a society so that as few people as possible would depend on any kind of service. 17 40

He further suggested that a new type of planning of less crisis-oriented, 4 Q · much more directed towards prevention and towards comprehensive programs

is required. And,

Such planning would also represent an on-going process, with high Tevels of citizen participation, which would facilitate the fast feedback needed for constant readjustment and up-dating of policies and plans of action.

The above comments indicate that community development is not

only a question of planning for social or economic well-being, it is also a question of how the planning takes place. Recently, newer concepts <sup>6</sup> in connection with the five issues which any planning must confront have

arisen. These concepts have to do with:

- Consensus and dissensus as to the course of action appropriate 1). for the community.
- 2) Directiveness or nondirectiveness of the role of the change agent. The strategy used may be that of helping citizen determine their own goals or of persuading groups to accept the proposed goals.
- Process orientation or task orientation in community develop-3) -5 ment.
- Inclusiveness of scope of program.concerns which may be a 4) wide spectrum of community problems or relatively narrow . substance areas:
- Inclusiveness of participation that efforts may be made at 5) securing participation by the total community, or be confined to interested groups and those with decision-making prerogative.<sup>19</sup>

Using the above concepts as the basis of analysis, the 

traditional approach to community development was oriented towards

- (1) seeking consensus, (2) a nondirective stance on the part of the
- change agent, (3) emphasizing the process of decision-making and change,
- (4) the inclusiveness in sample, with the total community and its well-
- being as the focus of concern, and (5) the broadest possible participation in planning and implementation by the entire population.20

The newer concepts, however, have moved away from this conventional position. Among these concepts, the author finds Warren's discussion of changes in the five aspects of community development most interesting. Warren argues that acting only on consensus means no

:42

action. He points out that

. there is no single set of community goals, but that various individuals and organizations have different goals, which often conflict with each other; and that it is precisely the most important and pressing issues about people are divided, rather than united. To act only in consensus is to immobilize oneself.

Likewise, he contends that training for community competence itself is at least partially beside the point, and that an organization dealing with community development should limit its activities to a few subject areas. Since the various concerns of the community are so comprehensive, an organization must most likely limit itself to one or a few subject areas rather than spreading itself thinly over a number of areas in which much stronger groups are likely to be involved, with higher stakes in the outcome.<sup>22</sup> Finally, he feels that total citizen participation is

impossible. He says,

. . . the idea of trying to approach total citizen participation on all important issues is mathematically impossible in all but the smallest communities. The problem of participation is not so simple as one of "involving the whole community" or "letting the people decide." Hence, choices must be made as to who is to participate and in what capacities.<sup>23</sup>

The author agrees, to some extent, with Warren's concepts. It is time that changes in the nature of today's communities, especially the complexity of urban areas in developed countries, affect the effective application of the 'puristic' approach to community development. For example, in the city, the population is more heterogeneous with various interests, to act only on consensus of the total population of all groups is difficult. In early community development projects, gr community development work in underdeveloped countries today, the basic unit of action was the rural village which was small enough to allow face-fo-face interaction, and the main concern of the people was in the satisfaction of basic needs. The urgency to find solutions to the various problems was so great that the population were quick to appreciate the importance of the issues. It was much easier to act on o

consensus of the total population of the community.

On the other hand, because of the size and the differentiated social structure of the urban population, total officen participation on all important issues is much more difficult to achieve. Some other forms of cittizen participation are likely to take place.

With reference to Warren's emphasis on the task orientation, the author of this study has a different point of view. Although the importance of the task-goal must be recognized, equal emphasis should also be placed on the process-goal by which people can become more competent to gain some control over local aspects of a changing world.

Such a process may foster favourable conditions for effective community development in the future.

## Citizen Participation

In the definitions of community-development, the promotion of citizen particing the in social affairs, or in the efforts to change. their situation is stressed by most authors. Despite the current popularity of citizen participation, there are conflicting viewpoints

of the value of citizen involvement. Some people criticize that

citizen participation will require more time and thus money to arrive at a decision. Another argument is that consumers who are unable to control their own lives have neither the energy nor capacity to take part in the

process of problem-solving.

, O

•

0.2

On the other hand, there are many social scientists who are in favour of citizen participation S. Edgar and J.C. Cahn said that · . • citizen participation is costly, time-consuming, and frustrating, but it cannot be dispensed for the three following reasons:

1.º The value of its acknowledgement and promotion of dignity and self-sufficiency; The value of its by-product in the utilization of untapped mangower resources;

The value of the knowledge it affords: the criticism, corrective insight, and continuing validation of efforts which are best informed hunches on how best to give ° v° ; c content to broad national goals which can be attained only through the perilous process of trial and error, experimentation and assessment. 24

In a society with a dynamic nature, the importance of citizen. participation becomes even clearer. In a dynamic society, the needs change fairly rapidly. If the general public participates in the work, the needs can be identified and formulated much more positively. As

્યત્ર દ્વાર્ટ દેવ welfare is so closely tied up with everyday life, it requires citizens to be closely involved in the various stages of planning and action.

Dunham, in discussing "How to involve citizens in 'target 16 .

areas' in programs concerned with social change and social welfare," 0. 0. 0 4 00 ° 0\* suggests the following ways of working toward citizen involvement in 0 0 0

#### the planning process:

- 1. Substantial <u>representation of residents</u> of target areas on <u>directive boards or committees</u> with real decision-making power is one of the most vital approaches.
  - 2. Another possibility is the use of <u>advisory bodies</u> drawn wholly or largely from the target areas. These of course lack the authority to make binding decisions.
  - 3. <u>Town meetings</u>, neighbourhood meetings, <u>and hearings</u>, are <u>ad</u> <u>hoc</u> occasions when representatives of the planning body and residents of the target area may meet and engage in at least a limited dialogue.
    - . . may use mass media . . to help <u>keep citizen informed</u> of developments. They may go>beyond this and <u>solicit sugges</u>-<u>tions</u> from target areas through various channels of communications.
  - 5. Leadership training may have an indirect effect in promoting citizen participation.
  - 6. <u>Volunteers</u> may be recruited from target areas to assist with many types of programs.
  - 7. Certain residents from target areas may be employed as indigenous nonprofessional aides.
  - 8. Special efforts may be made to see that target areas are adequately represented in community-wide events.
  - 9. <u>Neighbourhood service centers</u>, . . . , may service to make agency services more accessible and less remote.<sup>25</sup>

Some of these suggested approaches to citizen involvement, however, are considered as 'token' representation rather than 'substantial' representation. As previously noted, citizen participation in the community development sense implies a process of decision-making; as opposed to token representation. In fact, most of the time, citizen participation is viewed as a categorical term for citizen power.

Arnstein, based on three different programs, has attempted to cut across the euphemisms and the rhetoric of citizen participation and has proposed an eight-rung typology of citizen participation<sup>26</sup> which is illustrated

in Figure 1.

#### Figure<sub>1</sub>

The Eight-Rung Ladder of Citizen Participation

1. 2. 3.	Citizen Control Delegated Power Partnership	levels of Citizen Power
5.	Placation Consultation Informing	degree of Tokenism
7.	Therapy Manipulation	Non-Participation

In Figure 1, the bottom rungs of the ladder, are considered as non-participation because the real objective is to 'educate' and 'cure' the participants rather than involving citizens in planning or conducting programs. These two rungs signify the distortion of participation. Information and Consultation are considered as forms of

tokenism because emphasis is placed on the one-way flow of communication. There is no assurance that citizens' views are considered. At the level of Placation, some degree of influence begins but it is still considered

as tokenism. Some citizens are allowed to advice but the decisionmaking power is retained by others. Levels of Citizen Power with increasing influence and decision-making power are found further up the ladder. In terms of Partnership, planning and decision-making responsibilities are shared. Delegated Power, which is one level above, assures citizen with balance of power. Finally, citizens are given the full charge of

the program in the highest level--Citizen Control.

## The Analytical Framework

For the purpose of the study, an analytical framework will be developed from the review of literature in the preceding sections. Due to the fact that many elements of community development are reviewed in this chapter, only some of the more prominent ones will be included for the purpose of analysis. These elements are briefly described within the following dimensions of the analytical framework:

## A. Initiation of the Projects

As the concept of local involvement is emphasized in the principles of community development, the initiation of the projects will be analyzed in terms of the extent of local participation on the part of the community (people and groups).

# B: Operationalization of the Projects

The operationalization of the projects will be studied under four categories: 1) objectives, 2) clientele, 3) methods of work, and 4) organizational structure.

1. <u>Objectives</u>. First of all, the needs inherent in the objectives of the Preventive Social Services projects in this study will be analyzed. Then the preventive versus gehabilative orientation of the objectives of the projects will be analyzed. In the discussion on the strategy of community development, Francis J. Bregha has suggested that a new type of planning, much directed towards prevention is required for community development. Finally, the extent to which the objectives of the projects are developed by the people of the community based on their interests and needs will be examined. The reason is that one of the crucial elements of community development is the concept of felt-needs indicating the desirability for the objectives to be determined by the local people.

2. <u>Clientele</u>. The target population of the projects will be analyzed in terms of Ross's two categories-the geographic community . and the functional community. Community development, according to its principles, works with groups and local communities rather than with individuals.

3. <u>Methods of work</u>. In studying the methods of work of the projects, the process-orientation versus task-orientation of the planning strategy will be analyzed. Eurthermore, the scope of the projects will be examined. Both elements of analysis are selected on the basis of the discussion on the planning strategy of community development. Then the level of volunteer involvement in the delivery of services will be studied because the concept of developing and training of volunteers. is stressed in the principles of community development.

4. <u>Organizational structure</u>. According to the principles of community development, it is imperative that citizen involvement at all levels of operation be encouraged and local leadership be developed. Citizen involvement at the level of policy decision-making will then be examined in terms of prostein's eight-rung typology of citizen participation.

## Summary

This chapter provides a review of literature on various major concepts of community development. It also briefly describes the analytical framework which is formulated from the review of literature. Each component of the analytical framework will be explicated later in Chapter VI in which all the Preventive Social Services projects in this study will be analyzed.

The next chapter will focus on the description of the Preventive Social Services Program in Alberta.

#### Footnotes

50

<sup>1</sup>C.A.S. Hynam, "Community Development, An Example of Conceptual Confusion," B.Y. Card (Ed.) <u>Perspectives on Regions and Regionalism</u>, Western Association of Sociology and Anthropology Proceedings, 1968, p. 193.

<sup>2</sup>Irwin R. Sanders, "Theories of Community Development," <u>Rural</u> <u>Socialogy</u>, Vol. 23, no. 1-4, 1958, p. 5.

<sup>3</sup>Arthur K. Davis, "Rejoindus," <u>Human Organization</u>, Vol. 27, no. 1, Spring, 1968, p. 89.

<sup>4</sup><u>Community Development in Alberta</u>, Ottawa, Special Planning Secretariat, Privy Council Office, 1965, p. 2.

<sup>5</sup>C.A.S. Hynam, <u>op. cit</u>.

<sup>6</sup>Arthur F. Wileden, <u>Community Development</u>, <u>The Dynamics of Planned</u> <u>Change</u>, N.J., Totowa, The Bedminister Press, 1970, pp. 79-80.

<sup>7</sup>W.W. Biddle and L.J. Biddle, <u>The Community Development Process</u>, New York, Holt Rinehart & Winston, Inc., 1965, p. 28.

<sup>8</sup>Arthur Dunham, <u>The New Community Organization</u>, New York, Thomas Y. Crowell Co., 1970, p. 140.

<sup>9</sup>Freeman H. Compton, "Community Development Theory and Practice," in James A. Draper (Ed.) <u>Citizen Participation: Canada</u>, Toronto, New Press, 1971, p. 383.

10 Jim Lotz, "Thoughts on Community Development," <u>Adult Leadership</u> 1973, September, p. 79.

ll Brazilian Committee on the International Conference of Social Work, <u>Proceedings of the Eleventh International Conference on Social Work</u>, Brazil, Petropolis, 1962, p. 273.

<sup>12</sup>George A. Hillery, Jr., "Definitions of Community: Areas of Agreement," <u>Rural Sociology</u>, 20, 1955, June, p. 118.

13<sub>Murray</sub> G. Ross, <u>Community Organization</u>, <u>Theory</u>, <u>Principles and</u> <u>Practice</u>, New York, Harper & Row, 1967, pp. 41-42. Arthur Dunham, "Some Principles of Community Development," <u>Inter-</u> national Review of Community Development, 11, pp. 141-181.

<sup>15</sup>Joseph Di, Francois, <u>A Collection of Principles and Guides</u>, New York, New York State College of Agriculture at Cornell University, 1958.

<sup>16</sup> United Nations, <u>Popular Participation in Development</u>: <u>Emerging</u> <u>Trends in Community Development</u>, New York, United Nations, 1971, pp. 6-7.

<sup>17</sup>Francis J. Bregha, "Community Development in Canada." Problems and Strategies," in James A. Draper (Ed.), <u>Citizen Participation</u>: <u>Canada, A Book of Readings</u>, Toronto, New Press, **1**971, p. 79.

18 <u>Ibid</u>.

<sup>19</sup>Roland L. Warren, "The Theory and Practice in Community Development;" <u>American Journal of Agricultural Economies</u>, 50, 5, Dec., 1968, pp. 1234-1235.

<sup>20</sup><u>Ibid</u>., p. 1235.

<sup>21</sup>Ibid.

<sup>22</sup><u>Ibid</u>., p. 1236.

<sup>23</sup><u>Ibid</u>.

<sup>24</sup>S. Edgar and Jean C. Cahn, "Citizen Participation," in Hans B.C. Spiegel (Ed.), <u>Citizen Participation in Urban Development, Volume I -</u> <u>Concept and Issues</u>, Washington, D.C., NTL Institute for Applied Behavioral Science, 1968, p. 221.

<sup>25</sup>Arthur Dunham, The New Community Organization, op. cit., pp. 330-331.

<sup>26</sup>E.S. Cahn, and B.A. Bassett (Eds.), <u>Citizen Participation: Effecting</u> <u>Community Change</u>, New York, Praeger Publishers, 1971, pp. 61-91.

51

#### CHAPTER IV

# PREVENTIVE SOCIAL SERVICES: CONCEPT AND ORGANIZATION

The basic objective of this chapter is to provide some general information on the Preventive Social Services Program in Alberta before proceeding to the descriptions and analyses of the Preventive Social Services at the project level in the following chapters. ('Program' designates the entire Preventive Social Services Program. 'Project' designates the individual projects within the overall Preventive Social Services Program.) In this Chapter, the objectives, the organization and administration, and other major concept Services Program in Alberta will be described.

Preventive Social Services (FSS) is an innovative concept, which attempts to introduce new service objectives into social services and a new approach to improve the social welfare of the community. It embodies a new way of thinking about the causes of social problems and new approaches to meeting social needs. The idea of Preventive Social Services was initiated by the Department of Public Welfare (now, Department of Health and Social Development) of the Government of Alberta in the early sixties. On July 1st, 1966, the Preventive Social Service

Act became effective in Alberta.

52

#### The Principles

There were three fundamental principles built into the original

53.

plan of the program. They are as follows:

- 1) Local planning and decision-making; A salient premise based on the knowledge that communities are unique--not only in terms of the perception of need, but also of resources and their interaction, it challenged municipal governments to accept the responsibility of social concern, coupled with their traditional roles tied to physical and economic commitments.
- 2) The provision of funds to establish appropriate programs: Without a realistic input of money, many municipalities could "scarcely afford to contemplate the new dimension of service. A formula of up to 80 per cent provincial subsidy was agreed on by cabinet, with municipal government to provide 20 per cent. The public funding was to be applied to deficit costs of any program or project; that is, after alternate sources of revenue had been fully explored and applied. Provision was made for essential capital costs in select cases.

3) Orientation training and liaison services to be provided by the province:

In view of prevention's elusiveness a concept related to social service, municipal staff were offered assistance in realizing the potential of the new program. Pertinent formal training available was conceivably related to community development and community organization.

The Objectives

The objectives of Preventing Social Services are not clearly

stated in the documents, but they are contained in the following concept which is outlined in the <u>Administrative and Policy Outline</u>:

Preventive Social Service is designed to develop community awareness and resources, to strengthen and preserve human initiative and to preclude individual and family breakdown. It is any activity of which all members of a community can avail themselves for the enrichment of their physical, mental and social well-being The emphasis on the local community's responsibility rests on the belief that most communities, like individuals, have the desire, energy and resources for self-improvement. Many services dedicated to this task now exist. It is the aim of the Preventive Social Service Program to assist communities in innovative planning to meet social needs and where necessary to aid financially.

To bridge the gap between community resources and problems, opportunities must be found for group-sharing of experience and the development of a sense of contribution by everyone. Positive services designed to involve the community as a whole are able to strengthen and presume human initiative ' and independence for individuals and families in our modern society.<sup>2</sup>

In brief, the objectives of Preventive Social Services are: to develop community awareness and resources to work toward more useful and satisfying lives; and to preclude the emergence of social problems and social breakdown by assisting communities in innovative planning to meet their social needs.

The Organization and Administration

• Under the terms of the Preventive Social Service Act the Minister of Health and Social Development is empowered to enter into an agreement with a municipality to establish, administer, and operate preventive social services projects. In rural areas, county and municipal district boundaries are used in determining the geographical size of a particular administrative area. In other words, in this program, there are three levels of administration: the Department of Health and Social Development of the Government of Alberta, the municipal level of government, and the project. The latter may be either a function of, the municipal government; a voluntary citizen group, or a private

94

Q

agency. In this chapter, only the organizational structure at the provincial and municipal level will be presented. The organizational structure of individual projects will be described respectively in the following chapter.

On the provincial level, the administrative structure of Preventive Social Services is shown in the following chart:

> Figure 2 The Administrative Structure of Preventive Social Services--Provincial



(Adapted from the organizational chart of the Preventive Social Services Branch, Department of Health and Social Development, 1975).

On the municipal or local level, the organizational structure

varies in different municipalities or rural communities. As an example,

the organizational structure of Preventive Social Services in the City of



tet in the following diagram:

Edmonton is illy

56

(Adapted from the organizational chart of the Social Service Department, Alberta, Edmonton, 1975).

In figure 3, the Social Services Advisory Committee is composed of one alderman appointed by the City Council, one representative from the Social Planning Council of Edmonton, one representative from the United Way, four persons with expertise in the areas of health, education, recreation, and housing, three citizens-at-large, and the superintendent of the Social Service Department.

## In rufal areas, the administrative structure is usually

organized in the following form:

## Figure 4

The Generalized Administrative Structure of Preventive Social Services--Rural



(Adapted from <u>Administrative and Policy Guide to the Preventive Social</u> <u>Service Program</u>, Government of the Province of Alberta, Department of Pupplic Service, Social Planning and Development Branch, 1966, p. 9.)

#### A. The Read of Province

The role of the Province is to provide orientation, training, and consultation of an on-going nature to municipalities and local communities to assist directors, boards or advisory committees in the establishment, development and implementation of the program; and to maintain liaison with other government departments. The province will only consider those project applications approved by the local authority. In carrying out the on-going role of the Province, the Preventive Social Services Branch of the Department of Health and Social Development
#### is responsible for:

- a) Assisting the Municipality with the assessment and selection
- of a Director. b) Continuing consultation and liaison with the Director and Committee or Board."
- c) Receiving from the municipality project applications with support material and budgets, and recommending to the Minister approval, rejection or deferment with a view to further discussion
  - with the Municipality.
  - d) Continuing evaluation of projects.
  - e) Reporting on the total program and making recommendations to - - the Minister. 1
  - Stimulating new, imaginative and alternative ways to achieve £)
  - the objectives of prevention. Drafting an annual departmental budget of expenditures for the
  - **g**). Preventive Social Service Program and developing an annual report.3

## The Role of the Municipality

The Preventive Social Services Act reads:

#### A Municipality. 3.

• •

- May provide for the establishment, administration and . a) operation of preventive social service programs within : • . the municipality.
- b) May enter into agreements with other municipalities to provide for the establishment, administration and operation of joint preventive social service programs, . and

. . . . . . 

- May appropriate money for the establishment, administration a) ( operations of preventive social service programs.
  - (R.S.A. 1970, C. 282, s. 3)<sup>4</sup> . . . . . . . .

In addition, under the terms of the Preventive Social Services

Act, the responsibilities of the municipalities are: to initiate,

develop and administer social services of preventive nature, and to

provide consultation to groups and organizations in establishing

approved services within the area; b) to report to the Province

annually, or upon request, regarding the progress of selected projects; · · · · · · · · c) to consider the proposals and applications for the expansion or

development of specific projects. Projects approved at the local level will be forwarded to the Province for consideration. 59

## C. The Role of the Local Director

When a municipality decides to make an agreement with the province to establish. Preventive Social Services, the municipality must hire a full-time local director for organizing local projects. The local director is not an employee of the Provincial Government. He or she works and is responsible to the community which can be a municipality, or an amalgamation of municipalities depending on the boundary of the particular administrative area. The role of the local director will be one of community organization and planning rather than one of case-work service to individuals and families. Specific responsibilities of the local director as defined in the <u>Preventive</u> <u>Social Services Administrative and Policy Outline</u>, include the -

following:

- 1) To make an initial survey of the social resources and problems in his area.
  - 2) To work with the Preventive Social Services Advisory Committee or Board in reviewing community needs, problems, aspirations,
- and resources. 3) To inform and motivate the community to action on specific matters of social concern.
- .4) To assist the community in selecting preventive projects for attention and interest appropriate persons in implementing
- 5) To involve resource persons, agencies, departments and concerned citizens in the planning and development of the Preventive Social Services Program.
- 6) To review with the Committee or Board project applications
   received and to recommend approval, deferment, rejection, or planning alternatives.
- 7) To be involved in a continuing evaluation of all approved projects.

8) To assume responsibility of liaison, cooperation, and co-ordination of activities with related fields such as health, recreation, education, welfare, and commerce.
9) To understand his own role in terms of functions and approach,

consistent with the purpose and philosophy of the program. 10) To understand the relationship which should exist between his role and that of specific project directors and agency. heads.

It) To be responsible for using the consultative services of the Provincial departments in the best interest of the community, the taxpayer and the fulfilment of his own role.

12) To read and study constantly for the utmost benefit of himself
 in his Director's role and of the community he serves.<sup>5</sup>

In other words, his job is one of innovating, developing and assessing community services in conjunction with the local citizens. It should be stressed that the responsibility both for initiating preventive social services and for carrying out the projects rests with

preventive social services and for carrying out the projects total the local community.

· Cost-sharing

In major cities, having the highest density of population, the municipal government becomes the appropriate local authority for administering the funds for local projects. In rural areas, an amalgamation of municipalities, including towns and villages, may share cost and services as one unit. One authority only will be selected to represent the amalgamated area for administering the funds for local

projects.

As the Preventive Social Service Act provides the municipalities may claim re-imburgement from the Province relating to approved project budgets, the Province undertakes to cost-share with the municipality official authority any approved project on an 80 per cent - 20 per cent basis. contribute at least twenty per cent of the cost of each funded project. The Province will share the cost with the Municipality only up to an amount of eighty per cent. This cost-sharing includes the Preventive Social Service director and administrative staff salaries and travel expenses.<sup>6</sup>

#### Program Development and Selection

During its first year of operation, seven municipalities and local authorities composed of amalgamated units have completed the ministerial agreement required for participation in the program. In 1974, over thirty agreements existed, with numerous new areas demonstrating positive interest in participation. In the last few years that the program has been operative participating communities have established the following types of projects:

- Family Life Education

- Parent-child development

Day Care Service

- Drop-in centre for youth and senior citizens

- Family planning clinic aimed at all segments of the population - Counselling service

- Homemaker service of both an emergency and educational kind

- Multi-service centre

- Community, information service

- Volunteer bureau

Community development resource

- Information and referral service

- Club for senior citizens

- etc.

As one looks at the list of projects which are currently funded in Alberta, one probably would notice that a number of projects are repeated from one community to another. The staff members of the Preventive Social Services Branch, Department of Health and Social Development explained that "this is because several communities have become convinced of their value rather than because the Provincial Government has set these as "priority programs". In fact, the criteria of project selection are flexible, depending on local imagination and needs. The Preventive Social Services Act does not attempt to enforce precise definition, but provides support for those projects judged to be preventive in nature.<sup>7</sup>

Funding may be available to groups in the community who wish to develop new services or expand existing services which are preventive in nature. According to <u>A Guide to Application</u>, "priority will be given to programs which are primarily preventive in nature."<sup>8</sup> Moreover,

Programs must be equally available to all persons in the community. Programs sponsored by ethnic or Sectarian groups are therefore not within the terms of reference of the Preventive Social Services Program. Projects which duplicate existing services will usually not be supported.

Experimentation and innovative approaches to social development are also encouraged. However, it is difficult to determine the priorities in selecting and funding projects due to the absence of measures which define the relative importance of various social services, e.g. which decide between the merits of a community mental health centre as opposed to a family life education program. The result seems to be frequently a reliance on subjective judgements, or on relative cost, as a criterion.

63

#### What Is Prevention?

It has been customary to think of prevention in terms of three points of intervention. Primary prevention is oriented toward education in an effort to improve the ability of individuals, families, and society in general to satisfy human needs before the emergence of problems. Secondary prevention attempts to intervene as soon as possible after the emergence of signs of stress, breakdown and so on, to forestall further deterioration of the situation. Tertiary prevention aims at rehabilitation or re-establishing conditions of capability. The above

concepts are parallel to the medical model, e.g. Lavell and Clark talking about preventive medicine define a continuum of problem severity with five gradients under primary prevention, secondary prevention and

## Primary Prevention.

tertiary prevention.

- 1) Health promotion-enhancing the health, strengths and growth tendencies in individuals.
  - Specific Protection -- preventing a particular condition 2) from arising similar to medical immunization. . This emphasizes environmental alternation.

Secondary Prevention

Early diagnosis and prompt treatment -- incipient pathology is implied, and treatment set out with the goal of de-3)

- terring or precluding serious later pathology. Tertiary Prevention
  - 4) Disability limitation -- limitation of the extent or
    - intensity of the pathology.
    - Rehabilitation -- maximization of residual capacities. 5)

Putting these five gradients of problem severity into a diagram,

one will find, at the start of the continuum, healthy individuals, families, communities and social institutions. This end of the continuum represents healthy functionality where stress or the cause of breakdown is not visible as a symptom. As one moves along the continuum one encounters the various stages towards total breakdown and dysfunc-

tionality.

#### Figure 5

The Prevention Continuum

Healthy Function- ality	Specific protection & prevention	Diagnosis &	Disa- bility Limi- tation	Reha- bilitation	Dysfunc <del>.</del> tionality
<u>م</u>	c	(breakdown	)		•

•	•	• .		-
	PRIMARY PREVENTION	-	S ECON DA RY PREVENTION	TERTIARY PREVENTION

In the light of the prevention continuum as shown in figure 5, the emphasis of the Preventive Social Service Program in Alberta is *i* aimed at the primary and secondary prevention. It is mentioned in the <u>Preventive Social Services, Administrative and Policy Outline</u> that

Primary prevention is emphasized as being most effective and economic in long term planning. It is concerned as creating personal, family and community enhancement to offer more opportunity for a wholesome and satisfying life. This may be related to such topics as education toward more meaningful personal self-worth, a better understanding of inter-personal relationship, family life education and so on. 11

Secondary prevention may be thought of in terms of early diagnosis or identifying and treating the first symptoms of personal need. Examples are found in the use of homehelpers and in lay counselling projects where a minimum of early help precludes the need for crisis intervention at a later date, 12 A distinction is also made in the policy outline between the preventive program and other areas of social pervices: 65

Preventive, designed to enhance individual and family life, and to preclude breakdown or dependency upon public funds. Rehabilitative, designed for those who are social casualties, but who have potential to gain or regain their independence. Financial, designed for those who are deemed to require continuing financial support.<sup>13</sup>

The rehabilitative and financial services are considered as the functions

of other agencies.

Summary

This chapter describes the principles, the objectives, the administration, the cost-sharing, and the development of the Preventive Social Services Program in Alberta. The concepts of prevention are also discussed. The intent of the next chapter is to provide some

background information on various Preventive Social Services projects in Edmonton.

#### Footnotes

66

<sup>1</sup><u>Contact</u>, Department of Health and Social Development, Alberta, 1973, June, p. 1.

<sup>2</sup><u>Preventive Social Services, Administrative and Policy Outline</u>, Department of Health and Social Development, Alberta, Edmonton, 1970, pp. 1-2.

<sup>3</sup>Ibid., p.º14.

<sup>4</sup>Government of Alberta, "The Preventive Social Services Act," in <u>Revised Statutes of Alberta</u>, Government of Alberta, 1970, Chapter 282, sec. 3.

<sup>5</sup>Preventive Social Services, Administration and Policy Outline, op. cit., pp. 10-11.

<sup>6</sup><u>Ibid</u>., pp. 7-9.

<sup>7</sup>Refer to <u>People Need People</u>, a pamphlet from Preventive Social Services, Department of Health and Social Development.

<sup>8</sup>"A Guide to Application", Edmonton, Social Service Department, p. 1.

9<u>Ibid</u>.

<sup>10</sup>Leavell, H.R. and Clark, E.C., <u>Preventive Medicine for the Doctor</u> in <u>His Community</u>, New York, McGraw-Hill, 1965.

<sup>11</sup> <u>Preventive Social Services, Administration and Policy Outline,</u> <u>op. cit.</u>, pp. 16-17.

<sup>12</sup><u>Ibid</u>., p. 17.

13 Ibid., pp. 3-4.

CHAPTER V

PREVENTIVE SOCIAL SERVICES PROJECTS IN EDMONTON: A DESCRIPTION

The purpose of this chapter is to describe different types of Preventive Social Services projects funded in the City of Edmonton. The descriptions of the projects will be based entirely on the information obtained through personal interviews with the key workers of respective projects and an examination of accessible documents; neither time nor funds were available for field studies. Some of the information, however, was given on the basis of the respondents! (or the key workers') personal knowledge and does not necessarily represent the consensus of all the relevant employees and board members. There will be a good deal of variation in the depth of the descriptions of the projects; and this variation corresponds to the quality and quantity of information received by the author regarding each project.

At the time of the study (i.e. in the Spring of 1975), there were twenty funded Preventive Social Services projects: sixteen nongovernmental, and four governmental operated under the auspice's of the Edmonton Social Service Department. These funded projects, according to the various pamphlets put out by the Edmonton Social Service Department can roughly be classified into the following types:

A. <u>Family Services</u>
1. Edmonton Family Planning Service (E.F.P.S.)
2. Emergency Homemakers' Service (E.H.S.) (Family Service Association of Edmonton)
3. Family Aide (F.A.) (Edmonton Social Service Department)
4. Family Life Education Council of Edmonton (F.L.E.C.E.)
5. Urban Life Skills and Child Care Program (U.L.S. & C.C.)
(Canative Housing Corporation)
B. <u>Services to Youth</u>
6. Edmonton Big Sisters (E.B.S.)
7. St. Francis Youth Club (S.F.Y.C.) (Boys Club of Edmonton)
8. Teen Time
C. <u>Services to Children</u>
9. Bissell Ghild Care Centre (B.C.C.C.)
10. Day Care (D.C.) (Edmonton Social Service Department)
(Edmonton Social Scieles Department,
D. Services to Senior Citizens
11 $(0, another Friendelpin (0, F_{a}))$
11. Operation Friendship (O.F.)
12. Strathcona Blace Society (S.P.S.)
13. Society for the Retired and Semi-Retired (S.R.S.R.)
E. <u>Neighborhood Services</u>
14. Dickensfield We Care Centre (D.W.C.C.)
15. Norwood Community Service Gentre (N.C.S.C.)
16. Open Door (0.D.).
F. Support Services
.17. Aid Service of Edmonton (A.I.D.)
an a

A 18. Community Social Services (C.S.S.)
 (Edmonton Social Service, Department)

19. Home Care (H.C.) (Edmonton Social Service Department)

20. Meals-on-Wheels (M.O.W.) (Victorian Order of Nurses)

As the description of all the twenty projects is of considerable length, only some cases will be presented in this chapter. The descriptions of the other projects will be put in Appendix B for reference. In selecting the cases for this chapter, two criteria will be used. First, the cases should include one from each of the six different types

(Family Services, Services to Youth, Services to Children, Services to

Senior Citizens, Neighborhood Services, and Support Services). Second,

the selected cases should include governmental as well as non-governmental. As a result, six projects will be presented in this chapter. They are

as follows:

Family Life Education Council of Edmonton (non-governmental)
 Edmonton Big Sisters (non-governmental)

3. Day Care (governmental)

4: Society for the Retired and Semi-Retired (non-governmental)

5. Norwood Community Service Centre (non-governmental)

6. Home Care (governmental)

For the purpose of analysis, the framework for the description of the projects will be based on the analytical one developed from a review of literature in Chapter III. The projects will thus be described under the following headings:

a. \_Historical background

b. Sponsorship

# c. Objectives

d. Clientele

e. Methods of work

f. Organization and administration\*

The Description of Six Preventive Social Services Projects in Edmonton

#### 1. Family Life Education Council of Edmonton (F.L.E.C.E.)<sup>1</sup>

Historical background. In 1961 some citizens in Edmonton and some volunteers in the field of mental health had expressed the needs for more day care facilities for the pre-school children. Then, between 1961 and 1963, Edmonton Welfare Council, now the Social Planning Council set up a committee to study day care in the City of Edmonton. The study evealed many cases of inadequate ware arrangements which were potentially. harmful to children left the nese situations. In 1963 a committee was formed with representatives from many commentity organizations to explore the best methods and forms of implementing a program of parent training." In the summer, a report was prepared by collecting a great deal of materials on family life programs and interviewing key persons of various organizations. The results of the survey had indicated that there was a real need for a Family Life Program in the city. A workshop on Family Life Education was therefore held in 1964 with approximately seventy participants representing numerous sity agencies, and organizations. The committee which acted as the steering committee was later enlarged by representatives from other agencies. The enlarged committee met in

In the organizational structure charts, those paid staff will be indicated by an \*. September of 1964 and arrived at a definition of family life education

as follows:

Family Life Education attempts to assist people to develop - insights into themselves as individuals - an understanding of the needs and feelings of others

- satisfying family and community relationships.

The committee also recommended that the program be offered under the auspices of the Edmonton Welfare Council. But the proposal was turned down on the grounds that this was a direct service and thus outside the

scope of the Council's activity. Instead, it was proposed that a separate unit be established with help from the Council. Some committee members accepted this challenge. During the summer months of 1965 the committee prepared lists of speakers, films, and other resources that would be useful for family life education programs.

With the advent of the Preventive Social Services Program in 1966, the committee incorporated the Family Life Education Council and made a successful application for Preventive Social Services funding. The project was initiated on a formal basis in January of 1967.

b. <u>Sponsorship</u>. The Council is funded primarily by Preventive Social Services.

c. <u>Objectives</u>. The primary objective of the Council, as stated in the Constitution which was drafted by the committee in 1966, is to:

. promote, create and foster a program of activity for members of families to engage in with the aim of assisting people to develop insights into themselves as members of a family; an understanding of needs and feelings of others; increased confidence in themselves as people and as parents; satisfying family and community relationships.

The intent of this statement is reinforced in a later statement developed by the Board of Directors of the Council. It says,

The basic purpose of our programs is to provide more 'awareness of one's self as a person and also an awareness of others in order to develop sufficient understanding of one's own behavior and the impact of that behavior upon other people what the more genuine interpersonal communications be brought about with positive results in individual growth and responsibility.

d. <u>Clientele</u>. The program is open to any interested citizens or agencies in the City of Edmonton.

· .....

e. Methods of work. There are two methods through which the Council attempts to achieve its objectives. The first method is to provide direct services in two patterns. In the first pattern, programs are provided under the auspices of another organization, such as the Edmonton Public School Board, a church, or a community organization request the Council to have a series run for its members. In the second pattern; the Council develops a series under its own auspices, and through general advertising opens the program to any interested

individuals.

Another method that the Council has used is to involve a large . number of volunteers and train them as group discussion leaders or program. leaders. At present, the emphasis of their strategy is on training volunteers and working with organizations or groups than with schools.

The method that they use to design a program is to structure the content with the group requesting the service. In other words, " the program leaders and the coordinator will involve the clients or

groups in the planning and operation of the program content. Typical content areas include marriage preparation, better parenting programs, marriage enrichment, human relation skills, communication skills etc. The educational techniques used for the courses are mostly small group discussion, short presentation, role playing, video-type etc. Questionnatres are usually used to assess the effectiveness of the methods.

73





Figure 6



The Board of Directors is the policy-making body composed of representatives of agencies and associations which are related to the Council through its activity and interested citizens from the City. The role of the Executive Director is to carry out policies, to plan programs, to supervise the operation etc. The Demily Life Consultant and Special Communication Project Co-ordinator are responsible for programming and some co-ordination. At present, there are approximately eighty volunteers and most of them are involved as discussion leaders or program leaders.

## 2. Edmonton Big Storters (B.S.)

a. <u>History ground</u>. The Edmonton Big Stater Soctety began as a fully warry autonomous organization at the beginning of ober 1971 on the initiative of two community workers. One of the munity workers had operated a similar organization in Waterloo,

Intar lo

During October and November of the same year, the groundwork was done for the Edmonton project. Personal contact was made with various inclues and interested individuals in the community to assess the need for such A program and to enlist support. At the end of November and early December, the first referrals were received and the first matchings were done, i.e. one Big Sister for each Little Sister. Since then the program has grown steadily. The funding for Big Sisters has been a succession of short-term grants from the Local. Initiatives Program, and from the Opportunities for Youth. There was a period in 1972, when the staff members worked on a voluntary basis. In September 1974, the Society, received a Preventive Social Services grant and a \$5000 developmental allocation from United Way. b. <u>Sponsorship</u>. At present, the Edmonton Big Sisters Society is funded by Preventive Social Services. The house which the Edmonton Big Sisters Society uses is provided by the City without charge.

of the Society are:

a) To provide a climate for girls 8-18 years of age, to enrich their social, recreational, and emotional experience.

b) To develop in girls personal strength, character, selfdetermination and self-evaluation,

L) by fostering close, interpersonal (one-to-one) relationships

2) by demonstrating how one may communicate and relate effectively within one's environment

3) by promoting general social interaction and peer relationships

4) by holding events and activities conducive to companionship.

The specific objectives of Edmonton Big Sisters are developed by the Board of Directors of the Society.

d. <u>Clientele</u>. The program is open to girls from 8-18 years of age in all areas of Edmonton. But most of the girls are referred by School counsellors and social service agencies in the Beverly and Jasper Place areas.

e. <u>Methods of work</u>. The basic method of work of the Edmonton Big Sisters is based upon the concept of one-to-one relationship, i.e. to match a Little Sister with a Big Sister who is a volunteer from the community. The first procedure is to recruit volunteers through mass media or other agencies to act-as Big Sisters. Having been interviewed, the Big Sisters are expected to come out to some of the weekly activities

1 2 123 4

and the orientation program. The orientation is designed to inform the volunteer of the various aspects of volunteering and of what is required of a Big Sister.

On the other hand, the Little Sisters are referred by parents, friends, or social and educational agencies in the city, including the Department of Health and Social Development, City Social Services, schools, etc. The girls are contacted and invited to join.

In order to facilitate the matching process, activities are organized on a weekly basis. The activities are also designed to facilitate positive growth for the individual in relationship to their personal needs. The Little Sisters are divided into two groups, according to age, which allow for more individual attention. For the girls eight to twelve years of age, the emphasis of the program is placed upon recreation, socialization, and group participation.

For girls thirteen to eighteen years of age, the activities are designed to provide educational experience for them and relate to their social and emotional needs. Activities may include group... sessions on alcohol and drug abuse, or self-identity and development, etc. In planning the activities, the Little Sisters are encouraged to express their interests on certain topics. Most of activities are planned by the staff members, Big Sisters and Little Sisters from the group.

All the activities are designed to assist both the Little and Big Sisters In friendship development and spontaneous expression of feelings. Only after a friendship has formed through several activities is a formal matching considered. Once the matching has taken place, the Little Sister is encouraged to contact her Big Sister rather than the staff, when a problem arises in her life. The artif keeps in contact with the Big Sister through evaluations of the relationship which are arried out every month. The evaluation permits an exchange of information between staff and Big Sisters, give support and encouragement to Big Sisters and ensures that a resitive relationship is developed between the Big Sister and Little Sister. If the relationship is not developing along positive or constructive lines, a re-matching takes place.

With the help from the Department of Culture, Youth, and Recreation, the Society is in the process of setting up an overall evaluation of the Edmonton Big Sisters project.

f. Organization and administration. The Board of Directors is made up of citizens-at-large, a social worker, probation officer, and a few Big Sisters. The Board is responsible for recruiting volunteers, - raising funds, policy making etc.

The coordinator is responsible to the Board for the program and the program staff; planning and organizing the entire program with the cooperation of the Field worker and the Committee of the Board; presenting any new innovations to the Board; assisting the Board in management, and in recruiting new Board members, and so on. The Field workershare responsible for the screening and evaluation of volunteers, the matching of Big Sisters and Little Sisters, the referral of Little Sisters who would gain more from



another agency, evaluation etc.

Volunteers are from various walks of life. They provide guidance and friendship for the Little Sisters, assist in developing the programs and supervise group activities.

3. Day Care (D.C.)

a. <u>Historical background</u>. The City's involvement in subsidized day care can be traced back to 1966 after the submission to City Council of two major position papers, outlining the need for subsidized day care, by the Edmonton Welfare Council and the Family Service Association. At that time, there was only one Community Day Nursery which was jointly funded by the United Way and Edmonton Social Services. In response to the growing awareness and concern for more and better day care, the City Social Service Department hired the first director of Day Care

. in 1967. The Director working in conjunction with an Advisory Committee on Day Care attached to the Social Service Department began to formulate Important elements in this policy . the basic policy for the program. included the following:

- (1) extension of service would be sought through the purchase of service from non-profit community based organizations;
- public subsidy would be available primarily to families with special needs, for example single parent families, (2) and low income families, and; •
- (3) public funds, would only be available to programs of good quality, providing for all aspects of the child's developmental néeds.

Based on these guidelines the City again in conjunction with the United Way funded the Primrose Place Day Care Centre in 1967 and 

Downtown Day Care Cepton 1968,

.

During this period, the City was considering the development . 4 · of a high quality day care program under direct Cipy auspices. Since there was a shortage of trained staff in the area, the City felt that the initiation of such a program could serve as a training centre. On the other hand, the centre could be used to test out a number of approaches to provide day care service. As a result, the City proposed the development of Glengarry Day Care Centre as a teaching and demonstration centre, and applied for funding from the Provincial Government. But the application-for funding was turned down. Following a strong reaction from the public, the application was finally approved. Glengarry

was opened in 1969.

Due to the need for extended day care services for children, the City, at present, has subsidized eleven centers. While group

t and a second second

care for pre-school children is the major component in the day care service system, the City has involved itself in two other programs. They are the Family Day Care and After School Care. The Family Day Care is a service which utilizes private Homes in the community which have been screened and assessed. Subsidization is available to families on the same basis as in the case of the centres. The After School Care Program, which is the third component of the subsidized day care services, serves dependent school age children in a group setting. At present, three types of day care services provide a total of 821 day care space for the families.

b. <u>Sponsorship</u>. The subsidization of the 821 day care space provided by the three types of services as described above is funded entirely by Preventive Social Services.

c. <u>Objectives</u>. The goals for the Day Care project have, remained the same over the years. The goals as stated in the policy statement

It is the principal goal of the City to make available duality day care services to those families and children who have the greatest need for care depending on social and financial circumstances.

are:

1)

2) It is also the goal of the City to support a quality day care service in order to create citizen awareness and understanding of the importance of good standards to ensume that such services contribute to child development. 30.

d. <u>Clientele</u>. The program serves cilies within the City limits. Generally, families will receive the following

order:

1) families with low income

2) one parent families

3) families with special needs, i.e. having problems which ojustify providing special care for the child.

e. <u>Methods of work</u>. Instead of providing funds directly to the centers, the program purchases services on behalf of families selected by the centers. The basic approach is to go to various community groups and encourage them to develop day care services in keeping with the guidelines that have been set for the program.

Basically, the program subsidizes three types of services; the Group Care of Pre-school children, Family Day Care, and After School Care. Assessment for fee payments and admission into the day care program, which are based upon the priority system, are done by social workers in the City Social Services Department or in the centers. A common sliding fee scale is used to determine the amount of fees for each family. Since the program has guaranteed a certain amount for each child approved, the

subsidy per child is payable to the centre on behalf of the family on a monthly basis. In addition the program may provide 'building cost' For some of the centers.

At present, the program also encourages the centers to provide 10 per cent of their space to full fee families, which are not on low income or single parent families. In this program, there is no extensive or well-developed use of volunteers. In terms of evaluation, the director has stressed the difficulty in assessing the quality of the services because longitudinal studies of the children are often needed. However, resources are not available for such research.

f. Organization and administration. Since Day Care is a project operated under the auspices of the City Social Services Department, its organizational structure will not be shown in this section. (For meference, refer to figure 3 in Chapter 4).

At the present time the City Social Services is involved in a network of day care services through the funding and/or administration of the three types of services:

1) Group Care of Pre-school Children

City operated Glengarry Day Care Center
Community Board Operated Beverly Day Care Center
Downtown Day Care Center
Community Day Nursery
Jasper Place Day Care Center
Primrose Place Day Care Center
South Edmonton Child Care Center
Student's Union & Community Day Care Center
West End Day Care Center
Parent Co-operative Center D'Experience Pre-Scholaire

2) Family Day Care

City operated

3) After School Care

- City operated Glengarry Day Care Center - Community Board Operated Primrose Place Day Care Center

Fulton Child Care Center Edmonton After School Care Garneau After School Care

With the exception of Glengarry Day Care Center which is operated by the City Social Services, the other subsidized Day Care

centers are managed by their own Board of Directors, and staff.

For the program as a whole, there is a director of Day Care who carries responsibility in a number of areas including:

a). Consultation and coordination relating to developing

- 1. Day Care programs b) Liaison with existing subsidized Day Care Services c) Supervision of the Glengarry Day Care Centre
- d) Supervision of the Family Day Care Unit
- e). Coordination of Day Services vis-a-vis other community
- services Public education, public relations, and provision of f) resource information.

## Society for the Retired and Semi-Retired (S.R.S.R.)

a. <u>Historical background</u>. The Society was developed by groups of people working in nursing homes and senior citizens in the community who felt that certain social, economic, and housing needs were not being adequately met: The Edmonton Social Planning Council then involved

senior citizens at large with the help of resource people in setting up the objectives of the project. It was incorporated under the Societies Act in October, 1970 and then funded by the United Way and Preventive

Social Services in the same year. The Society is a non-profit organization. **W**7

Sponsorship. At present the Society is primarily funded by Preventive Social Services.

Objectives. The objectives of the Society are clearly defined

in the Annual Report, 1974. They are as follows:

The Society will do whatever may be necessary to help the aging person achieve the highest possible quality of living through:

- 1) Improving the quality of life for older persons by providing fellowship and aiding in their physical, social, and emotional\_well=being.
  - Providing co-ordination of existing services for older persons and initiating action towards meeting unmet needs.
  - 3) Presenting opportunities for the resources and abilities of retired persons to help themselves and the community.
- 4) Enabling the integration of senior citizens with all facets
- 5) The provision of services to meet the special needs of the older adult.

The major emphasis of the objectives of the Society has remained the same over the years, only the original objectives were expanded in response to the needs of the rapidly growing population of senior citizens.

d. <u>Clientele</u>. The program is open to retired and semi-retired citizens in-the City of Edmonton. Membership fee is one dollar a year, and members are entitled to vote in the Society. But it is not necessary to be a member to participate in any of the activities of the Society.

e. <u>Methods of work</u>. The first method of work is to provide direct services such as counselling, information and referral service for older people and their families. These types of services are basically provided by staff members of the Society. One of the vital programs of the Society is the volunteer hostesses. It is through the hostesses and staff that the Society as a drop-in centre provides a friendly and supportive atmosphere. Other volunteer services include volunteer visiting, telephone reassurance for elderly senior citizens, clerical services at the centre etc.

The third method of work is to involve senior citizens in planning and organizing various activities at the centre. Social activities are generally planned by the Activities Committee which is a group of retired volunteers. Education forums which are held every week are planned and organized by the Education Committee which is also composed of senior citizers.

In the community, older members of the society have participated in the Heritage Program and talked to school classes about pioneer days. The Society has also been active in informing various city departments and community organizations about the needs of senior citizens in regard to transportation, housing and health.

f. Organization and administration. Policy decision is made through the Board whose members are elected by general membership every year. At present, board members are mostly professional people. The Co-ordinator is responsible for the overall administration, program planning with staff and committees, community education, and co-ordination. The Assistant Co-ordinator shares in the responsibility of the Co-ordinator with particular emphasis on counselling, and supervision of direct



86

service. Information Counsellor is responsible for individual information and referral with older people as recorded under direct service. Primary responsibility of the Information Assistant is assisting the Information Counsellor in direct service. Volunteer Co-ordinator is responsible for co-ordinating the visiting program and volunteer workers, recruiting,

selecting, training anf providing supportive help to volunteers and co-ordinating volunteer activities in co-operation with staff. The Herftage Co-ordinator is responsible for arranging with schools for presentations by elderly people on pioneer days. The Advisory Committees which are made up of at least one board member plus some other senior citizens are responsible for planning various social and educational activities. Other volunteers are mainly involved in the provision of Y

#### Norwood Community Service Centre (N.C.S.C.)

a. <u>Historical background</u>. Norwood Community Service Centre was started in 1966 as a headstart program for five year old children. At that time head-start programs were so popular\_in U.S.A. that the National Council of Jewish Women was stimulated to take action in Edmonton. The Norwood area was chosen for a pilot project. It was more of pressure laid on from outside in its formation. All the initial volunteers were members of the National Council of Jewish Women and the Edmonton Junior League. They concentrated at first on children, but evaluation of certain programs in the U.S.A. indicated gains made by children were lost after two or three years if she family was not involved. The Norwood program. was, therefore, expanded to include mothers and youth.

From its inception as a limited pilot project, the centre has developed to its present multi-service status. b. <u>Sponsorship</u>. The National Council of Jewish Women and Junior League gradually withdrew their control.and funding in the past two years. At present, the center is funded primarily by Preventive Social Services with some community support.

c. <u>Objectives</u>. The Norwood Community Service Center is a neighbourhood center which strives to provide 'opportunities for individuals and families in the Norwood Community to develop to their fullest potential. More specially, the aims as determined by the Norwood Community Service Center Policy Board at their 'Process Planning' in 1974 are:

- 1) To develop children's potential for learning
- 2) To provide a meeting place and to develop programs for various interests and activities in the Norwood Community
- 3) To provide an initiating and co-ordinating resource centre for the Norwood Community
- 4) To be aware of what other groups in the community are doing and support those projects that the Board fees are improving the quality of life in the Norwood Community.

Since the inception of the Center, the scope of the objectives has been broadened. The Center has developed from a pre-school program to a multi-service center trying to respond to various kinds of social, educational and recreational needs in the community.

d. <u>Clientele</u>. The Center basically serves residents in the Norwood area, but any interested citizens from outside are welcomed. Except for the pre-school program, they have set up a service boundary from 90 Street to 103 Street and 111 Avenue to 118 Avenue because there is an increasing demand for the programs in the area and priority has I to be given to local residents. e. <u>Methods of work</u>. The school programs which are designed to be-school children in the areas are conducted by well-experienced "teacher's aides, and volunteers. In these programs some parent volunteers are recruited to assist in classroom activities. Similarly, nursery volunteers are recruited to assist in the nursery services provided by the Centre. For the group activities, many other volunteers are involved as group leaders who are responsible for planning and conducting various programs, including craft groups, social activities for senior citizens or for housewives in the area, speakers from various local departments, field 'trips, fund raising, and so on. In addition, the Center has organized a summer youth program and provided leadership training for users of the center, staff, and board members.

Evaluation of the services is mainly through informal follow-up of clients by telephoning, visiting, etc. Upon the request of the Preventive Social Services advisor, the Policy Board undertook a 'Process Planning' in June, 1974. Goals and plans of action were established.

f. Organization and administration. The Policy Board members are responsible for policy decision-making, and are elected by the community in the annual general meeting. Nomination forms are sent out to the community with the newsletter. At present, over 50 per cent of the board members are people in the community mostly parents of pre-school program. The other part is composed of resource persons from Parks and Recreation Department, principal, librarian, child



specialist, and psychologist in Edmonton School Board. The role of the Executive Director is to administer project, co-ordinate activities between committees, groups and individuals, supervise staff, try to be aware of changing needs in the community and initiate action to meet those needs, try to acquire funding, monitor budget, support other groups or individuals that try to upgrade the community in any way.

## 6. <u>Home Care (H.C.)</u>

a: <u>Historical background</u>: In 1971 the Social Service Advisory Committee of the Edmonton Social Service Department initiated a series of community meetings to look into the area of home-based services. As a result, a community task force representing approximately forty agencies, organizations and consumer groups was formed to carry out a survey. Questionnaires were administered to various agencies, and consumers of the service. The results of the study indicated that the service delivery system had gaps in some areas; overlapped in others and there was a total lack of co-ordination throughout. The task force, therefore proposed to the Edmonton Social Service Department the development of a community based co-ordinated Home Care Program.

In December 1972 a proposal for a Home Care Program was presented to the Provincial Department of Health and Social Development by the City Social Services Department and Local Board of Health. In April 1973 the financial resources for operation and development of the Home Care Program was approved by Preventive Social Services.

b. <u>Sponsorship</u>. The program was sponsored by Preventive Social Services and the Local Board of Health.

c. <u>Objectives</u>. The basic objective of the program as stated in the Annual Report is:

To provide co-ordinated services within the home which would enable individuals and/or their families to better cope with a particular situation, whether emergency or long-term in nature.

#### Other objectives are:

Ð

- To establish a centralized administration for the co-ordination of home based services so people may receive those services more easily and in a co-ordinated manner.
- 2) To provide a level of care at home not readily available at present so that existing community facilities may be more appropriately used and family resources reinforced.
- 3) To provide care at home to people who would otherwise have to be instigutionalized, thereby gaining more effective utilization of institutional facilities.
- 4) To continually evaluate the benefit of the program to those served and to the community so further services may be stimulated to fill unmet needs.
- 5) To ensure that the ability to pay shall not restrict the right to receive services within the program.
- 6) To facilitate the referral of applicants whose needs are not within the terms of reference of the Home Care Program to appropriate agencies in the community.

d. <u>Clientele</u>. The program is open to any person living within Edmonton City limits, who is under medical supervision and a health service is required, who is in need of one or two basic services, i.e. nursing, physiotherapy, homemaking or home help, whose condition is such that he can be treated adequately at home, and whose home and family environment are suitable.

e. <u>Methods of work</u>. A basic procedure is to arrange for and co-ordinate a variety of home related services, such as nursing, home care, physiotherapy, Meals-on-Wheels, etc. for people in their own homes so that institutionalization or hospitalization may be reduced or prevented. These supportive services are not provided directly by the Home Care Program, but purchased by the Program through contracts with various agencies.

A client or referral may come from any source, community or hospital. The individual is then evaluated for suitability for home care, using the admission criteria as a basis. If the criteria are not met, the applicant will be referred to appropriate agencies in the community.

In cases where provision of Edmonton Home Care Program services can replace or shorten hospitalization, the program provides a maximum of one month full service free. In other cases, payment for services are assessed on an individual basis. The staff does its own evaluation.

f. Organization and administration: The Edmonton Home Care Program is jointly administered by the City of Edmonton Local Board of Health and the Edmonton Social Services. The program is administered by a Home Gare Management Committee composed of the Deputy Medical. Officer of Health from the Local Board of Health and the Director of Planning from the Edmonton Social Services. An Advisory Committee of representatives: appointed by both departments is set up to assist in the planning and on-going development of the program. There is also a Technical Subcommittee made up of representatives from various service agencies to assist the Advisory Committee.

The Co-ordinator or Administrator is responsible for the overall planning and administration of the Home Care Program. In October of 1974 the Home Care Administrator's duties were expanded to include responsibilities for financial planning and control of P.S.S. funds for the Emergency Homemaker Service and Meals-on-Wheels projects.


(Adapted from the organizational chart of the Home Care Program)

The Nurse Co-ordinator is responsible for obtaining and assessing referrals from the community and discharges from hospital and for 

co-ordinating services, etc.

• ....

The Public Health Nurse acts as a hospital-home care liaison . ۰. nurse and is responsible for identification and referral of patients to home care. The Home Care Social Worker is responsible for obtaining and assessing referrals from the community, to ordinating services, short- o

term counselling, and liazing with other agancies and hospital social service departments.

Summarý

In this chapter six Preventive SocialaServices projects in Edmonton, i.e. the Family Life EducationoCouncil of Edmonton, Edmonton Big Sisters, Day Care, the Society for the detired and Semi-Retired, the Norwood Community Service Centre, and Home Care are described. The descriptions of the six projects are in the following categories: historical background, sponsorship, objectives, clientels, methods of work, and organization and administration. The selected cases which represent six<sup>0</sup>major types of services include four son-governmental and two governmental projects. From this description of Preventive Social Services projects in Edmonton, attention will be shifted in the next chapter to the focal point of the study: the analysis of the Preventive Social Services

projects in the light of community development concepts.

Footnotes.

<sup>1</sup>From an interview with A. Greenwood, Executive Director of Family Life Education Council; "The Philosophical Base." The Family Life Education Council of Edmonton; and "A Local Family Life Education Council: Emphasis in two areas of programming," a round table discussion paper by Ed. Branch.

<sup>2</sup>From an interview with J. McKinnon, the Co-ordinator, and two field workers of Edmonton Big Sisters Society; Edmonton Big Sisters Society, <u>Annual Report</u>, 1974; and <u>Constitution of the Edmonton Big Sisters</u> <u>Society</u>

<sup>3</sup>From an interview with M. Day, Director of Day Care; Edmonton Social Services, Day Care Branch, <u>1974 Year End Project Report</u>; <u>Day Care Policy Statement</u>; and the <u>Position Statement on Day Care</u> to Commissioner of Public Affairs.

<sup>4</sup>From an interview with B. Bryden, director of Society for the Retired and Semi-Retired, and Society for the Retired and Semi-Retired, <u>Annual</u> Report, 1974.

<sup>5</sup>From an interview with L. Golightly, secretary of Norwood Community Service Centre, and Norwood Community Service Centre, <u>Annual Report</u>, 1974-75.

<sup>6</sup>From an interview with M. Sorochan, Director of Edmonton Home Care Program: Edmonton Home Care Program 'General Information'; <u>Annual</u> Report for Edmonton Home Care, Program for Calendar Year 1974'.

#### CHAPTER VI

# PREVENTIVE SOCIAL SERVICES VIEWED THROUGH COMMUNITY DEVELOPMENT CONCEPTS

The preceding chapters have outlined the purposes of this study, reviewed the literature pertinent to the analysis of the projects, described the major concepts and organization of the Preventive Social Services Program and six Preventive Social Services projects in Edmonton. The purpose of the present chapter is to study Preventive Social Services in the light of community development concepts.

This chapter will be divided into two major sections. In the first section the analytical framework developed from a review of literature in Chapter III will be described. On the basis of the analytical framework, the various aspects of the Preventive Social Services projects will be studied in the second section.

The Analytical Framework

The analytical framework which emerges from the literature of community development in Chapter III. is illustrated in figure 11. In this section, each element of analysis will be described respectively within the various dimensions of the analytical framework.

97

#### Figure 11

The Analytical Framework



### Initiation of the Projects

The first dimension of the analytical framework concerns how the projects were initiated from the stage of identification of needs to the establishment of the projects on a formal basis. The analysis will be based on the available information regarding the historical background of respective projects. The analysis will be in terms of the extent of participation and initiative on the part of the people and local groups in the community and the development agencies, i.e. the Provincial and Municipal Preventive Social Service Department. The level of participation on both parts of the community and the development agency can be arranged in two, intersecting axes as illustrated in figure 12.

# Figure 12

High

# Level of Participation in the Initiation of the Projects.

(C)

Participation.on the part of the community (people & groups) High

(B) (A)

Participation on the part of the Social Service Department (Provincial & Municipal)

Low

(D)

Low

Based on the principles of community development as illustrated in Chapter III, quadrant (A) is the most desirable situation suggesting that there are high interests on both parts of the community and the development agencies, and thus, the local development efforts are united with the government's. Quadrant (B) is also a desirable situation because most people who are involved are community members or representatives of local groups. Both Quadrant (C) and (D) indicate some form of imposition from outside with the implication that this may engender dependence. With reference to quadrant (D) the projects were initiated by persons or organizations external to the community.

# Operationalization of the Projects

The second dimension of analysis of the framework concerns the operationalization of the projects which includes the objectives, the clientele, the methods of work, and the organizational structure.

# 1. Objectives

In this section, analyses of the objectives of the Preventive Social Services projects include: a) the needs inherent in the objectives, b) preventive orientation versus rehabilitative orientation of the objectives, and c) development of the objectives.

a. <u>An analysis of needs</u>. Every development objective, whether pursued by a social or economic development organization, should aim at the betterment of people and the satisfaction of human needs. In order to understand the needs inherent in the objectives of the Preventive

101

Social Services projects in this study, it would be useful to define

the terms and to distinguish the various needs from one another. The following definitions are found in A. Montagu's The Direction of Human

Development:

QITAL BASIC NEEDS: Any biological urge or need of the organism which must be satisfied if the individual or the group is to survive.

EMOTIONAL (NON VITAL) BASIC NEEDS: Any biological urge or need which is not necessary for the physical survival of the organism, but which must be satisfied if the organism is to develop and maintain adequate mental health.

DERIVED OR SOCIALLY, EMERGENT NEEDS, Any need which arises out of the process of satisfying basic needs, which is not necessary for the physical survival of the organism, and which is not biologically, though it may under certain conditions become socially, necessary for the maintenance of mental health.

AQUIRED NEED: Any need which does not arise directly out of the process of satisfying basic needs, which is not necessary for the physical survival of the organism, but which grows out of the person's relation to the derived or socially emergent needs, and is not usually necessary for the maintenance of mental health.<sup>1</sup>

Satisfaction of needs make for health and co-operativeness, frustration of needs for disorder and hostility, whether the needs be hasic, derived or acquired.<sup>2</sup> Maslow has distinguished between 'lower' and 'higher' needs, and has pointed out the basic needs arrange themselves in a fairly definite hierarchy (survival, security, belonging, esteem and self-actualization).<sup>3</sup>

Based on the above concepts, the nature of the needs inherent in the objectives will be examined. Such an analysis would provide an understanding of the emphasis of the projects--whether it be social, economic, cultural, political or comprehensive.

Preventative orientation versus rehabiliative prientation of the objectives. The extent to which the objectives of the projects are preventative or rehabilative will be illustrated on the prevention continuum which is based on the five gradients of problem severity dis-

# Figure 13

cussed in Chapter /IV.

Preventativ

Preventative Orientation Versus Rehabilative Orientation of the Objectives

Preventat	PRIMA ŘY PREVENTIO	N	SECONDARY PREVENTION		T IARY VENT ION
tal or I		B) pecific	(C) Early		
lopmen	Education P &	rotection &	Diagnosis & Treatment	(D) Disability Limitation	
Deve			(breakdown)	0	

On the prevention continuum as shown in figure 13, the left extreme position is that of primary prevention. It is the area viewed as preferable in most concepts of community development. As mentioned earlier in Chapter III, Francis J. Bregha has suggested that a new type of planning, much directed towards prevention is required for community development. Primary prevention is also emphasized as being most

**Curative** 

5

Rehabilative

effective and economic in long term planning. On the rehabilative side there are various stages towards total breakdown where usually the individual is institutionalized. At this point, the organizations are almost entirely concerned with individuals with special needs and try to re-establish the conditions of capability.

Development of the objectives. The analysis of the developc. ment of the objectives will be in terms of the extent to which objectives are developed by the people of the community. The analysis will also be illustrated on a continuum as shown in figure 14.

### Figure 14

Development of the Objectives (people determined versus predetermined by the government)

			(C) ,	TC .
•		(B)	determined	
and the second	(A)	deter-	by the	
· · · · · · · · · · · · · · · · · · ·	determined	mined by	government	
	by commu-	individu-	agency i.e.	•
	nity repre-	als exter-	the Social	
People	sentat ives		Service	Government
determined	or mémbers	community	Department .	determined

At the left end of the continuum, the objectives are developed by people of the community based on their interests and needs. The concepts of self-help and local involvement which are stressed in the principles of community development imply that the objectives and needs of local communities should not be externally determined or centrally planned by the development agency. At the other end of the continuum,

the objectives of the projects, however, are packaged by the government agency, i.e. the Social Service Department.

104

#### 2. Clientele

Who is served by the project is a crucial factor in community development. A project may deal with all the people or work only with people who meet certain criteria. The alternatives are shown in figure 15.

> Figure 15 & Types of Clientele

All the	Specific groups	
people in the area (geographic community)	in the area (functional community)	Selected Individuals

In early community development practices, the unit of action was basically the local community. It dealt with all the people in the area. The community it served, in Ross's concept, refers to the geographic community which can be all the people in a village, a neighbourhood, a city, etc. A community development project may also deal with a group of people who share specific interests or characteristics. This specific group of people, in Ross's concept, is the functional community which is located in the middle of the continuum because it is a community segment. Finally, a projectomay deal only with selected individuals who meet certain criteria, and these individuals cannot be identified as a community.

#### 3. Methods of Work

It is suggested in Chapter III that the mode of intervention depends in large part on a number of factors related particularly to the situation and the point of development of a community. But there are certain basic elements which are included in any method of community development.

In analyzing the methods employed by the various Preventive Social Service projects in Edmonton, the following will be considered: a) process-orientation versus task-orientation of the strategy; b) inclusiveness of scope of the project issues or activities; and c) mobilization of human resources that the project may utilize a large number of volunteers or no volunteers at all in the delivery of services:

a. <u>Process-orientation versus task-orientation of the strategy</u>. The discussion on community development in Chapter III indicates that the relative importance of the process-oriented and task-oriented approach has been a debatable question.

Proponents of community development such as Biddle and Biddle, Ross, in their writings, advocate the process-oriented approach in which people of the community are encouraged to define their own goals, actions and methods in the planning process. Contrasted with it, the task-oriented approach places primary emphasis on the development and provision of services and resources that people are seen to need. The developmental aspects of citizen participation in the program planning process are not seen as being of primary importance. But the author agrees with Dunham who stated that "if one conceives of a continuum with task goals at one end and process goals at the other, the most reasonable and practical position is somewhere in the middle."

Again, the analysis of the process orientation versus task orientation of the planning strategy can be illustrated on a continuum indicating that any combination of process and task orientation is

possible.

## Figure 16

Process Orientation Versus Task Orientation

_			· · ·	•		,	Task
Process	1.1		•		 	·	Orientation
Orientation			<u> </u>		 		(service
(citizen		•	e	<b>.</b>		•	
development)						•	development)
deveropment.	• .	•			• .	1. A. S.	1. S. 19

At the left end of the continuum shown in figure 16, which represents the process orientation, the major concern is to help people in strengthening qualities of participation, self-direction and cooperation. At the other end, the provision of services is of primary importance in the task-oriented approach.

b. <u>Inclusiveness of scope</u>. An analysis of the inclusiveness of scope of theproject, presumably, will reflect the nature of the project and the focus of the planning strategy. A project may be a generalized one set up to deal with a wide range of issues, or a specialized one dealing with narrow range of problems.

The analysis will be presented in a table listing all of the areas of concern of respective projects and followed by discussions.

<sup>4</sup>c. <u>Mobilization of human resources</u>. The last element of analysis in this section is the mobilization of human resources in the delivery of services that the project may involve a large number of volunteers, or, at the other extreme, with no volunteer involvement at all.

The development and mobilization of latent human resources in communities is central to the concept of community development. It is generally, though unnecessarily, assumed that the degree and direction of the development program are determined by the experiences of the community. A development project, therefore, should aim at the recruitment and training of local citizens as volunteers.

The mobilization of human resources in the projects will be analyzed in terms of the approximate numbers of volunteers involved in the delivery of services. Volunteers involved as board members will not be included in the analysis. The analysis can be illustrated in a continuum as shown in figure 17.

#### •Figure 17

Mobilization of Human Resources

many

50+

none

(number of volunteers involved in the delivery of services)

'As the level' of volunteer involvement in the projects fluctuates from time to time, the numbers of volunteers reported are only estimations.

107

In analyzing the involvement of volunteers, the scope of the projects will also be considered. The reason is that the number of volunteers involved to some degree is covariant with the scope of the project. At least, having more than one issue or many issues will attract and make it possible for a wide segment of the community to join.

108.

# 4. Organizational Structure

In this section, citizen participation in the policy decisionmaking of the projects will be analyzed in light of Arnstein's eightrung typology described in Chapter III. Moreover, the composition of the Boards of Directors of respective projects will be examined.

# Preventive Social Services in Edmonton: An Analysia

Based on the analytical framework as described earlier, data obtained through personal interviews with the key personnel of the twenty Preventive Social Services projects, and through a review of accessible documents will be analyzed in this section. The summary of the findings and the conclusions drawn from the analysis; however, will be presented in the next chapter.

# Initiation of the Projects

In quadrant (A), there were high levels of participation on both parts of the community and the government in the initiation of five projects.

In quadrant (A), the Municipal Social Service Department was

Figure 18 Level of Participation in the Initiation

of the P.S.S. Projects

n 109

0

n 0

0

0

0

С

Participation on the part of the community (people & groups)

		High
·	( <b>A</b> )	(B)
• •	D.W.C.C.	F.L.E.C.E.
	0.F.	S.R.S.R.
	A.I.D.	B.Ç.C.C.
	ч H.C. 🤄 🦻	O.D.
•	D.C.	S.P.S'.
Partici-	ú	M.O.W.
pation on the part	4	E.F.P.S.
part of the		.U.L.S. & C.C.P.
Social Service		<b>F.</b> H.S.
Department	` • <b>•</b>	E.B.S.
(provincial	1. Sec. 1. Sec	
& municipal) High	· · · · · · · · · · · · · · · · · · ·	Low
	· · ·	
	(C)	(D)
	F.A.	N.C.S.C.

Low

С

T.Ť.

S.F.Y.C.

Ũ

N.B. (See Chapter V for full names of the projects)  $^\circ$ 

°C.S.S.

instrumental in the formation of Home Care Program, Day Care Program and A.I.D. Community workers from the Department were involved in the establishment of Dickensfield We Care Centre and Operation Friendship. But participation on the part of the community was also evident in these projects as Dickensfield We Care Centre was initiated by a group of residents, in response to the needs identified in a survey in the area. Operation Friendship began with the work of some volunteers in the area and the support from the Society for Retired and Semi-Retired which is an organization for senior citizens. The Social Service Department's involvement in subsidized day care was also in response to the growing concern for better day care as two position papers outlining the need for such services were submitted to City Council in 1966. In the case of Home Care Program, there was a community task force representing approximately forty organizations and consumer groups formed to carry out survey looking into the necessity of such home-based services. A.I.D. was established in response to a citizen committee which included concerned citizens and government officials from the provincial and municipal

levels.

Relatively high levels of community efforts were found in the initiation of the Family Life Education Council of Edmonton. The Society for Retired and Semi-Retired, the Bissell Child Care Centre, Open Door, the Strathcona Place Society, Meals-on-Wheels, the Edmonton Family Planning Services, the Urban Life Skills and Child Care Program, the Emergency Homemaker's Service and Edmonton Big Sisters (see Quadrant (B)).

Projects like Family Life Education, Open Door, Strathcona Place Society and the Emergency Homemaker's Service were actually proposed after a survey or study was done to assess the needs. Most projects identified in quadrant (B) were initiated by community members and/or by representatives of various community groups. In the case of Edmonton Big Sisters, the service began on the initiatives of two community workers, one of them from Eastern Canada. However, support was received from community groups and citizens-at-large in the development of the project.

Although Preventive Social Service Department was not involved in the initial stages of the development of the Society for the Retired and Semi-Retired, Open Door, and the Strathcona Place Society, it came. in immediately as a sponsor after 'the rise of interest' and some form of organization occurred. Preventive Social Services which did not become effective until late 1966 could only come in as a sponsor of the Family Life Education Council of Edmonton after it had been started. In quadrant (C), the Family Aid, and Community Social Services

were the ideas of the Department. There was no sign of community participation in the initiation of these two projects. Finally in quadrant (D), Norwood Community Service Centre was started by the National Council of Jewish Women, not by local residents of Norwood District. It was, therefore, more a case of pressure being laid on from outside. With regard to the St. Francis Youth Club, it was started as a branch organization of the Boy's Club of Edmonton. Finally, Teen Time was formed mainly through the effort of the executive director. On the part of the Social Service Department, there was also a relatively low level of participation in the initiation of these projects.

The analysis in this section reveals that there were high levels

of community efforts in the initiation of fifteen of the twenty Preventive Social Services projects which can be identified as community develop-

ment process.

Operationalization of the Projects

1. Objectives

a. <u>An analysis of needs</u>. Objectives are an evolving set of ideals which are determined by the interaction of the people or groups internal and external to the projects. In the area where standard of living is low and basic needs are vital, the objective of community development always goes along with economic development for the fulfilment of basic human needs. However, this does not mean that the fulfil-

ment of 'higher-needs' is down-played.

In the developed Western World, like Canada, where the standard of living is higher, the majority of the people are free from the physiological tension of hunger, lack of clothes, shelter etc., the major energies of community development programs tend to be directed to providing those services which best foster the satisfaction of people's higher-needs.

With regard to Preventive Social Service projects operating in Edmonton, concerns were expressed about child care, family life education, information provision, recreation, family planning, urban life skills, family help or home-help, housing, meals-services, transportation and so on, which are bascially social, psychological, educational and recreational in nature. Only a few of the projects, in this study are dealing with biological and physical-ecological needs. Most of the needs as identified in these preventive social services, according to Montagu's definitions of needs, are not 'vital basic needs' but 'emotional' and 'social' needs for personal effectiveness, for maintaining mental health and for enhancement and enrichment of individuals.

With the exception of the food services provided by Meals-on-Wheels and the concerns for better housing in other projects like Operation Friendship, Strathcona Place Society, and the Society for the Retired and Semi-Retired, which can be considered as 'vital basic needs', most of the preventive social services are directed towards the socio-psychological aspect of the community and to facilitate purposive social change and social development. Part of the reason is that the social problems which affect the urban centres in North America today mainly result from a complicated mix of patterns of industrial society-urbanization, sub-urbanization, the increasing specialization of labor, the development of bureautracies and a consequent increase in impersonalization and a shifting of values.<sup>5</sup>

As mentioned earlier in Chapter III, community development should be based upon the economic and social needs of the community. It is obvious that the major emphasis of most of the projects in this study is on social needs; the need for improvement of the material or economic conditions of life is not of primary importance.

b. <u>Preventative orientation versus rehabilative orientation</u> of the objectives. An examination of the basic objectives of the Preventive Social Services projects shows that most of the stated intent

# of the projects fall very clearly within the conception of primary

prevention.

eventative

Developmental or Pr

# Figure 19

114

Curative

0L

Rehabilative

### Preventative Orientation Versus Rehabilative Orientation of the Objectives of the P.S.S. Projects

	PRIMAR PREVENT	ION	SET NDARY	° TERT PREV	IARY, ENTION	
	(A); Education & Promotion	°(B) Specific Protection & Prevention	(C) Early Dia- gnosis & Treatment	(D) Disability Limitation		
	↑ (A) FLECE	↑   (B)&   EBS	(c)	ð	.0	
	NCSC TT EFPS	CSS	elling	9		
	ULS&CC CSS (community	(B) DWCC SRSR	(C) MOW, AID	•		
Lu 	organiza- tion)	BC CC OD		ss line)		
ς - ΄		SPS OF AID	FA		0	
		(informatio SFYC DC	n 	•		O

The orientation of the Family Life Education Council of Edmonton, Norwood Community Service Centre, Teen Time, the Edmonton Family Planning Service, and the Urban Life Skills and Child Care program are basically in the area of education and promotion. These projects

016

are located at the preventative end of the continuum because their objectives are to increase human functionality, to improve the quality of life or to improve growth promoting conditions in the supporting environment.

115

On the other hand, the Dickersfield We Care Centre, the Society for the Retired and Semi-Retired, the Bissell Child Care Centre; Open Door, the Strathcona Place Society, Operation Friendship, A.T.D. (information), St. Francis Youth Club, and Day Care are also aimed at primary prevention. Their basic objectives are (like medical immunization) to prevent a particular condition from arising. The emphasis of their activities are in the area of knowledge dissemination, environmental alteration, protection against potential life-cycle

hazards etc.

Edmonton Big Sisters and Community Counselling Services are located between the Primary and Secondary Prevention. The basic objectives of Edmonton Big Sisters are to develop personal strength and to foster a growth-promoting environment, which are primary prevention objectives. However, most of the girls are referred by school counsellors, social service agencies and so on. In other words, the focus of their helping resources can be regarded as secondary prevention because it is found in a problem situation.

In the case of the Community Counselling Service offered by the City Social Service Department, the project is also aimed at both primary and secondary prevention. The services include one-to-one style of intervention and, on the other hand, a wide range of group programs which deal with the causes of personal and community problems. The analysis in this. section indicates that majority of the objectives of the Preventive Social Services projects are primary prevention objectives, with only a few exceptions which are considered as secondary prevention. None of the projects in this study fall within the area of tertiary prevention. In other words, most of the projects in this study are directed towards prevention which is emphasized in the literature of community development.

c. <u>Development of the objectives</u>. In figure 20, fourteen projects are located on the left side of the continuum. These projects are: the Family Life Education Council of Edmonton, the Dickensfield We Care

					•	 ~~	
F	1	g	u	r	е	20	

The Development of the Objectives of the P.S.S. Projects

	(A) determined by community re- presentatives for members	(B) determined by outside orga- nizations or individuals	(C) determined by the govern- ment agency i.e. Social Service Department	Government
People determined	<u>۸</u>	Τ.Τ.	† F.A.	determined
	F.L.E.C.E.	S.F.Y.C.	C.S.S.	
	D.W.C.C. E.H.S.		D.C.	$\frac{1}{2}$ $\frac{1}$
	0.D.		H.C.	α ματαγραφικά τη αγοραγική τη αγ
4	N.C.S.C.		8	
	S.R.S.R.	•		0
	B.C.C.C.			
	0.F.			
	M.O.W.	0		
	A.I.D.			
	JE.F.P.S.			۵
	U.L.S. & C.	3•P•	9 0	<b>e</b>
	E.B.S.			
	S.P.S.	<b>.</b>		

Centre, the Strathcona Place Society, the Emergency Homemaker's Service, Open Door, the Norwood Community Service Centre, the Society for the Retired and Semi-Retired, the Bissell Child Care Centre, Operation Friendship, Meals-on-Wheels, A.I.D., the Edmonton Family Planning Service, the Urban Life Skills and Child Care Program, and the Edmonton Big Sisters Society. The objectives of these projects were developed by community representatives or members.

The objectives of the Family Life Education Council was developed by the Family Life Committee which was made up of representatives of local groups and organizations. On the other hand, the objectives of the Dickensfield We Care Centre were set up by a group of local residents in response to the needs identified in a survey of the area. For the Strathcona Place Society, the objectives were formulated by a group of concerned people in the community after a survey was done in Garneau and Strathcona neighbourhoods. In the case of the Emergency Homemaker's Service, the basic objective was proposed by a community organization and further developed by the Board of Directors of the Family Service Association.

For the Norwood Community Service Centre and Open Door, the present objectives were determined by their board members at their 'Process Planning' meeting in 1974. As the major emphasis of the objectives of the Society for the Retired and Semi-Retired has remained the same over the years, the object the project up by senior citizens-at-large with the object of some resource people from Social Planning Council. For the Bissell Child Care project the aim was developed by the Bissell Child Care Committee which is made up of Bissell Board members and a few citizens. On the other hand, volunteers at the Bissell Centre, community workers at the City Department, Boyle Street Community Service Co-op, the Society for the Retired and Semi-Retired and other local groups were involved in the development of the objectives of Operation Friendship.

With regard to Meals-on-Wheels, the objectives were basically determined by representatives from various community organizations and church groups. The basic objectives of A.I.D. were determined by a committee which was made of concerned citizens, government officials and representatives from other community organizations. For Edmonton Family Planning Service, the basic aim was developed by seven women in the community and a doctor who met together and drew up the constitution. Finally, the objectives of the Urban Life Skills and Child Care Program was developed by the Board of the Canative Housing Corporation which is made up of native people.

Further to the right of the continuum, at point (B), the development of the objectives of Teen Times were basically through the efforts of the Executive Director. The objectives of the St. Francis Youth Club are more or less the same as those of the Boys' Club of Edmonton, so the development of the objectives were influenced by its parent organization, not by the clientele.

At the right end of the continuum, the objectives of the Family Aide Service and Community Social Services were basically the ideas of the Social Service Department, and thus, are considered as determined by the Department. Finally, although the objectives of Home Care and Day Care were based on people's needs and interests, the stated objectives of the projects were formulated by the Department and thus are considered as government determined.

The analysis in this section shows that the objectives of fourteen projects in this study were basically determined by community members or representatives. There was high influence by the people in the development of the Preventive Social Services objectives which is stressed in the principles of community development. It is imperative that the objectives of the projects be determined by the people of local communities.

~	- 1		A	le
2.				

in the geople	Specific groups in the area	People selected by conditions
FLECE NCSC DWCC	S R&S R TT S PS OF	° MOW EHS HC FA
BCCC OD AID EFPS	ULS&CC SFYC EBS	DC

Figure 21

Of the projects studied, seven are serving the geographic communities. For Family Life Education Council of Edmonton, A.I.D. and Edmonton Family Planning Service, the total population of the Edmonton community are eligible for the services. Bissell Child Care Centre also provides services to the Edmonton district but most people served are referred by other agencies, like Boyle Street Co-op, City Social Service Department etc. The Norwood Community Service Centre, the Dickensfield We Care Centre, and Open Door are dealing with all the residents in their own territorial neighbourhoods. On the other hand, the clientele of the Community Social Services project are also geographic communities (the seven units of Social Service Department).

For those projects dealing with specific groups in the area, the society for the Retired and Semi-Retired, the Strathcona Place Society and Operation Friendship are working for the interests of the senior citizens. Both Teen Time and the St. Francis Youth Club are dealing with youth groups in Edmonton. The program of Edmonton Big Sisters is open to girls from 8-18 years of age in the area of Edmonton but most of the girls are referred by school counsellors or Social Service agencies. Finally, the target group of the Urban Life Skills and Child Care program is the native 'tenants' in the city.

To the right of the continuum, there are five projects that work only with people having specific requirements. These projects include Meals-on-Wheels, Emergency Homemaker's Service, Home Care, and Family Aide and Day Care. Although services provided by these projects are open to any person or family in the Edmonton area, priority will be given to those who meet the conditions or criteria set by the individual project. Most of the time, if the criteria are not able to be met, the applicants are referred to other agencies. In sum, the analysis shows that eight of the projects are serving geographic communities and seven are working with functional

120

communities. The other five projects are basically serving selected individuals or families instead of local communities or groups, which cannot be regarded as community development activities.

# 3. Methods of Work

a. <u>Process-orientation versus task-orientation of the strategy.</u> It is evident in Figure 22 that the Community Social Services (community organization) offered respectively by the seven local units of the

Social Service Department is the only method which is considered as purely process-oriented. Since the role of the community workers who have been assigned to each area is to encourage the people in the area or community to look into their own situations, to define their own goals, actions and methods and to use their own resources to meet their own goals, the method of work is basically a citizen development gr process-oriented approach.

#### Figure 22

Process-orientation Versus Task-orientation of the Strategy of the P.S.S. Projects

Process orientat:	ion Ł				 ļ	Task orientation
(human on citizen		Î				(service development)
developme	c.S.S.	FLECE	NCSC	• •	 MOW	
	(community	SRSR	DWCC		HC	
	organization)	SPS	TT		 EHS	
		EBS	OF SEYC		BCCC AID	
	<i>.</i>	<b>4</b>	DC		EFPS FA	
			OD		 ្រំខេត	CCP

The program planning method of Family Life Education Council of Edmonton, the Society for the Retired and Semi-Retired, and the Strathcona Place Society are also considered as process-oriented. Although the provision of services is emphasized in these projects, there are some forms of built-in mechanism to include citizens in the planning process or the development of servives. In their method of programming, the F.L.E.C.E. tries to work with the group requesting the service to structure the content. In other words, the Program Leaders and the Coordinator of F.L.E.C.E. have involved the clients ... or groups in the planning and operation of the program content. For Strathcona Place Society and the Society for the Retired and Semi-Retired, most of the programs or activities are recommended or planned by the various advisory subcommittees or ad hoc committees which are. made up of interested senior citizens. Both societies have provided seniors with opportunities of planning for, and participation in the programs. The methods of work of the Edmonton Big Sisters is also considered as process-oriented because it aims at facilitating personal growth and participation.

For the seven projects located at the center of the continuum, the methods of work are considered as both process-oriented and taskoriented. As the provision of pre-school programs, youth programs, mother's programs and so on are emphasized in both Norwood Community Service Centre and Dickensfield We Care Centre, the local residents are also encouraged to plan or organize their own activities. Also, both centers have organized workshops on 'communication skills' and 'leadership training' for the users of the centre, Board Members, volunteers and so on to strengthen the qualities of participation. "In Teen Time, the teenager groups have elected an executive and area representatives, all of whom are encouraged to work directly with the Program Co-ordinator in selecting and planning activities. In addition to the various social " and recreational activities, leadership and counsellor training programs are also organized for active teens members." The St. Francis Youth Club, on the other hand, have emphasized the small group approach, i.e. members who have interests in common will form a group to do their own things.

Instead of providing funds directly to the centres, the Day Care Program purchases services on behalf of selective families from the centres. The second basic approach of the Day Care Program is to go to various community groups and encourage them to develop day care service within the context of their set guidelines. For Operation Friendship, the basic methods of work are to organize a variety of services for older persons in the McCauley area, and, on the other hand, to encourage the older persons to participate in the organization of various programs and to use their resources and abilities to help themselves and the community.

With regard to Open Door, the major methods of work are to provide information and referral services, and to facilitate the communication process within the area by disseminating community news regularly. The latter approach is considered as work process-oriented in the sense that it strives for better understanding and cohesion of the community.

123

Further to the right side of the contiuum, the methods used by Meals-on-Wheels, Home Care, the Emergency Homemaker's Service, the Bissell Child Care Centre, A.L.D., the Edmonton Family Planning Service, Family Aide, and the Urban Life Skills and Okild Care Program are oriented to meeting certain personal and community needs. Their methods of work place heavy emphasis on the development and provision of various kinds of services. The developmental aspects of citizen participation in the program planning are not seen as being of primary importance.

b. Inclusiveness of scope. A review of the areas of concern of the various Preventive Social Service projects (see Table II) shows that the Society for the Retired and Semi-Retired the Strathcona Place Society, Operation Friendship, the Norwood Community Service Centre and the Dickensfield We Care Center are dealing with more than six problems areas each, ranging from child care and recreation to information provision etc. Teen Time, Edmonton Big Sisters, Family Aide, the Family Life Education Council of Edmonton, the Edmonton Family Planning Service, the Urban Life Skills, and Child Care Program, the Emergency Homemaker's Service and the St. Francis Youth Club are providing a moderate range of services. While Open Door, Home Care, A.I.D., Day Care, Meals-on-Wheels, and the Bissell Child Care Centre are dealing with relatively narrow areas.

With regard to the Community Social Services which is not included in the table, the scope of activities varies from unit to unit. But most of the units are providing a wide range of services like



marriage counselling, youth programs and so on.

ଚ

· · · · · · · · ·

ø

In fact, the scope of the activities of most projects in this study has been broadened since the inception of the operation. For example, A.I.D. has provided the distress line in addition to the information service. Open Door has also taken the responsibility

able 2 · · · · ·

125

The Scope of the P.S.S. Projects

Areas of Concern	Projects*
Areas of Goncern	F N D S T B O S O M A E U E S E H F D L C W R T C D P F O I F L H F B C A C E S C S C S W D P S S Y S C C C R C , S & C E C
Family life	<u>x x x x x</u> .
Communication skills	xxx
Personal enrichment	x x x x x x
Neighbourhood	
Improvement (physical	
social)	x x x x x x x x x x x x x x x x x x x
Child care	
Récreation	<u>x x x x x x x x x x x x x x x x x x x </u>
Information provision	××××××
(referral)	
Education	
Housing & transportation	
Social activities	
Cultural enrichment	× × × × × × × × × × × × × × × × × × ×
Youth	· · · · · · · · · · · · · · · · · · ·
Elderly	x x xx xx
Health	
Meals - service	× × × × × ×
Counselling	
Life skills	
Outreach, Home Assistance	

\*(see Chapter V for full names of the projects)
N.B. (Community Social Services is not included in this diagram.)

• 5

of Area 13 News and co-ordinated the oBlock Parents/Good Neighbours program for the area after the information and referral had been set up. Also, those projects identified as dealing with a wide range of issues, have already expanded their scope of activities since the inception of the operation. In sum, most of the projects in this study bear a resemblance to successful community development projects which, as described in the literature, have continually expanded their scope of activities and substituted new-level problems for old ones.

126

c. <u>Mobilization of human resources</u>. Based on figure 23, A.I.D., the Family Life Education Council of Edmonton, the Norwood Community Service Centre, Meals-on-Wheels, the Strathcona Place Society and the

# Figure 23

Mobilization of Human Resources

(approx. numbers of volunteers involved in the delivery of services)

C (	50+		0 · · ·	0 °
Many	<b>h</b>	· · ·	<u>^</u>	1
	M.O.W.	D.W.C.C.∝ 。	B.C.C.C.	D.C.
	A.I.D.	S.F.Y.C. 0.D.		н.С.
•	F.L.E.C.E. N.C.S.C.	E.F.P.S.		U.L.S.&C.C. E.H.S.
	S.P.S.	0.F. E.B.S.	5 C	F.A.
0	S.R.S.R.	<b>T</b> . <b>T</b>	a	<i>۵</i> ۵

Society for the Retired and Semi-Retired have involved a large number of volunteers to help in the delivery of services. The Bissell Child Care Centre has recruited a few volunteers while the Dickensfield We Care Centre, the St. Francis Youth Club, Open Door, the Edmonton Family Planning Services, Operation Friendship, Edmonton Big Sisters and Teen Time have involved relatively a fair amount of volunteers in the delivery of services. Finally, the services of Home Gare, Day Care, Urban Life Skills and Child Care Program, the Family Aide and the Emergency Homemaker's Service are provided by paid staff; there is no volunteer involvement at this level. As the recruitment and training of volunteers is an important element of community development, these five projects cannot be considered as community development activities,

The above analysis also reflects that the level of volunteer involvement largely dependent upon the individual philosophy of the projects rather than the scope of the projects. For example, there are lots of volunteers involved in A.I.D., the Family Life Education Council of Edmonton, Meals-on-Wheels although these projects are dealing with a relatively narrow range of issues. On the other hand, there are also a lot of volunteers involved in the three projects which are working for senior citizens, which may be due to the fact that most senior citizens, as compared to the other target population, have more leisure time to participate in various activities.

With regard to the Community Social Services project, the Duggan Unit has involved a comparatively large number of volunteers because it has an extensive training program. The other local units

127

have either used a moderate amount of volunteers or no volunteer involvement at all.

128

# 4. Organizational Structure

The participation of citizens or service users in drganizational decision-making is considered as one means of promoting citizens' needs and protecting citizens' interests. In keeping with the principles and practice of community development it is imperative that citizen participation at all levels of operation be encouraged and local leadership be developed. To date there have been two major methods of implementing citizen participation. The first has taken the form of recruiting and training of community members as volunceers in the service; this has already been discussed in the preceding sections. The second involves representation of community members on Boards of Directors and Advisory Committees. This aspect of citizen participation will be the focus of analysis in the following pages.

Obviously, all the non-governmental projects in this study have chosen the form of Directive Boards or Committees to ensure citizen involvement at the level of policy decision-making. For many of these non-governmental projects, Advisory Committees are also established to facilitate the planning process or to provide an advisory support base for the Board of Directors mainly composed of agency representatives and interested citizens. With regard to the four governmental projects, the Home Care Program is jointly administered by a Home Care Management Committee composed of staff members from the Local Board of Health and the Edmonton Social Services. There is an Advisory Committee of representatives appointed by both departments to assist in theon-going development of the program. There is also a Technical Subcommittee made up of representatives from various-service agencies to assist the Advisory Committee. For Day Care Program, with the exception of the Glengarry Day Care Center which is administered by the Department, the other subsidized Day Care Centers are managed by their own Board of Directors. Finally, since both Family Aide and Community Social Services are administered directly by the Social Service Department; their policy will be influenced by the Social Service Advisory Committee to the Department.

# Figure 24

Arnstein's Eight-rung Typology of Citizen Participation

(control of decision

making)

Degree

High

Citizen Control Delegated Power Partnership

Levels of Citizen Power

Placation

Consultation Informing

Therapy

(no responsi-Low billity for decision making)

Manipulation

Non-Participation

Degree of tokenism

# Arnstein's Typology
# According to Arnstein's eight-rung typology which is

illustrated in Figure 24, projects with substantial representation of citizens or clientele on the Policy Board or Committee which is the decision-making body for overall policy, have demonstrated a high level of citizen participation. The reason is that citizens have legitimated power to make key decisions and control the administration of the program. ("Power", in an organizational context, indicates a degree of control over the resources and actions of the organization, e.g. funds, staff, facilities, etc.)

A review of the composition of the Boards of Directors of various Preventive Social Services projects shows that the Board of the Family Life Education Council of Edmonton is composed of representatives of agencies and association and interested citizens. Half of the board members of Norwood Community Service Centre are people from the community, e.g. parents of pre-school children, and the rest are professional people. The Boards of both the Dickensfield We Care Centre and Open Door are made up of local residents. For the three projects which are working for senior citizens, the Board members of the Society for the Retired and Semi-Retired are mostly professionals. Half of the Board members of Strathcona Place Society are non-senior citizens, . but influential persons in the community. " Operation Friendship has also recruited some professionals plus two senior citizens in the area as Board Members. The Bissell Child Care Committee is a group of staff persons from Park and Recreation Department and City Social Services, and a few interested parents. Both Teen Time and the

130

Boys' Club of Edmonton have involved people from different walks of life or concerned citizens as their Board members; there is no representation from the youth groups. Edmonton Big Sisters has also recruited citizens-at-large, social workers and a few Big Sisters on the Policy Board. While the Advisory Committee of the Emergency Homemaker's Service is composed of a wide representation of community agencies and service programs, the Boards of Management or Boards of Directors of the Victorian Order of Nurses, A.I.D., and the Edmonton Family Planning Service are made up of a cross-section of the community, including people from other agencies. Finally, the Board of Directors of the Canative Housing Corporation is mainly composed of native people.

It is evident that there are relatively high levels of citizen participation over most of the non-governmental projects. For example, the Boards of Directors of the Family Life Education Council of Edmonton, the Dickensfield We Care Centre, Open Door, the Emergency Homemaker's Service, Victorian Order of Nurses, A.I.D., the Edmonton Family Planning Service and the Canative Housing Corporation are made up of community members, clientele or community representatives. According to Arnstein's concept, these projects have demonstrated the level of 'citizen control'. On the other hand, the Boards of Directors of the Norwood Community Service Centre, Teen Time, the Society for the Retired and Semi-Retired, the Strathcona Place Society, Operation Friendship, the Bissell Child Care Committee, the Edmonton Big Sisters Society and the Boy's Club of Edmonton are made up of professional people and some local residents or community members. Citalzen participation in these projects is at the level of 'delegated power' because the decision-making responsibilities are shared with other non-members. In fact, there are advantages and disadvantages of electing professional people as Board members. One advantage is that these professionals can act as resource persons to the representatives of the

132

group, and their skills can be an asset to the project. In order to establish a program, one needs fund and often the influential and knowledgeable people in the community are most capable of procuring the necessary support. However, sometimes, the most educated and articulate members of the board may overwhelm the other members of the board, and that their motions are accepted regardless of whether or not they are the best choice for the clientele or the members.

Another form of involvement which has been employed by the four governmental projects: Family Aide, Home Care, Day Care, and Community Social Services, is to draw the Advisory Committee partly from the community. In this form, of course, Committees lack the authority to make binding decisions but the Advisory Committee is often requested to review and recommend certain applications or policies. If a cooperative relationship exists between the Department and the representatives, the Advisory Committee may be effective in representing the community. However, if the Department has no intention of taking. the advice seriously, the Advisory groups are likely to be impotent. Therefore, in Arnstein's concept, the Advisory form is at the level of 'placation', and at this level, citizens begin to have some degree of influence though toRenism is still apparent.

Summary

133

This chapter is diviged into two sections. In the first section, the analytical framework for the study is described. In the second section, the analysis of the Preventive Social Service projects in

Edmonton is presented. An analysis of the initiation of the projects reveals that there were high levels of community efforts in the initiation of fifteen projects. Most of the needs inherent in the objectives of the projects are identified as 'social needs' for personal effectiveness rather than for basic survival. An examination of the basic objectives of the projects shows that most of the stated intent of the projects fall within the conception of primary prevention. Only a few of the objectives "aim at secondary prevention. On the other hand, the analysis of the development of the objectives of the projects indicates that fourteen projects in this study were basically determined by community members or representatives. There was high influence by the people in the "development of the objectives.

With regard to the types of celientele served by the projects, eight projects serve the geographic communities while seven work with functional groups. The other five projects work only with individuals or families who are selected by the criteria set by the projects. Of the methods of work, only the community organization service which is under the Community Social Services project is regarded as purely process-oriented. The methods of work of four other projects are also regarded as process-oriented. Seven projects place emphasis on both process-oriented and task-oriented approach. The methods of work of the other eight projects are basically task-oriented. With regard to the inclusiveness of scope, the Preventive Social Services projects range from single-issue or specialized projects to generalized projects dealing with more than six issue-areas. In terms of mobilization of human resources, six projects have involved a large number of of volunteers while seven have involved a fair number of volunteers. Five projects have no volunteer involved in the delivery of services.

134

Finally, most non-governmental projects have chosen the form of Directive Boards to ensure citizen participation at the policy decision-making levely. In terms of Arnstein's eight-rung typology, most non-governmental projects have demonstrated a high level of

citizen control.

In the next chapter, the summary of the findings, the conclusion reached in the study and the implications of the study will be presented. Footnotes

135

<sup>1</sup>A. Montagu, <u>The Direction of Human Development</u>, New York, Hawthorn Books, 1970, pp. 150-151.

<sup>2</sup>A.H. Maslow, "Some Theoretical Consequences of Basic-Need Gratification", <u>Journal of Personality</u>, vol. 16, 1948, pp. 402-416.

<sup>3</sup>A.H. Maslow, "'Higher' and 'Lower' Needs", <u>Journal of Psychology</u>, vol. 25, 1948, pp. 433-436.

<sup>4</sup>Arthur Dunham, <u>The New Community Organization</u>, New York, Thomas Y. Crowell Company, 1970, p. 188.

5 Roland Warren, <u>The Community in America</u>, Chicago, Rand McNally, 1964.

# CHAPTER VII

### SUMMARY, CONCLUSIONS AND IMPLICATIONS

The purposes of this final chapter are 'to summarize the findings and to present the major conclusions and implications of the study.

#### Review

The basic objective of this thesis was to study the various aspects of the Preventive Social Services projects in Edmonton as to what extent these projects have been employing the principles and methods of community development.

Two major methods of data collection were employed to obtain information for the study. The basic information for the study, which includes the descriptions of the twenty Preventive Social Services projects, was obtained through personal interviews with the key personnel in respective projects; and also through a review of accessible reports and documents,

Since the identification of a project as community development, is based on the objectives of the project, the issues involved, the types of clientele served, the method employed and so on, an analytical framework by which to examine the above aspects of the project was developed from the review of literature on community development. Analyses of the projects included: the level of participation on the part of the

136

community (people and groups) in the initiation of the projects, the needs inherent in the objectives, the preventative versus rehabilative orientation of the objectives, the development of the objectives, the types of clientele, the process-orientation versus task-orientation of the methods of work, the inclusiveness of scope, the mobilization of human resources, and citizen participation at the level of policy decision-

On the basis of the analysis, projects were considered community development when they were operated within the given framework.

making.

#### The Findings

The findings in this chapter are based on the analysis of the projects presented in the preceding chapter.

The analysis reveals that there were high levels of community efforts in the initiation of fifteen projects. In other words, local initiatives were evident in seventy per cent of the projects studied, which are considered as desirable in community development concepts.

In analyzing the objectives of the projects, most of the needs inherent in the stated objectives are 'social needs' or 'growth needs' for enhancement and enrichment of individuals rather than 'vital, needs' for survival. They are basically 'higher' meeds as defined by Maslow for strengthening human growth. Obviously, ninety per cent of the projects place heavy emphasis on social development rather than on the improvement of the material conditions of life. As the setting of this

1 8

study is an urban industrial society with a high level of standard of living, the objectives of the project are, therefore, likely to be social objectives for facilitating the social development process conducive to a balanced human and material betterment.

An examination of the present objectives of the projects also shows that most of the objectives fall within the conception of primary prevention. Only a few of the objectives are considered as secondary prevention objectives. None of the objectives of the projects in this study aim at tertiary prevention. In other words, most of the projects are directed towards prevention which is emphasized as most effective and economic in community development planning.

On the other hand, the analysis also indicates that the objectives of fourteen projects in this study were developed by community members or representatives. Therefore, there was high influence by the people in the development of the objectives. It is stressed in the principles of community development that the goals and needs of local communities should not be externally determined or centrally planned.

In this study, most of the projects are designed to serve the total population of the City or neighbourhoods or of certain groups in the City or area, i.e. thergeographic and functional communities. However, there are five projects which are restricted to certain individuals or families with specific needs. As the elientele served by these five projects are assessed and selected by certain criteria, these projects are not serving the geographic nor functional communities.

With regard to the methods of work, only five projects are

1 2 0

considered as process-oriented. Most of the projects are either process and task-oriented or basically task-oriented.

139

An analysis of the inclusiveness of scope of the projects, shows that projects range from Single-issue or specialized projects to generalized projects dealing with more than six issues. But the two neighbourhood centres and the three projects serving senior citizens tend to be generalized projectS concerning many issues or providing many kinds of services. In terms of mobilization of human resources, six projects have a large number of volunteers involved to help in the projects, while five have not utilized any volunteers in the delivery of services. The others have recruited a fair size of volunteers, ranging from ten to twenty-five people.

Most non-governmental projects have demonstrated a high level of citizen participation at the policy decision-making level and chosen the form of Directive Boards.

### Non-Governmental Projects

Of the projects studied, the Family Life Education Council of Edmonton, the Society for the Retired and Semi-Retired and the Strathcona Place Society are found very similar to community development activities. First of all, the objectives of these three projects are primary prevention objectives and were developed by community members or local citizens. These projects are open to any member of either a geographic or a functional community. Their methods of work are processoriented and place heavy emphasis on the recruitment and the training of volunteers. These projects, in fact, have involved a large number of volunteers. While both the Society for the Retired and Semi-Retired, and Strathcona Place Society are generalized ones dealing with a wide range of issues, the Family Life Education Council of Edmonton focuses on a few issues. These three projects have citizen controlled boards of policy-making. But in both the Socfety for the Retired and Semi-Retired, and Strathcona Place Society, there are many professionals.

The analysis also shows that the Norwood Community Service Centre, Dickensfield We Caré Centre, Operation Friendship, Open Door, Teen Time, St. Francis Youth Club and Edmonton Big Sisters have been ° employing some of the principles and methods of community development to a large extent. The objectives of these projects, with

exception of the Edmonton Big Sisters, fall clearly with conception of primary prevention. The objectives of the Edmonton Big of ters aim at primary as well as secondary prevention. With the exception of the St. Francis Youth Club and Teen Time, the objectives of the above projects were basically developed by local people or community etc. All of these projects work either with the people in a neighbourhood or with groups, i.e. the geographic and functional community.

The methods of work of these projects place equal emphasis on the process-oriented and task-oriented approach. Edmonton Big Sisters, in fact, is identified as more process-oriented. Volunteer involvement is also emphasized in their methods of work. These projects have involved a fair number of volunteers to assist in the projects. Of these projects, Norwood Community Service Centre, Dickensfield We Care Centre, Operation Friendship are more generalized projects dealing with a wide range of issues, and the others provide a moderate

141

All these projects have chosen the Directive Board to ensure citizen participation. In fact, these projects have demonstrated a high level of citizen participation at the policipation-making level with citizen controlled boards.

All the others, Meals-on-Wheels, Bissell Child Care Centre, A.I.D., Urban Life Skills and Child Care Program, Edmonton Family Planning Service and Emergency Homemaker's Service have been employing, only to

a certain extent, some of the principles and methods of community development. Basically, the objectives of these projects were developed by the community people. While the objectives of Edmonton Family Planning Service, Bissell Child Care Centre and Urban Life Skills and Child

Care Program are primary prevention objectives, the objectives of Meals-on-Wheels; Emergency Homemaker's Service aim at secondary prevention. The objectives of A.I.R. aim at both primary and secondary prevention.

Of these projects, the clientele served by the Urban Life Skills and Child Care Program, Edmonton Family Planning Service, Bissell Child Care, and A.I.D. are considered as either geographic communities or functional communities. But the Emergency Homemaker's service and Mealson-Wheels serve only selected individuals or families with special needs. Their clientele served are not considered as a community. The methods of work of these six projects are basically taskoriented. Also, the scope of the activities of these projects is relatively narrow. Of these projects, A.I.D. and Meals-on-Wheels have recruited a large number of volunteers in the delivery of services. The Edmonton Family Planning Service and the Bissell Child Care Centre have also emphasized the involvement of volunteers. But both the Emergency Homemaker's Service and Urban Life Skills and Child Care Program have not utilized and volunteer in the delivery of service. All the programs or activities are conducted by staff.

In terms of citizen participation at the policy decisionmaking level, these projects have chosen the form of Directive Boards or Committees which are controlled by local citizens or community members. All six projects have demonstrated a high level of citizen participation at this level.

### Governmental Projects

Of the projects operated under the auspices of the Edmonton Social Service Department, the community organization service (part of the Community Social Service) is the only service considered as similar to community development. First of all, all the objectives of the four projects were developed by the Department although the objectives of Day Care and Home Care are based on 'community needs'.

All these projects, except Day Care and Community Organization Service, aim at secondary prevention. The clientele served by Day Care, Home Care, and Family Aide are available only to selected individuals and families, not all the people in a community. Community Social Service (which includes counselling service and community organization service) serves seven geographic communities.

The methods of work of Home Care and Family Aide are basically task-oriented. But Day Care has employed a process and task-oriented approach. Community organization service, on the other hand, is process-criented. All these projects place no emphasis on volunteer involvement in the delivery of services, and their scope of activities is varied.

All these governmental projects, with the exception of Day Care, have not demonstrated a high level of citizen participation at the level of policy decision-making because Advisory Committee is the only form used to ensure citizen participation.

### Conclusions

Based on the findings of this study, eighteen per cent of the projects can be identified as community development activities. These projects include: the Strathcona Place Society, the Society for the Retired and Semi-Retired, the Family Life Education Council of Edmonton, and the Community Organization Services.

Thirty-five per cent of the projects which include the Norwood Community Service Centre, Dickensfield We Care Centre, Operation Friendship, Open Door, Teen Time, St. Francis Youth Club, and Edmonton Big Sisters have been employing some principles and methods of community development to-a large extent.

144

On the other hand, fifteen per cent of the projects including A.I.D., Edmonton Family Planning Service and Bissell Child Care Centre have shown a few characteristics of a community development activity. The rest of the projects or thirty-two per cent of the projects in this study, which include Meals-on-Wheels, Urban Life Skills and Child Care, Home Care, Family Aide, Emergency Homemakers Service, Community Counselling Service, and Day Care are not considered as community development activities.

The findings in this study also indicate that the objectives of most Preventive Social Services are preventative and are basically social objectives because they are based on people's 'emotional needs' and 'social needs'.

With regard to citizen participation, the author has mentioned earlier that as a result of the size of the population in the urban areas and the differentiated social structure, the participation of the total population is difficult to attain. Therefore, citizen participation in these Preventive Social Service projects has taken two major forms:

 Representation from the community serving on Boards of Directors.
 Recruiting members of the community to serve as advisory committees. The findings also reveal that the Preventive Social Services
 projects are, basically, not government 'laid-on' projects because
 there was high level of community effort in the initiation of the

projects and high influence by the community in the development of the objectives.

. Preventive Social Services is also found quite different from

the Statutory Social Services that it aims at providing services of preventive nature and has local commitment, input or volunteerism and local control. As shown in the findings, it tends to create climate for citizen involvement. The variety of Preventive Social Services in this study also reveals that the Preventive Social Service Program tends to encourage exploration of local problems, and the development of programs designed to prevent problems.

### Implications

This study reflects that the community development method affords an additional and potentially powerful intervention strategy in addition to the traditional casework method. It can be an effective means and flexible approach to meet the diversified and ever changing needs in social services. The application of community development, first applied to the rural and developing areas, has been widened and seems to become more direct in an evolving approach to general welfare problems in urban society.

The study also reveals that community development is more effective in projects dealing with attitudes and services of a preventive and educational nature than with crisis and rehabilitative situations. The reason is that the clientele served by the latter type of project, are mostly selected individuals with special needs and problem at a certain time (e.g. Homemaker's Service, Family Aide, etc.). It is difficult to achieve an identification of the population concerned, or of the total problem-group partly due to the

145

less manifest sense of belonging to a certain community. It is hard to initiate community development where there is little sense of community.

On the other hand, this study reveals that the Preventive Social Services Program does present a very promising situation for community development. Two factors as illustrated in the study are considered as the significant areas in creating favorable situations conducive to community development.

The first is the network, not hierarchy, of decision-making that has been developed. It is the mark of success of the Preventive Social Services Program that this network in most communities is well established. It has achieved a good deal of decentralization in decision-making.

The second factor is that the projects have illustrated an increasing degree of cooperation with other services and organizations.

The study also suggests that citizens, when offered the opportunity, will participate in improving the environment and solving their local problems.

Another implication of this study concerns the dual responsibility of the Municipal Social Service Department. The Municipal responsibility is to deliver service under its auspices and on the ° other hand, to find and encourage submission of private proposals. There is a competition for funding between the governmental and the community projects. In other words, a conflict of interest is embodied in the dual responsibility for developing services and funding. Therefore the City Department should avoid delivering preventive social services under its own auspices for such a role may induce competing claims and interests between the governmental and the non-governmental sectors.

147

### Concluding Remarks

As stated in the objectives of this thesis, the author has provided the descriptions of the Preventive Social Services Program and twenty funded Preventive Social Services projects in Edmonton, and has examined the extent to which these projects have been employing the principles and methods of community development. But it must be stressed that the analysis of this study should be regarded as the author's perceptions and personal opinions based on the information 'obtained from the key personnel of respective projects.

Also due to the limitations of resources, time and accessible information, the scope of the study is rather narrow. For a better understanding of the Preventive Social Service Program, research in depth is meeded.

#### BIBL IOGRAPHY

148

#### Books

Abueva, J.V., Focus on the Barrio, Institute of Public Administration, University of the Phillipines, 1959.

Anderson, Nels, <u>The Urban Community, A World Perspective</u>, New York, Holt, Rinehart & Winston, 1959.

Armitage, Andrew, <u>Social Welfare in Canada, Ideals and Realities</u>, Toronto, McClelland and Stewart Ltd., 1975.

Baltzell, E. Digby (ed.), <u>The Search for Community in Modern America</u>, New York, Harper & Row, 1968.

Becker, Howard S. (ed.), <u>Social Problems: A Modern Approach</u>, New York, John Wiley & Sons, 1966.

Biddle, William J. and Laureide J. Biddle, <u>The Community Development</u> <u>Process, The Rediscovery of Local Initiative</u>, New York, N.Y., Holt, Rinehart & Wintson, 1965.

Biddle, William J. and Laureide J. Biddle, <u>Encouraging Community</u> <u>Development: A Training Guide for Social Workers</u>, New York, N.Y., Holt, Rinehart and Winston, 1968.

Blishen, Bernard R., and Frank E. Jones, Kasper Naegele, John Porter, <u>Ganadian Society: Sociological Perspectives</u>, Toronto, Macmillan of Canada, 1968.

Bloomberg, Warner Jr., and Henry J. Schmandt, <u>The Quality of Urban Life</u>, Beverly Hills, Calif., Sage Publications Inc., 1970.

Brager, G. and H. Specht, <u>Community Organization</u>, New York, Columbia U. Press, 1973.

Brokensha, David, and Peter Hodge, <u>Community Development: An Interpre-</u> <u>tation</u>, San Francisco, Chandler Publishing Co., 1969.

Bruyn, S.T., <u>Communities in Action: Pattern and Process</u>, New Haven, Conn., College & University Press Services Inc., 1986.

Cahn, E.S. and B.A. Passett, <u>Citizen Participation: Effecting Community</u> <u>Change</u>, New York, Praeger Pub., 1971. Cary, Lee J. (ed.), <u>Community Development as a Process</u>, Columbia, Missouri, U. of Missouri Press, **2**70.

Clinard, Marshall B., <u>Slums and Community Development: experiments</u> in self-help, New York, The Free Press, 1966.

Community Development in Alberta, Ottawa, Special Planning Secretariat, Privy Council Office, 1965.

<u>Community Planning & Community Development, Some common distinct</u> <u>elements</u>, Council on Social Work Education, 1961.

Connor, Desmond M., Strategies for Development, Ottawa, Development Press, 1968.

Cox, Fred M. and others, <u>Strategies of Community Organization: a book</u> of <u>Readings</u>, Ithaca, Ill., Peacock, 1970.

Dasgupta, S., A Poet and an Plan, Calcutta, Thacker Spink, 1963.

De Coca, G.A., <u>Citizen Participation: Doomed to Extinction of last</u> <u>foothold of democracy</u>? Synthoff-Leyden, 1969.

Di Franco, J. <u>A Collection of Prince and Guides</u>, New York, New York State College of Agricultur, 1958.

Draper, James A. (ed.), <u>Citizen Participation</u>; <u>Canada</u>, Toronto, New Press, 1971.

Dunham, Arthur, <u>The New Community Organization</u>, New York, Thomas Y. Crowell Co., 1970.

Dyck, Harold J. and others, <u>Social Development Planning in Alberta</u>: <u>A Preliminary Study</u>, Government of the Province of Alberta, Dept. of Health and Social Development, 1970.

Franklin, Richard (ed.), <u>Patterns of Community Development</u>, Washington, D.C., Public Affairs Press, 1966.

Gandhi, M.K., <u>Constructive Programme - Its Meaning and Place</u>, Ahmedabad, Navjivan Publishing House, Reprint, 1948.

Government of Alberta, <u>Social Development: A Position Paper</u>, Alberta, Edmonton, April, 1969.

Great Britain, Colonial Office, <u>Community Development: A Handbook</u>, London, Her Majesty's Stationery Office, 1958.

Great Britain, Reference Division, Central Office of Information, <u>Community Development in the United Kingdom and Dependencies</u>, London, Overseas Information Services, 1958.

150

Harper, Ernest B., and Arthur Dunham (eds.), Community Organization in Action, New York, Association Press, 1968.

- Hayes, Samuel P. Jr., Measuring the Results of Development Projects, Paris, France, UNESCO, 1959.
- Kahn, Alfred J., Social Policy and Social Services, New York, Random House, 1973.
- Kramer, R.H. and Harry Specht, Readings in Community Organization Practice, Englewood Cliff, Prentice Hall, 1969.
- Leavell, H.R. and Clark, E.C., Preventive Medicine for the Doctor in His Community, New York, McGraw-Hill, 1965.
- Lloyd, Anthony J., Community Development in Canada, Ottawa, Canadian Research Centre for Anthropology, St. Paul University, 1967.
- Manning, Ernest C.A., White Paper on Human Resources Development, Alberta, Edmonton, 1967.
- Mial, Dorothy and H. Curtis Mial (eds.), Forces in Community Development, Washington, D.C., NTL Institute for Applied Behaviorial Science, 1961.
- Minar, David W. and Scott Greer (eds.), The Concept of Community, Chicago, Aldine Publishing Co., 1969.
- of Rental Development, New York, Hawthorn Montagu, A., The Direction Books, 1970.
- Morgan, John S., Changing Services For Changing Clients, NASW, New York, Columbia University Tress, 1969.
- National Association of Social Workers, Champing Services for Changing Clients, NASW, New York, Columbia University Press, 1969.
- Naylor, Harriet H., Volunteers-Finding. Training, and Working With Them, New York, Association Press, 1967.
- New Strategies for Social Development, Role of Social Welfare, Proceedings of the XVth International Conference on Social Welfare, New York, Columbia University Press, 1971.

. . .

- Oliver, M. (ed. , Social Purposes for Canada, Toronto, University of Toronto Press, 1961.
- Pande, V.P., Village Community Projects in India, London, Asia Publishing Eg., 1961.

. ? ¥

Perlman, R. and A. Gurin, <u>Community Organization and Social Planning</u>, New York, John Wiley & Sons, 1972.

Preventive Social Services, Administrative and Policy Outline, Dept. of Health and Social Development, Alberta, Edmonton, 1970.

Romanyshyn, John M., <u>Social Welfare</u>, Charity to Justice, New York, <sup>©</sup>Random House, C.S.W.E., 1970.

Ross, Murray G., <u>Community Organization, Theory, Principles and</u> <u>Practice</u>, N.Y., Harper & Row, 197.

Rossides, Daniel W., <u>Voluntary Participation in Canada, A Comparative</u> <u>Analysis</u>, Toronto, York University, 1966.

Ruopp, Phillips (ed.), <u>Approaches to Community Development</u>, the Hague, W. Van Hoeve Ltd., 1953.

Sanders, Irwin T., <u>The Community: An Introduction to Social System</u>, New York, N.Y., The Ronald Press Co., 1966.

Sen, Sudhir, <u>Rabindranath Tagore on Rural Reconstruction</u>, VisvaBharati, 1943.

Shostak, Arthur B. (ed.), <u>Sociology im Actions. Case Studies in Social</u> <u>Problems and Directed Social Change</u>, adelphia, U. of Pennsylvania Press, 1966.

Spergel, Irving A. (ed.), <u>Community Organization, Studies in Constraint</u>, Beverly Hills, Sage Publications, 1972.

Spiegel, Hans B.C. (ed.), <u>Citizen Participation in Urban Development</u>, <u>Vol. 1 Concepts and Issues</u>, Washington, D.C., Institute for Applied Behavioral Science, 1968.

Spiegel, Hans B.C. (ed.), <u>Citizen Participation in Urban Development</u> <u>Vol. II, Cases and Programs</u>, Washington, D.C., NTL Institute for Applied Behavioral Science, 1969.

Teare, Robert and Harold McPheeters, <u>Manpower Utilization in Social</u> <u>Welfare</u>, Alberta<sup>°</sup>, Southern Regional Education Board, 1970.

Turner, John B. (ed.), <u>Neighbourhood Organization for Community Action</u>, New York, National Association of Social Workers, 1968.

U.N., Bureau of Social Affairs, <u>Social Programs Through Community</u> <u>Development</u>, New York, United Nations, 1955.

U.N., Dept. of Economic and Social Affairs, <u>Popular Participation in</u> <u>Development: Emerging Trends in Community Development</u>, New York, U.N., 1971. Urban and Rural Community Development, Proceedings of the XIth International Conference of Social Work, 1962.

152

- Warren, Roland, <u>The Community in America</u>, Chicago, Rand McNally, 1964.
  - Warren, Roland L., <u>Perspectives on the American Community</u>, Chicago, Rand McNálly & Co., 1969.
- Warren, Roland L., <u>Truth, Love and Social Change</u>, Chicago, Rand McNally & Co., 1971.
- Weiss, Carol H., <u>Evaluation Research: Methods of Assessing Program</u> <u>Effectiveness</u>, Englewood Cliffs, New Jersey, Prentice-Hall, Inc., 1972.
  - Wharf, Brian and Novia Carter, <u>Planning for the Social Services</u>: <u>Canadian Experiences</u>, Ottawa, Canadian Council on Social Development, 1972.
  - Wileden, Arthur F., Community Development, The Dynamics of Planned Change, N.J. Totowa, The Bedminister Press, 1965.
  - Wilensky, H. and C. Lebeaux, <u>Industrial Society and Social Welfare</u>, New York, Russell Sage Foundation, 1957.
  - Zaid, Mayer N. (ed.), Organizing for Community Welline, New York, Quadrangle Books, 1967.

Articles

- Anderson, N., "Rethinking Our Ideas About Community," <u>International</u> <u>Review of Community Development</u>, 1962, No. 10, pp. 143-152.
- Bottum, C.J., "Community Development Operation," American J. of Agricultural Economics, Vol. 50, No. 5, Dec., 1968, pp. 1239-1246.
- Bregha, Francis J., "Community Development in Canada, Problems and Strategies," <u>Community Development Journal</u>, Vol. 5, No. 1, Jan. 1970, pp. 30-36.
- Buell, B., "Community Planning for Health and Welfare Services," <u>International Review of Community Development</u>, 1959, No. 4, pp. 153-185.

at The Concepts of Community and Community Organization, A Review," <u>Social Work</u>, XV, 3, July, 1970, pp. 82-92.

DuSautoy, P. and R.D. Waller, "Community Development and Adult Education in Urban Areas," <u>International Review of Community Development</u>, 1961, No. 8, pp. 33-50.

Dunham, A., "Some Principles of Community Development," <u>International</u> <u>Review of Community Development</u>, 1963, No. 11; pp. 141-148.

Dunham, A., "The Outlook for Community Development--An International Symposium," <u>International Review of Community Development</u>, 1960, No. 5<sub>3</sub>. pp. 33-35.

Eaton, J.W., "Community Development Ideologies," <u>International Review</u> of Community Development, 1963, No. 11, pp. 37-63.

Foskett, J.N., 'The Influence of Social Participation on Community Programs and Activities," in M.B. Sussman (ed.), <u>Community Structure</u> <u>and Analysis</u>, New York, N.Y. Thomas Y. Crowell Co., 1959.

Government of Alberta, "The Preventive Social Services Act," in <u>Revised Statutes of Alberta</u>, Government of Alberta, 1970; Chapter 282, Sec. 3.

Groenman, S., "Community Development in Urban Areas," <u>International</u> <u>Review of Community Development</u>, 1961, No. 7, pp. 61-69.

Hillery, George A. Jr., "Definitions of Community: Areas of Agreement," <u>Rural Sociology</u>, 20, June, 1955, pp. 111-23.

Hynam, C.A.S., "Community Development, An Example of Conceptual Confusion," B.Y.Card (ed.), <u>Perspectives on Regions and Regionalism</u>, Western Association of Sociology and Anthropology Proceedings, 1968.

Janes, R.W. "Measures of Effective Community Development: An Appraisal of Community Action as a Social Movement," <u>International Review</u> of Community Development, 1961, No. 8, pp. 5-13.

Kaufman, H.F. and L.W. Cole, "Sociological and Social Psychological Research for Community Development," <u>International Review of</u> <u>Community Development</u>, 1959, No. 4, pp. 193-211.

Lotz, Jim, "Thoughts on Community Developers," <u>Adult Leadership</u>, 1972, September, pp. 79-81.

Lyfield, W.G., and W.H. Schmidt, "Trends in Community Development--Some Results of a Survey," <u>International Review of Community</u> <u>Development</u>, 1959; No. 4, pp. 33-40. Maslow, A.H., "'High' and 'Lower' Needs," Journal of Psychology, Vol. 25, 1948, pp. 433-436.

Mial, C., "Community Development -- U.S.A.," <u>International Review of</u> <u>Community Development</u>, 1959, No. 4, pp. 13-19.

Miniclier, L.M., 'Value's and Principle's of Community Development," <u>International Review of Community Development</u>, 1960, No. 5, pp. 57-61.

Nagpaul, H., "Conformity and Community Development," <u>International</u> <u>Review of Community Development</u>, 1963, No. 11, pp. 105-118.

Ponsioen, J.A., "Community Development as a Process," <u>International</u> <u>Review of Community Development</u>, 1960, No. 6, pp. 2038.

Rogers, M.L., "Autonomous Groups and Community Development in the U.S.A.," <u>International Review of Community Development</u>, 1959, No. 4, pp. 41-56.

Rogers, W.C., "Voluntary Associations and Urban Community Development," <u>International Review of Community Development</u>, 1967, No. 87, pp. 135-145.

Sanders, Irwin<sup>°</sup>T., "Theories of Community Development," <u>Rural Sociology</u>, 1958, Vol. 23, pp. 1-12.

Stensland, G., "Some Prerequisites for Community Development," International Review of Community Development, 1960, pp. 81-94.

tton, W.A. and J. Kolaja; "Elements of Community Action," <u>Social</u> Forces, 1968, 38, May, pp. 325-331.

Toussaint, W.D., "Discussion: The Theory and Strategy of Community Development," <u>American Journal of Agricultural Economics</u>, Vol. 50, No. 5, Dec., 1968, pp. 1247-1251.

"Toward Understanding in Prevention," The Gity of Edmonton, Social Service Department.

### APPENDIX A

155

# PREVENTIVE SOCIAL SERVICES PROJECTS

#### List of Questions

### 1. Historical background

- a) Who initiated the project?
- b) Who expressed and defined the needs of the project?
- c) How was it established?
- d) Who has been involved in developing the project?

### 2. Objectives

- What needs was it designed to meet initially?
   Were the objectives ever defined?
- b) Since the inception of the project, have there been changes in objectives?
- c) If so, what are they? How did this come about? Who was involved?

### 3. Clientele

or

- a) Who is your clientele or community served? 1/ total geographical community?
  - 2/ certain community segments?

### Implementation strategies

- a) What methods of strategies do you use to achieve the objectives?
- What activities have been designed for achieving the objectives?
- b) Since the inception of the project, have there been changes in methods used?
  - c) If so, why were these changes made?

- 5. Community involvement and participation
  - a) In what ways do citizens or client groups participate in the project?
  - b) To what extent do citizens or client groups participate in the decision-making process?
  - c) What is the role of the board or committee?
    - d) Is there any training program for volunteers?
- 6. Organization

7.

- a) What is the organizational structure of the project?
- b) What are the roles of the staff?
- c) What is the working relationship between professional staff and lay staff, consumers, board etc.?
- Accomplishments and consequences
  - a) Is there any evaluation done? .
    - b) Do you think your project is achieving its stated objectives?
    - c) What are the other achievements of the project?
    - d) What kinds of problems did/do you encounter?

### APPENDIX B

A DESCRIPTION OF FOURTEEN

PREVENTIVE SOCIAL SERVICES

PROJECTS IN EDMONTON

. 0

This appendix includes fourteen Preventive Social Services

projects studied. They are as follows:

### A. Family Service:

1)	Edmonton Family Planning Service (E.F.P.S.) Emergency Homemaker's Service (E.H.S.)
-/	(Family Service Association of Edmonton)
3)	Family Aide (F.A.)
4)	(Edmonton Social Service Department) Urban Life Skills and Child Care Program (U.L.S. & CC.)
100	(Canative Housing Corporation)

### B. Services to Youth

5) St. Francis Youth Club (S.F.Y.C.)
(Boys Club of Edmonton)
6) Teen Time (T.T.)

### C. Services to Children

D.

7) Bissell Child Care Centre (B.C.C.C.)

# Services to Senior Citizens

- 8) Operation Friendship (O.F.)
- 9) Strathcona Place Society (S.P.S.)

# E. Neighbourhood Services

10) Dickensfield We Give Centre (D.W.C.C.)

11) Open Door (0.D.)

### F. Support Services

13) Community Social Services (C.S.S.). (Edmonton Social Service Department)
14) Meals-on-Wheels (M.O.W.) (Victorian Order of Nurses)

 <u>Edmonton Family Planning Service</u> (E.F.P.S.)<sup>1</sup>

a. <u>Historical background</u>. This organization was at one time known as Planned Parenthood and was pretty inactive. A community worker who had organized similar groups in the Northwest Territories reactivated. In the spring of 1973 seven women and a medical doctor met together, drew up a constitution and incorporated the Edmonton Family Planning Services under the Societies Act. It is a member agency of the Family Planning Federation of Canada, along with similar groups from nearly every province in Canada. Initially, the Edmonton Family Planning Service was funded by the Family Planning Federation of Canada.

1.58

b. <u>Sponsorship</u>. At present the Edmonton Family Planning Service is funded by the Family Planning Federation of Canada, Preventive Social Services and Federal Health and Welfare that has provided funds for an Education Director.

c. <u>Objectives</u>. The basic aim of Edmonton Family Planning Service is defined: "To promote and foster enlightened family planning." thereby contributing to the physical, emotional and social well-being of individuals, and to the health and social development of the community.

d. <u>Clientele</u>. The Edit Construction of Planning Service serves the total Edmonton community. e. <u>Methods of Work</u>. The first strategy is to provide information, counselling and referral services to individuals on such things as pregnancy, birth control, venereal disease, abortion and related topics.

159

The second method is to organize educational programs, seminars; presentations etc. for groups and agencies in the city who request them. The Service also distributes information and literature aimed

at increasing knowledge in this area.

Volunteers are also involved in various kinds of activities such as clerical work, counselling, and education outreach program.

f. <u>Organization and administration</u>. The Board of Directors is elected by members and their job is to conduct the affairs of the society and to determine policy. The Advisory Board is appointed by the Board of

#### Figure 25

The Organizational Structure of Edmonton Family Planning Service



Directors and its job is to act in an advisory capacity to the Societ. The duties of the Executive Director include administration, exploring for funding and public relations function. The Education Director is responsible for all the educational aspects of the program and to involve more volunteers.

160°

2. Emergency Homemaker's Service (E.H.S.)<sup>2</sup>

a. <u>Historical background</u>. The Emergency Homemaker's Service is administered by the Family Service Association of Edmonton which was established in 1942 for the purpose of strengthening family life in the community through programs of prevention and treatment. The service was begun by Junior League after a community study was done. At that time, there were many Homemaker's service operating in other parts of North America. Junior League put money ávailable and asked Family Service Association to staff and administer the program. In 1950, the Homemaker's service was run totally by the Family Service Association. It received funding from Preventive Social Services in 1967.

b. <u>Sponsorship</u>. The Emergency Homemaker's Service program is sponsored by United Way, Preventive Social Services, service fees, and other small grants.

c. <u>Objectores</u>. The primary objective of the program is to assist families over crises occasioned by death, desertion or illness of the mother or the person who ordinarily takes care of the members. d. <u>Clientele</u>. The program is open to families and individuals who need the service in the City of Edmonton.

e. <u>Methods of work</u>. The basic procedure is to place a homemaker, who<sup>®</sup> is employed by the Family Service Association, to the family that needs the service.

Consultation and counselling is an integral part of the service to assess and determine whether the children's needs can best be met by homemaker service, some other service, or through the familiy's own resources.

The agency also provides pre-employment orientation and continuing in-service training to increase homemaker's skills, knowledge and understanding of the needs of children and parents.

There is no volunteer involvement in the delivery of this service. For evaluation there are periodic client questioning by telephone or through mailed questionnaires and homemakers evaluate continually.

f. Organization and administration. Since the Emergency Homemaker's Service is administered by the Family Service Association.. the overall policy is set by the Board of Directors for that agency. For the program itself, there is a community Homemaker Advisory Committee which is composed of a wide representation of community agencies and service programs who refer families for homemaker service. The committee meets regularly to review the nature and work of the homemaker program and is encouraged to put forward recommendations.



Figure 26

The senior counsellor in charge of Homemaker Service is the supervisor of the counsellors and homemakers, who is generally responsible for helping the homemakers and counsellors to develop the attitude and skills they need for the job.

The counsellor is also directly responsible for some problem solving work with the family and/or family referral to the appropriate community resource.

The duties of the homemaker varies in each family and situation and situations within families may change rapidly. Her duties usually

162

include taking care of the children, helping the mother, or housekeeping. The homemaker learns her specific responsibilities in each assignment from the counsellor and has access to ongoing guidance and instruction from the counsellor in carrying out her responsibilities. At times, she is working on a team with a physician, nurse, social worker, etc.

3. Family Aide Service (F.A.)<sup>3</sup>

. a. Mistorical background. The Family Aide Service was initiated by Social Service Department workers in 1965 as part of the municipal social service program. At that time, half of the cost of the service was reimbursed through Federal grants. In 1966, this was the first departmental service funded by Preventive Social Services.

b. <u>Sponsorship</u>. The Family Aide Service is primarily funded by Preventive Social Services.

c. <u>Objectives</u>. The basic aim of the service is to improve the functioning of the family and prevent family and individual breakdown.

d. <u>Clientele</u>. Any family in the City of Edmonton is eligible for service. Most of the time Family Aides have worked with young immature mothers, families of low income, single parent families, overloaded families with a large number of children, etc. e. <u>Methods of work</u>: The service does not aim at replacing mothers who are temporarily out of the home, but at allocating family aides to a family where serious problems are evident. Its procedures are based upon its philosophy that all families want to have a better life, and are concerned about the health, education and happiness of their members. However, they often need to be "shown the way". A major goal of the Family Aide is to establish rapport through "doing together" and gradually building up an atmosphere of caring, and mutual trust in which learning can take place.

164

Referrals for service are made by the social worker from the Social Service Department, other social agencies, hospitals, etc. to the Family Aide Supervisor. The Family Aide sent, who is a paid staff member with skills in home management and housekeeping, is briefed on the family situation by the supervisor and social workers before she works with the family. The length of time she spends in a home .depends on need and speed of improvement of the situation.

When the service is terminated, a written evaluation of the service received is requested from the client.

There is no utilization of volunteers in the delivery of service.

f. Organization and administration. Since the Family Aide Service is directly operated by Edmonton Social Services the organizational chart of the project will not be shown in this section (for reference, refer to figure 2 in Chapter IV).

In this project, there is a Family Aide Supervisor who is

responsible for the supervision of Family Aides, coordination of the services, reviewing the background or family situation of the referrals,

165

The duties of the Family Aide varies in each family depending upon the needs and the family situation. Her duties may include helping the family with budgeting, household management and health improvement; providing assistance with child care for part of the day, encouraging the family to participate in community activities and so on. She works with social workers and other professionals in a team approach to families.

### Urban Life Skills and Child Care Program (U.L.S. & C.C.P.)<sup>4</sup>

etc.

1974.

a. <u>Historical background</u>. The Urban Life Skills and Child Gare Program is conducted by the Canative Housing Corporation which is a nonprofit organization established by Albertans to provide housing for disadvantaged people like the natives in the major urban centres of the Province. The program was started because of the conviction that success in improving the living standards of such people would not be met by merely providing a better quality home, but rather by assistance in adapting to the entirely new urban environment. The corporation, therefore, decided to develop an Urban Skills Program to provide opportunities for improvement in family life success has tarted with a LIP grant and then funded by Preventive Social Services hast fall, in
b. <u>Sponsorship</u>. Preventive Social Services provides the operation costs of this sub-program.

c. <u>Objectives</u>. The basic objectives of the program is "to provide a centre where mothers possessing minimal experience in contemporary life skills may, in a confidence-inspiring and home-like setting, learn to cope with the demands of rapidly changing social and economic

urban conditions."

d. <u>Clientele</u>. The major target group is comprised of tenant mothers, most of whom are native. However, any other interested persons are welcomed to register in the course.

e. <u>Methods of work</u>. The rural-urban emigrant problems, which the program tries to alleviate, result from cultural and language barriers, poor education, and ignorance of urban life skills. People within the disadvantaged group are therefore unable and unwilling to get involved in existing programs or facilities mainly oriented toward urban whites. In this urban-skills program, native personnel are hired to conduct courses in home management, health and diet, use 'of local services and agencies, explanations of urban attitudes and how to cope with social pressures. It is hoped that through group, class and private discussion, many of the potential problems encountered by the target group will be overcome or circumvented.

In order to encourage mothers to attend the program, a child care service was developed with the assistance of West-Ten, as a regular part of the program. It has proved itself to be indispensable. In addition, transportation to and from the centre for students is provided. As yet, there has been no recruitment or training program for

167

volunteers.

f. Organization and administration. The Program is operated under the supervision of Canative's General Manager, Board of Directors, and Advisory Board. The Board is made up of native people, and is responsible for policy-making and control of finances. The General

#### Figure 27

#### The Organizational Structure of Urban Life Skills and Child Care Program



Manager is responsible to the Board for all the programs conducted by the Canative Housing Corporation.

A social development supervisor is hired to control the operation and staff of the Urban Skills and Child Care program, plan course content, keep in liaison with other agencies, and contact prospective students.

The Instructor/Group Leaders would be responsible for the presentation and implementation of the courses.

## 5. St. Francis Youth Club (S.F.Y.C.)

a. <u>Historical background</u>. The Boys' Club of Edmonton was opened in 1962 in a house in the Boyle Street area. A new building was opened in 1966 after a successful \$300,000 fund-raising project. The St. Francis Youth Club which is an offshoot of the Boys' Club started operations in January, 1972 in St. Anthony's College with the support of federal grants. In May, 1973 it received a \$30,000 non-fenewable 'grant from the Department of Culture, Youth and Recreation, and moved its operations to a trailer in North Edmonton Park. In December, 1973, it moved into a two-bedroom house in that area. But the house was vacated in October, 1974, To date the St. Francis Club is operating with community facilities, using gyms of the North Edmonton Public.School

and the St. Francis Separate School for their activities. The St. Francis Youth Club first received Preventive Social Services funding in 1974. b. <u>Sponsorship</u>. It is sponsored by United Way, Preventive Social Services, and other small grants.

c. <u>Objectives</u>. The objectives of the St. Francis Youth Club are more or less the same as that of other boys' clubs. Basically, the Club tries to meet the needs of its members, which may be social or recreational.

d. <u>Clientele</u>. The program is open to any youth in the City. But over 90 per cent of its members come from within a ten block radius of the activity-centre, i.e. the North-east Edmonton area.

e. <u>Methods of work</u>. For recreational needs, the Club provides outdoor and camping services on a year-round basis. Moreover, there are physical education activities, social recreation, arts and crafts programs etc. Work in the area to meet the unmet needs of those youths experiencing difficulty with personal, social and emotional growth includes

- Group Clubs

years old.

- Informal, individual guidance

- Liaison with other youth services and community groups

The Club has emphasized the small group approach, i.e. members who have interests in common will form a group to do their own thing. The small groups are usually organized with around six persons at three levels: the Junior with youth from seven to eleven; the intermediate with youth from eleven to twelve; and senior from twelve to seventeen

The last but not least important method is the recruitment and training of volunteers. In 1974 the volunteers came from the community at large, teen membership and from student personnel specially recommended for field placements.

f. Organization and administration. The overall policy is made by the Board of Directors of the Boy's Club of Edmonton. The Board

Figure 28

The Organizational Structure of the St. Francis Youth Club



members are concerned citizens, nominated by the Board itself. For the St. Francis Youth Club, there is a Director responsible for the management of the total operation of the project and the training of volunteers. The Program Director shares the work load with the Director, but is mainly responsible for the co-ordination and development of programs.

171

# 6. Tex Time (C.T.)<sup>6</sup>

when a group of teenagers met at the present executive director's house for a Valentine Party. Following that party, informal social gather file took place regularly and within three years those small gatherings had organized themselves into a formal club offering regular sport programs. Further growth occurred in 1967 when a week-long camp was a success and camping became a regular part of their program. At the same time, the present Executive Director began to develop his ranch site which has provided Teen Time with a place for many recreat final activities.

The Club continued to grow until 1969 it was incorporated as. a Society and registered as a charitable organization. In 1972, an office centre was obtained. The Club kept growing over the years and eventually purchased 400 acres of land for a teen ranch and a yearround lodge. It received funding from Preventive Social Services in April 1974.

b. <u>Sponsorship</u>. Besides a membership fee the Club is supported by fund-raising projects, donations, Preventive Social Services, and small grants from time to time from other civic and provincial departments, like the Department of Culture, Youth and Recreation.

c. <u>Objectives</u>. The objectives as stated in the Report of Teen

to aid teens with problems by helping them to understand the problems and themselves better.
to provide opportunity for development of a balanced life, socially, spiritually, physically and mentally and mentally.

d. <u>Clientele</u>. The program is open to any teenager in the City" Edmonton.

e. <u>Methods of work</u>. Teen Time's methods of work are based on the conviction that negative forms of youth behavior are in large part due to boredom, frustration, fack of adventure, and a failure to fulfil one's potentia. Its programs are also based upon its philosophy that if any one of the four areas--social, recreational, mental and spiritualare neglected one cannot have a full and balanced life. With this

philosophy Teen Time offers the following programs:

- Weekly Jocial evenings

- <u>Gym nights</u> at Jasper Place Composite High School With loosely organized sports activities.

- <u>Chapel</u> - Sunday afternoon non-demoninational discussion period

- Music and drama

- Summer and Winter Camping

- Leadership and Counsellor-Training

This is a training program for those teens actively involved in Teen Time, who have the desire to become non-professional counsellors within the club.

Evaluation means are informal. In 1974, evaluation forms were given to campers to find out whether they felt accepted and had learned something. But the director has stressed the difficulty in evaluating human development.

f. <u>Organization and administration</u>. The Board of Directors which is the decision-making body is responsible for making policy, defining objectives, negotiating for funding etc. The Board members are elected bi-annually by governing membership (members 18 years old and over who have paid \$10 club fees). At present, the Board consists

#### Figure 29

#### The Organizational Structure of Teen Time



of physical education teachers, businessmen, accountants, housewives,

All staff members are volunteers with the exception of three-the Secretary, the Program Co-ordinator (part-time) and the Camp Program Co-ordinator (part-time). The roles of the Executive Director are those of promotion, public relations and financing. The City Program Co-ordinator is remonsible for the overall direction of the Teen Times programs. The Camp Program Co-ordinator is responsible for planning, promoting and directing weekend retreats or camps for both junior and senior teenagers.

The senior teenager groups have elected an executive and area representatives, all of whom work directly with the Program Co-ordinator in selecting and planning activities. The Co-ordinator is only involved in the planning stage, then the older seniors take over and run the activities themselves. The junior group on the other hand is helped by older teens. The juniors' views are brought forth by area representatives, who with the assistance of the older teens plan and run their own activities.

7. Bissell Child Care Centre (B.C.C.C.)<sup>7</sup>

etc.

a. <u>Historical background</u>. The Bissell Child Care Service began operation in February, 1974 under the auspices of the United Church of Canada's Bissell Centre, and funded by the Local Initiatives Program. This service was initiated by the Board members of the Bissell Centre who felt that there was a great need for such a service in the area. Funds have been obtained from Preventive Social Services since January,

b. <u>Sponsorship</u>. The Centre is funded by Preventive Social Services, and the space, utilities, and office supplies are provided by the Bissell Centre.

c. <u>Objectives</u>. According to the Director of the centre, the major objective is to provide community based free drop-in child care services for families, particularly single parent families, and people who really need such services, but it is not a regular day-care centre.

d. <u>Clientele</u>. The service is open to any family with children from infancy to 5 years old. But parents served usually include:

- People who cannot pay for or canning find a reliable part-time baby-sitter.
- families referred by the Boyle Street Co-op, the Public Health Nurse, the Royal Alexandra Social Service, the City Social Service Department or etc.
- Social assistance recipients referred by Provincial Government Social workers.
- Parents who want to visit doctors or psychiatrists, shop, attend classes at colleges or vocational centres, etc.

e. <u>Methods of work</u>. The major method of work is to provide child care service mainly by paid staff from 8:30 a.m. to 4:30 p.m., Monday through Friday. A special program for 3, 4, and 5 year old children has been developed and involves field trips, films, special handi-crafts and outdoor activity.

In 1974, a communication workshop in 'Child Care' was

offered to interested parents. The centre would like to organize more workshops, but funds are inadequate.

The centre is also attempting to involve parents and start a volunteer program with a volunteer co-ordinator. Recently, five students and a mother have participated as volunteers.

f. Organization and administration. The Bissell Child Care Service is operated under the auspices of the Bissell Centre which is governed by a board of directors. The Child Care Service, however,



has a separate Bissell Child Care Service Committee who is responsible for getting funds for the services, hiring staff, and programming. At present, the Committee is composed of staff persons from Park and Recreation Department, City Social Services, the Director of Bissell Centre, the Director of Bissell Child Care Service and a few interested parents. The Child Care Service Director is responsible for the administration of the Service and the organization of various activities.

### 8. Operation Friendship (O.F.)

a. <u>Historical background</u>. Operation Friendship began with the work of some volunteers at the Biasell Centre, a United Church Agency in the city centre area. These volunteers found many isolated people with a multiplicity of health and social problems, and began to do some door-to-door out-reach visiting and to help individual elderly with specific problems. The program received impetus when the original volunteer organizer was provided with a small salary on a part-time basis by the Society for the Retired and Semi-Retired. Further help was provided by volunteers from the Alberta Service Corps in the summer of 1972.

Through the efforts of the Community Development worker in the City Centre unit, two staff people with Edmonton Social Services, funded by a LIP grant, started working with Operation Friendship in December 1972, particularly in the McCauley area. With their help and that of many other volunteers, a second drop-in centre was opened. As the funding for these two staff people was terminated in December 1973, Operation Friendship has received to hims unlist the provide from the Department of Culture, Youth and Recreation. Since December 1974;

Operation Friendship has received funding from Preventive Social Services.

b. <u>Sponsorship</u>. Operation endship is funded by New Horizons Program, Preventive Social Service and the United Way. Bissell Centre has contributed the space for two drop-ins--at Boyle Street and McCauley. The Boyle Street Openity Service Co-op has contributed office space for the Co-ordinator and some transportation services. Many churches, community organizations and private individuals have contributed food and money for food costs at the drop-ins.

178

#### c. Objectives.

To locate and bring information about services for senior citizens to older persons living in this area.

The never desired by the older persons, to visit on a regular basis, providing friendship and practical help with everyday problems.

To be aware of the variety of services and resources in the community and if needed, help the older person reach a needed service.

To encourage the older person to participate in programs available and to use their resources and abilities to help themselves and the community.

To educate the community as to the needs of the older persons in this area and to stimulate action.

To speak out on social concerns which affect the elderly in the area.

To provide opportunities for fellowship in group activities.

d. <u>Clientele</u>. Most of the clientele are elderly persons in

the McCauley area.

Ľ1

e. <u>Methods of work</u>. The methods of work are based upon the needs and problems that exist in the area. Some of these needs are listed in 'the Boyle Street Study', others are perceived by the workers. The crucial needs and problems of the older people in the Inner City area are identified to be: loneliness, depression and isolation, poor housing, health problems, low income, lack of social and recreational facilities in the area and language problems.

179

The major program of Operation Friendship is outreach home visiting to the elderly. This door-to-door work has a socializing, supportive, and informative function. The elderly are also provided with opportunities for socializing and recreation at the drop-ins. As a result, the Operation Friendship Action group was formed. This action group is made up of senior citizens who discuss and help with plans for programs as well as make the needs and ideas of the elderly known to appropriate levels of Government and community

groups.

With regard to housing needs, Operation Friendship staff working with other interested agencies and meeting with various city departments have helped their elderly people become involved and have a voice in various housing plans. With the help of public health nurses, Operation Friendship has completed a housing survey of the elderly in the area

in 1972.

In connection with the problem of health, paid workers and volunteers continually arrange for health care services through Public Health nurses and doctors etc. The drop-in centres also help to supplement the older person's diet, with their daily lunches and Friday hot meals.

180

With regard to the 'low-income' or 'lack of income' problem, particularly for those under 65 years of age, Operation Friendship frequently acts in an advocacy role for these people, arranging transportation, helping them to obtain assistance that is due them etc. Since Operation Friendship works very closely with other services in the area, there, are many consultation and referrals of service back and

forth.

f: Organization and administration. Operation Friendship has become independent of the Society for the Retired and Semi-Retired, and

#### Figure 31

The Organizational Structure of Operation Friendship



the Boyle Street Community Service Co-op, with their own Board of Directors and staff incorporated as a separate organization. At present, the Board is the decision-making body made up of professionals and two senior citizens in the area. The Board members are not elected, but are invited by previous members or other workers of Operation Friendship.

The role of the Co-ordinator is to recruit volunteers and coordinate the work of the volunteers and the workers. She makes contacts with people who are referred by hospitals and agencies, assesses the situation and arranges for appropriate folow-ups. Part of the job of the Co-ordinator is to keep in touch with all of the agencies Operation Friendship works with.

The two non-professional staff members are responsible for running the drop-ins, helping with human relations problems, visiting etc. There are also volunteers from outside the area and in the area, who have helped with all aspects of the work.

9. <u>Strathcona Place Society (S.P.S.)</u>9

a. <u>Historical background</u>. The project was initiated by a group of concerned people in the community, who were concerned, originally, with housing for senior citizens. Most people involved in this group were from Churches. Around 1969, they did a survey of people over 60 years old in the Garneau and Strathcona neighbourhoods asking about housing and other needs. Since recreation needs were expressed by most such persons in the community, the group developed the idea of having " covered recreation facilities. The City came forward with an old church building and asked the group if they would explore the possibility of funding for capital expenditure. The group and the community, therefore, formed the Society and later received a budget from Preventive Social Services.

b. <u>Sponsorship</u>. The basic operating funds are provided by Preventive Social Services.

C. Objectives. The basic goal of the Society is "to help the elderly to lead rewarding and happy lives, and to assist them to maintain their independence and dignity." The Society is concerned with the total needs of the elderly.

d. <u>Clientele</u>. The Society basically serves people over 60 years of age, and most of the members come from South Edmonton. But the Society also has an arrangement that allows people who are under 60 to come with their older friends under certain conditions.

e. <u>Methods of work</u>. Many of the activities at the Strathcona Place Society are centered on recreation, but the Board, and many members believe that through recreation, many other unmet needs may be identified and later acted upon in terms of new programs, referrals etc. The Society has provided in-centre, out-of-centre, and out-reach programs and services on a daily, weekly, monthly or annual basis.

For the in-center and out-of-centre programs, the Society has provided a regular food service (lunch, and afternoon coffee), and has made health, legal and family counselling readily available. But most of the social and recreational activities such as bus tours, dances, evening entertainments etc. are organized on an ad hoc basis by various advisory sub-committees.which are made up of elderly members. During 1974, a number of special classes and discussions were held, e.g. on Defensive Driving; discussions on Income Tax, deafness and blood

For outreach programs, the Society has sponsored the Strathcona Outreach Proposal (a New Horizons Project) which was presented by fifteen senior Strathcona Place members. At present, Strathcona Place members provide 90 per cent of the 170 volunteers required to carry the outreach program to home or hospital bound elderly citizens.

pressure.

In sum, the Society has provided seniors with opportunities of planning for, and participation in crafts programs, continuing education, recreation and social events.

f. Organization and administration. The Board of Directors is elected by members in the annual meeting. At present, half of the Board members are non-senior citizens, but influential and knowledgeable persons in the community. The Board is responsible for policy decisions, management of finance etc. There is also an Advisory Committee consisting mostly of ordinary members, appointed through the nominating committee of the Board. The Advisory Committee has over fifteen sub-committees who plan and recommend centre activities.

The Executive Director has the general responsibility for all phases of operation, and is accountable to the Board of Directors. His duties include personnel administration, public relations, supervision



The Organizational Structure of Strathcona Place Society



of budget, scheduling of programs etc. He is also the official representative for the Society in negotiations with City departments. The staff perceive their roles as consultants and resource people to the groups.

185

10. Dickensfield We Care Centre (D.W.C.C.)<sup>10</sup>

a. <u>Historical background</u>. In January 1973 a group of tenants from Dickensfield Public Housing identified the need of a community drop-in centre. Some students from Grant MacEwan College were invited to do a survey in the area. The survey indicated that such a centre would be well received. As a result an agreement was reached in October, 1973 with the Edmonton Housing Authority, making a free three bedroom unit available. In April, 1974 the Dickensfield We Care Centre was incorporated under the Societies Act.

From the start, management of the Centre was in the hands of a small group of dedicated residents (forming a Board of Directors) with, the part-time help of a community worker from the Glengarry Office of the City Social Services and some financial assistance from the same office. On July 1, 1974 a grant from Preventive Social Services enable the Centre to hire a full-time Co-ordinator.

b. <u>Sponsorship</u>. This Centre is funded primarily by Preventive Social Services.

c. <u>Objectives</u>. The objectives of the Centre as outlined in the Annual Report are:

- To provide social, educational and recreational opportunities for residents of the area To improve the quality of life in the area of high

in the image of the area by showing initiative the image of the area by showing initiative the second secon

d. <u>Clientele</u>. The Centre is open to all residents of the Dickensfield neighbourhood (not restricted to public housing tenants).

e. <u>Methods of work</u>. To accomplish the objectives, the Centre has organized the following educational, social, recreational activities for the area:

- Mother's Day Out Program

Mother's Group - a 'getting together' to break the monotony of the week. The program is centered around the interests of the mothers.

Children's Arts and Crafts Program

Teen Program

- Legal Aid Services for every Monday evening.

In addition, there were workshops on communication skills, income tax information etc. The centre also sends out newsletters to every household in the neighbourhood. Volunteers are recruited to organize various programs.

Evaluation of the methods used is informal, mainly through verbal feedback.

f. <u>Organization and administration</u>. The Board is the decisionmaking body made up of residents in the neighbourhood except one who is also a volunteer director for one of the programs. The Board members are elected by membership of the Centre. Membership is open to any resident in the area. The Co-ordinator is the only paid member of staff. She is responsible for administration and co-ordination. All programs are organized by volunteers.







## 11. Open Door (0.D.)<sup>11</sup>

a. <u>Historical background</u>. Pioneer work on this information and referral centre was done in Area 13 by a O.F.Y. Team during the summer of 1971. A Committee responsible for acting on the findings of the O.F.Y. team submitted a brief to the City's Social Services Department concerning the need for such a centre. A Preventive Social Service grant was finally received in April, 1972 to cover office supplies - rent, and a part-time secretary. It was operated by the Area 13 co-ordinating council as an information and referral centre in the southeast area of the City. At present, Open Door has its own Board and works in cooperation with the Area 13 co-ordinating council.

b. <u>Sponsorship</u>. It is funded primarily by Preventive Social Services.

c. <u>Objectives</u>. The stated purpose of Open Door as set down by the Board at a Process Planning meeting, 1974, is to be an information exchange for the Area 13 (Capilano, Forest Heights, Fulton Place, Gold Bar, Holyrood and Terrace Heights community leagues) and to provide the community with a hospitable, readily identifiable contact point, so that residents or service oriented resource persons, or groups, may more easily get/and/or give the help needed, if possible from within the community.

d. <u>Clientele</u>. Open Door basically serves residents in Area 13 and the six community leagues therein. Sometimes, there are phone calls and requests from outside the community.

e. <u>Methods of work.</u> As stated in its purpose, a primary activity is to provide information and referral service to the people in Area 13. In order to gain public recognition and understanding of Open Door, a letter was sent to all organizations and professional people working in the community and some agencies in and beyond the boundaries of Area 13

explaining their purpose and facilities and inviting participation. A similar pamphlet was also distributed to all homes in the Area. Open Door has set up a program to recruit, train and maintain volunteers as Information Aids, Block parents/Good Neighbours, and Emergency volunteers. At present, Open Door is acting as the co-ordinator for Area 13 in cooperation with the Police Department and schools in the community in the development of a Block Parents program.

Open Doer has also taken the responsibility of disseminating community news through Area 13 News which reaches all homes in the area free of charge.

Since 1974, Open Door has organized Annual process planning sessions to evaluate the work done and to set up it's future course of action.

f. Organization and administration. Since Open Door has taken the responsibility of Area 13 Community News, the Open Door Board has amalgamated with the once separate Area 13 News Board. The Board is made up of residents in the area and is responsible for policy, decisionmaking, exploring sources of funding etc. The role of the Volunteers-Co-ordinator is to recruit and train volunteers, maintain contact with other groups and agencies such as A.I.D., Social Services Department etc., co-ordinate programs and serve as a telephone listener.

Volunteers are recruited and trained to staff the Open Door/ Area 13 News service. Most of them act as telephone listening aides, assist in the work of getting out the newspaper and in contacting community leagues, churches, schools and agencies for suitable material.



#### Advice, Information and Direction (A.I.D.)<sup>11</sup>

11.

a. <u>Historical background</u>. The agency was established in 1960 by the Edmonton Welfare Council (Edmonton Social Planning Council) in response to a Citizens Committee's request for some sort of 24 hour clearing house for the Social Service Agencies in the City. This committee, which included concerned citizens and government officials from the Provincial and Municipal levels, invited the Edmonton Social Welfare Council to look into the possibility of establishing a central information and referral centre. The Edmonton Welfare Council then set up a subcommittee to look into the problem and they acquired funding to set up the agency. Until 1972, A.I.D. service was part of the Edmonton Social Planning Council. Original funding was by cost-sharing

Figure 34

arrangements between the Brovince, the City and the United Way.

b. <u>Sponsorship</u>. A.I.D. is jointly funded by Preventive Social Services and the United Way.

c. <u>Objectives</u>. "The objective of the agency is to provide for the development of fully functioning people by linking needs and resources."

191

The objective has not been changed since its inception. In addition, it was decided in 1972 to provide a distress line which is a 24-hour listening service for people facing crisis situations. Prior to 1972, A.I.D., was mainly an information centre.

d. <u>Clientele</u>., The agency serves the entire Edmonton community.

e. <u>Methods of work</u>. The basic method of work of the A.I.D. service is to develop and co-ordinate an information and referral centre in the City of Edmonton. Its activities include the following:

- acquiring, compiling and up-dating comprehensive information on community resources;
- disseminating the information, in response to requests, by an active outreach program and via telephone, personal interviews, correspondence, special publications, speaking engagements, information seminars, and such other methods as are deemed appropriate;
- keeping comprehensive statistics on contacts with clients
  - i. °accurately describe the service offered,
  - ii. provide data for improving staff patterns and revising training techniques
  - iii. record information on problems, apparent service malfunctions or gaps.
- stimulating the development of a more effective communications network by acting as a resource to neighbourhood and specialized information services.

A.I.D. also operates a distress line to help the caller through



o

192.

initial crisis, and to assist the person in distress in finding the right community resource to provide him with either immediate or long-

term support.

In addition, A.I.D. recruits, screens, and trains volunteers as telephone listeners in information dissemination and crisis intervention.

In part, the methods of work have been changed in the last two or three years. Prior to the establishment of the Distress Line Program, the direct service provided by the agency was handled by paidstaff, and volunteers were used casually in a clerical capacity only. With the introduction of the Distress Line Program and the extension of the information/referral service hours of operation, the volunteer training program has been expanded. Now volunteers are being utilized as telephone listeners, resource leaders, trainers etc.

Traditionally, A.I.D. Service provided information only upon request. Now, the agency explores less passive methods of disseminating information to the community. Techniques used include presenting seminars, publishing directories like day care directories, youth service directories etc.; and so on.

There'is no formal on-going evaluation of the methods simply because of the nature of the service. The director stressed that follow-up studies of the clients are difficult because the clients are often anonymous.

other agencies, a few citizens and two volunteers. The role of the Board is to ensure the continuity of the agency, to set the policy, to seek adequate funding and to look at program development and evaluation. The Board has established a committee structure in which each committee is made up of three Board members and some other citizens so that local expertise will also be utilized.

There are five paid staff members: the Executive Director, Volunteer Program Director, two Information Officers, Secretary/ Bookkeeper, and Statistician. This staff is responsible for the implementation of policy, co-ordination of the service, and the provision of support to the volunteer staff. Most volunteers act as telephone listeners, but some others may be utilized as resource leaders and trainers.

### 13. Community Social Services (C.S.S.)<sup>13</sup>

a. <u>Historical background</u>. With the advent of Preventive Social Services in 1966, the Provincial Department of Health and Social Development was taking over responsibility for essentially maintenance and rehabilitation functions, thus freeing the municipalities to go into the areas of Preventive Social Services. The Community Social Services is part of the City's Preventive Social Services Program. Around 1970,. there was a general trend to decentralization of services, so the City Social Services Department has gradually developed a number of local community units in the City. Basically, Community Social Services are provided in seven local units or geographic divisions of the city. Five areas are served from area offices (Jasper Place, Glengarry, Duggan, West 10 and Beverly) and two from the central office (downtown and west central). Unit services include community preventive counselling, group programs and community organizations.

b. <u>Sponsorship</u>. The <u>Community</u> Counsellors and Community workers for the seven areas are funded primarily by Preventive Social Services. However, the budget for each local unit comes in 'a package for both preventive social services and statutory services.

c. <u>Objectives</u>. The objectives of the Community Social Services are not clearly defined. In general, the services are designed to develop community awareness and resources, to strengthen and preserve human initiative and to preclude individual or family breakdown.

d. <u>Clientele</u>. Each local unit basically serves the community in its own geographic area.

e. <u>Methods of work</u>. The City has been arbitrarily divided into 'seven units with a community counsellor and a community worker assigned to each area. At present no strict policy guidelines have been established for the community social service program, but in general the workers in the program have been trying to get to know each community and to help people maximize their ability to cope with their environment.

Most community counsellors in the local units have provided . marriage counselling service, counselling for families or individuals

195

-

on social assistance whose counselling needs are not really related to wheir financialoneed, etc. The counsellor's role is to help the client identify his need and cope with his own problem. The preventive counselling service is basically a direct service. In some areas, however, counsellors have provided back-up services for other workers as well.

A wide range of group programs covering youth, family, special interest, and senior citizen are also provided through unit offices.

For community development service, community workers are the resource persons--consultants, trainers, analysts, and enablers for many community groups in their areas. They also assist in the initiation of new local projects. According to the supervisors and community workers in the local units, the kind of approach taken in each area has largely been dictated by the particular situation and by the personality of the worker.

Up to now, the community workers of the seven local units have been involved in developing such programs as Teen Club, Mother's day out, Parent Effectiveness, etc. They also provide resources for the large number of varied groups of people within the neighbourhood including action groups, neighbourhood improvement groups, senior citizen groups concerning housing, etc. In such there are three types of involvement with community groups. The first type involves initiation only with little or no carry over. The second type involves regular contact. The third type requires periodic contact with individuals or groups which are functioning on their own. These kinds of involvement are largely dependent upon the situations in each area. Basically, the role of the community worker is to encourage community groups in the area to look into their own situations, to define their goals and to use their own resources to meet these goals. The needs of the community groups may be day care facilities, development of parks in the area, better transportation services etc.

With regard to the use of volunteers, the Duggan unit has an extensive training program.

f. <u>Organization and administration</u>. Since Community Social Services are provided directly by the City Social Service Department, the organizational structure will not be shown in this section.

### 14. Meals-on-Wheels (M.O.W.)<sup>14</sup>

a. <u>Historical background</u>. Around 1969, there were requests for the service from the community, then various community organizations grouped together to explore the need for such a service. They came to the Victorian Order of Nurses and asked if this agency would like to administer the service. They agreed. A pilot project was then started in the South West section of the City, with funds from the Edmonton community; an average of six recipients received meals. Since that time the service has grown to include the South East and Northwestern section of the City and the North Side with the Good Samaritan Nursing Home, the Hardisty Nursing Home, the Royal Alexandra Hospital, and the Grey Nuns Regional Centre providing meals for an average of 180 recipients.

It has been funded by Preventive Social Services since 1970, one year after its operation has begun, but is still administered by the Victorian Order of Nurses.

198

b. <u>Sponsorship</u>. The meals-on-wheels project is funded by Preventive Social Services, service fees, and private donations.

c. <u>Objectives</u>. Basically, the objective of Meals-on-Wheels is to provide well-balanced meals to homebound individuals who are unable to prepare meals themselves.

d. <u>Clientele</u>. Individuals will be eligible for service providing the following conditions are met. Persons receiving meals must:

- 1) live in the area served (in the City of Edmonton)
- 2) have no one in the household able to prepare meals
- 3) be unable to prepare adequate meals
- 4) be homebound but able to feed himself
- 5) have available resources for the provision of meals on wheels on weekends and holidays
  - 6) be unable to afford to have meals prepared or delivered from private sources.

e. <u>Methods of work</u>. The basic modus operandi is to deliver the packaged meals to the clients. Volunteers are recruited to be drivers and servers, delivering hot meals five days a week. They are expected as well to be sensitive to the needs of the recipients.

Requests for service may come from any source such as community services, social agencies or the general public. Persons requesting services will be interviewed to determine eligibility based on the conditions necessary for the provision of the service. If an individual is not eligible for this kind of service, an effort will be made to arrange for more appropriate services.

199

There is no formal evaluation of the service, but a questionnaire is sent out from time to time.

f. Organization and administration. The Board of the Victorian Order of Nurses is the overall policy decision-making body which is

#### Figure 36

The Organizational Structure of Meals-on-Wheels



composed of professionals, retired persons, and interested citizens. The Board members are elected annually in a public meeting. There is a Director who is responsible for the supervision of all programs under Victorian Order of Nurses. For the Meals-on-Wheels project, there is a Co-ordinator who is responsible for its general operation. She also assesses and keeps records of people who request the service.

There is also an Advisory Committee for Meals-on-Wheels, which is composed of three members from the Board, a working volunteer, interested citizens, and some founders of the project. The Advisory Committee is appointed by the Board of Management. They meet four or five times a year and make recommendations about current services to the Board of Management.

This project is almost entirely dependent on volunteers. The Coordinator is the only person paid.

#### Footnotes

<sup>1</sup>From an interview with S. Greenhill, the Executive Director of Edmonton Family Planning Service.

<sup>2</sup>From an interview with D. Storch, Chief of Services of Family Service Association; and Family Service Association of Edmonton, and 1973 Annual Report.

.0

<sup>3</sup>From an interview with L. Hovedskov, the supervisor of Family Aide; and a handout on 'Family Aide Service' by Edmonton Social Service Department.

<sup>4</sup>From an interview with G. Hornby, General Manager of Canative Housing Corporation.

<sup>5</sup>From an interview with K. Purves, Director of St. Francis Youth Club; and a pamphlet from St. Francis Youth Club.

<sup>6</sup>From an interview with W. Szebel, Executive Director of Teen Times; <u>1974 Teen Time of Edmonton Report and Evaluation</u> prepared for Department of Preventive Social Services, City of Edmonton and Province of Alberta; and <u>Report on Teen Time of Edmonton</u> by Karen Wundorf.

<sup>7</sup>From an interview with J. Dorn, Coordinator of the Bissell Child Care Centre; and a telephone conversation with J. Lightfoot, a founder of the Bissell Child Sare Centre.

 $\beta_{\rm From an}$  interview with G. Wallis <u>et al</u>. of Operation Friendship, and a report prepared by Operation Friendship.

<sup>9</sup>From an interview with T. Allen, Strathcona Place Society, and Annual Report 1974, submitted to Preventive Social Service in 1975.

10<sub>From an interview with C. Rensing, Coordinator of Dickensfield We Care Centre, and Report on the Activities of the Dickensfied We Care Centre, 1973-74.</sub>

11 From an interview with A. Panych, Coordinator of Open Door, and <u>Annual Report</u> of Open Door, Information and Referral Centre, 1974.

<sup>12</sup> From an interview with G. Wood, Executive Director of A.I.D. Service, and <u>Annual Report, 1974</u>, by A.I.D. Service of Edmonton. 202

13<sub>From</sub> an interview with E. Schlesinger, F. Klemen, D. Freeman, A. Marchand, D. Jacques, L. Desjardins, R. Levin, and A. Thal of Edmonton Social Services Department.

÷.•

0

14 From interviews with N. Davis, Dirstict 'Director of Victoria Order of Nurses and the Coordinator of Meals-on-Wheels, and pamphlets prepared by Meals-on-Wheels.